

Brown, Raphael (2015) *A novel AlGaN/GaN based enhancement-mode high electron mobility transistor with sub-critical barrier thickness.* PhD thesis.

http://theses.gla.ac.uk/6590/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk UNIVERSITY OF GLASGOW

## A NOVEL AlGaN/GaN BASED ENHANCEMENT MODE HIGH ELECTRON MOBILITY TRANSISTOR WITH SUB-CRITICAL BARRIER THICKNESS

by

### RAPHAEL BROWN

A thesis submitted in partial fulfillment for the degree of Doctor of Philosophy in the

School of Engineering University of Glasgow

Copyright ©~2015 by Raphael Brown

### Abstract

Power-switching devices require low on-state conduction losses, high-switching speed, high thermal stability, and high input impedance. Using gallium nitride (GaN) based field-effect transistors, these properties for switching devices can be satisfied. GaN-based High Electron Mobility Transistors (HEMTs) are emerging as promising candidates for high-temperature, high-power (power electronics) and radio-frequency (RF) electronics due to their unique capabilities of achieving higher current density, higher breakdown voltage, higher operating temperatures and higher cut-off frequencies compared to silicon (Si). Conventional GaN HEMTs with an aluminium gallium nitride (AlGaN) barrier are of depletion-mode (d-mode) or normally-on which require a negative polarity power supply to turn off. On the other hand, enhancement-mode (e-mode) or normally-off Al-GaN/GaN HEMTs are attracting increasing interest in recent years because no negative gate voltage is necessary to turn off the devices. This leads to the advantage of simple circuit design and low stand-by power dissipation. For power electronics applications, power switches which incorporate e-mode devices provide the highly desirable essential fail-safe operation.

high performance normally-off GaN-based In this research, a new metal-oxidesemiconductor (MOS) high electron mobility transistor (HEMT) that employs an ultrathin sub-critical 3 nm Al<sub>0.25</sub>Ga<sub>0.75</sub>N barrier layer and relies on an induced two dimensional electron gas (2DEG) for operation was designed, fabricated and characterized. The device consists of source and drain Ohmic contacts nominally overlapped by the gate contact and employs a gate dielectric. With no or low gate-to-source voltage ( $V_{GS}$ ), there is no two dimensional electron gas (2DEG) channel at the AlGaN/GaN interface to allow conduction of current between the drain and source contacts as the AlGaN barrier thickness is below the critical thickness required for the formation of such channel. However, if a large enough positive bias voltage  $V_{GS}$  is applied, it causes the formation of a quantum well at the AlGaN/GaN interface into which electrons from the source and drain Ohmic regions are attracted (by the positive gate voltage), effectively creating a 2DEG channel, and so the structure is a normally-off field effect transistor.

Normally-off GaN MOS-HEMT devices were fabricated using plasma enhanced chemical vapour-deposited (PECVD) silicon dioxide (SiO<sub>2</sub>) as the gate dielectric. They demonstrated positive threshold voltages (V<sub>th</sub>) in the range of +1 V to +3 V, and very high maximum drain currents (I<sub>DSmax</sub>) in the range of 450 mA/mm to 650 mA/mm, at high gate voltage (V<sub>GS</sub>) of around 6 V. The devices also exhibited breakdown voltages in the range of 9 V and 17 V depending on the gate dielectric thickness, making them suitable for realising high current low voltage power devices required, for instance, for buck converters for mobile phones, tablets, laptop chargers, etc.

### Acknowledgements

First of all, I would like to thank the Almighty God for granting me the strength and willpower to carry out this research.

Secondly, I would like to thank my family and parents, especially my father, Mr. Raphael A. Brown, for financially supporting me and giving me moral support and guidance throughout my research.

My sincerest gratitude goes out to my project supervisor, Dr. Edward Wasige, for giving me the opportunity to carry out this project and for mentoring and advising me throughout the project.

I would also like to extend my sincerest gratitude for the contributions made by Prof. Iain Thayne, Dr. Gary Ternent and Dr. Xu Li, all who provided useful advice during this project.

I am also grateful to my colleagues Abdullah Al-Khalidi, Konstantinos Floros and Dr. Sung-Jin Cho who contributed towards the successful completion of this project through informative academic discussions and conversations, as well as helping with device characterization.

I would like to thank the staff of the James Watt Nanofabrication Centre (JWNC) at University of Glasgow for providing training on equipments and for their help in fabricating the devices reported in this project. My special thanks also goes out to Dr. Haiping Zhou, Dougie Lang, Mark Dragsnes, Eve Aitkenhead, Linda Pollock, Leslie Donaldson, Helen Mclelland, Ronnie Roger, Thomas Reilly and Robert Harkins for their technical support in assisting in all fabrication work.

I would also like to acknowledge Dr. M. Caesar, Prof. Michael Uren, and Prof. Martin Kuball of the Centre for Device Thermography and Reliability (CDTR) at Bristol University for evaluating the stability of the devices.

Finally, a special thank you to my wife Nan Korotoum Brown, who has consistently been an encouraging influence and a source of inspiration and whose love and support has helped me tremendously throughout. I am also most grateful to my children who have provided me with a source of joy and happiness throughout this project.

### Publications

- <u>R. Brown</u>, D. Macfarlane, A. Al-Khalidi, X. Li, G. Ternent, H. Zhou, I. Thayne and E. Wasige, "A Sub-Critical Barrier Thickness Normally-Off AlGaN/GaN MOS-HEMT," *IEEE Electron Device Letters*, vol. 35, no. 9, pp. 906-908, September 2014.
- <u>R. Brown</u>, A. Al-Khalidi, D. Macfarlane, S. Taking, G. Ternent, I. Thayne and E. Wasige, "Novel High Performance AlGaN/GaN based Enhancement-Mode Metal-Oxide Semiconductor High Electron Mobility Transistor," *Physica Status Solidi* (C), vol. 11, no. 3-4, pp. 844-847, January 2014.
- <u>R. Brown</u>, A. Al-Khalidi, D. Macfarlane, S. Taking and E. Wasige, "A Novel High Performance AlGaN/GaN based Enhancement-Mode Metal-Oxide-Semiconductor High Electron Mobility Transistor," *The* 10<sup>th</sup> *International Conference on Nitride Semiconductors*. Washington, August 2013.
- <u>R. Brown</u>, A. Al-Khalidi, G. Ternent, I. Thayne and E. Wasige, "A Normallyoff AlGaN/GaN MOSHEMT Technology," *The* 22<sup>nd</sup> European Workshop on Heterostructure Technology. Glasgow, September 2013.
- <u>R. Brown</u>, A. Al-Khalidi, D. Macfarlane, S. Taking, G. Ternent, I. Thayne and E. Wasige, "A Normally-off AlGaN/GaN HEMT Technology," UK Nitrides Consortium. Sheffield, July 2013.
- J. Sinclair, <u>R. Brown</u>, A. Al-Khalidi and E. Wasige, "Fabrication and Characterisation of normally-off GaN-based MOS-HEMTs," *The* 7<sup>th</sup> *Institution of Engineering and Technology International Conference on Power Electronics, Machines and Drives.* Manchester, April 2014.
- J. Wang, K. Alharbi, A. Ofiare, <u>R. Brown</u>, A. Khalid, D. Cumming and E. Wasige, "MMIC Resonant Tunnelling Diode Oscillators for THz applications," *The* 11<sup>th</sup> *Conference on PhD Research in Microelectronics and Electronics (IEEE PRIME* 2015). Glasgow, June 2015.
- J. Wang, K. Alharbi, A. Ofiare, <u>R. Brown</u>, A. Khalid, D. Cumming and E. Wasige, "D-band Resonant Tunnelling Diode Oscillator with High Output Power," UK Semiconductors 2015. Sheffield, July 2015.
- A. Ofiare, <u>R. Brown</u>, J. Wang, K. Alharbi, A. Khalid, D. Cumming and E. Wasige, "Accurate Pulsed I-V Measurement of the NDR Region of Tunnel Diode and Resonant Tunnelling Diodes," *UK Semiconductors 2015.* Sheffield, July 2015.

## Contents

Abstract	i
Acknowledgements	ii
Publications	iii
List of Figures	viii
List of Tables	xii
Abbreviations	xiii
Physical Constants	xv
Symbols	xvi

1	INT	FRODUCTION	1
	1.1	Brief History of GaN Technology	7
	1.2	Application Areas of GaN	10
	1.3	Gallium Nitride Power Semiconductors Market	10
	1.4	Research Aims and Objectives	12
	1.5	Organization of Thesis	13
	1.6	Original Contribution to Research Field	14
2	Gal	N HIGH ELECTRON MOBILITY TRANSISTORS	15
	2.1	GaN Material System	15
		2.1.1 Material Properties and Crystal Structure of III-Nitride Semicon-	
		ductors	16
		2.1.2 Polarization Effects in the AlGaN/GaN Heterostructure	17
		2.1.2.1 Simulation of 2DEG Sheet Density in GaN-Based HEMTs	21
	2.2	2.1.2.1 Simulation of 2DEG Sheet Density in GaN-Based HEMTs The AlGaN/GaN HEMT	$21 \\ 25$
	2.2	2.1.2.1Simulation of 2DEG Sheet Density in GaN-Based HEMTsThe AlGaN/GaN HEMT2.2.1Fundamentals of HEMTs2.2.1	$21 \\ 25 \\ 25$
	2.2	2.1.2.1Simulation of 2DEG Sheet Density in GaN-Based HEMTsThe AlGaN/GaN HEMT2.2.12.2.1Fundamentals of HEMTs2.2.2Operation of an AlGaN/GaN HEMT	21 25 25 26

	2.3	Growt	h Techniques for Gallium-Based Binary and			
		Ternar	y Compounds		•	 29
		2.3.1	Hydride Vapour Phase Epitaxy			 30
		2.3.2	Metal Organic Chemical Vapour Deposition			 30
		2.3.3	Molecular Beam Epitaxy			 31
	2.4	Substr	ate Choice for AlGaN/GaN HEMTs			 31
		2.4.1	Silicon Carbide Substrates (SiC)			 31
		2.4.2	Sapphire Substrates			 31
		2.4.3	Silicon Substrates			 32
		2.4.4	GaN and AlN Substrates			 33
	2.5	Metal-	Semiconductor Contacts			 33
		2.5.1	Schottky Contact			 33
		2.5.2	Ohmic Contact			 34
		2.5.3	Transfer Length Method Structures			 35
			2.5.3.1 Linear TLMs			 36
			2.5.3.2 Circular TLMs (CTLMs)			 38
	2.6	Summ	ary	•	•	 39
3	Gal	N DEV	ICE PROCESSING TECHNOLOGY			40
	3.1	Etchin	g Techniques for GaN-Based Devices		•	 40
		3.1.1	Dry Etching			 40
			3.1.1.1 Reactive Ion Etching		•	 41
			3.1.1.2 Inductively-Coupled Plasma Reactive Ion Etching		•	 42
		3.1.2	Wet Etching		•	 43
	3.2	Lithog	raphy		•	 43
		3.2.1	Optical Lithography		•	 43
		3.2.2	Electron Beam Lithography		•	 44
	3.3	Dielect	tric Deposition Techniques		•	 46
		3.3.1	Plasma Enhanced Chemical Vapour Deposition		•	 46
		3.3.2	Inductively-Coupled Plasma Chemical Vapour Deposition .		•	 48
		3.3.3	Atomic Layer Deposition		•	 49
	3.4	Summ	ary	•	•	 49
4	RE	VIEW	OF GaN-BASED E-MODE HEMTs			51
	4.1	Double	e Barrier HEMT with an Etch-Stop Layer	•	•	 51
	4.2	Fluori	ne Ion Implantation	•	•	 52
	4.3	P-N J	unction Gate GaN HEMTs	•	•	 53
	4.4	InGaN	Cap Layer GaN HEMTs	•	•	 54
	4.5	Tunne	l-Junction GaN HEMTs	•	•	 54
	4.6	N-Pola	ar GaN HEMTs	•	•	 55
	4.7	Gate I	Recessed GaN HEMTs		•	 56
	4.8	Thin I	Barrier Devices	•	•	 57
	4.9	Summ	ary	•	•	 58
5	DE	VICE 1	FABRICATION PROCESSES			59
	5.1	Device	e Layout and Process Flows		•	 59
		5.1.1	Device Structure			 -59

		5.1.2	Process Flows					. 61
			5.1.2.1 Sample Cleanir	g and Preparation				. 62
			5.1.2.2 Photolithograp	hy				. 62
			5.1.2.3 Metallization .					. 65
			5.1.2.4 Ohmic Anneali	ng				. 66
			5.1.2.5 Dielectric Depc	sition				. 66
			5.1.2.6 Dry Etching .					. 67
		5.1.3	Mask Design and Lavout					. 67
	5.2	Summ	ary					. 68
6	NO	VEL N	ORMALLY-OFF AIG	aN/GaN MOS-H	ЕМТ			72
Ŭ	6.1	Norm	llv-off AlGaN/GaN MOS	-HEMT on Sapphir	e Substra	ite		72
	0.1	6.1.1	Device Structure and Fa	brication				. 72
		612	Device Output and Tran	sfer Characteristics				73
		613	Offstate Breakdown Cha	racteristics				75
		614	Device Reliability					. 10
	6.2	Norm	llv-off AlGaN/GaN MOS	HEMT on Silicon	Substrat	••••		
	0.2	6 2 1	Device Structure and Fa	brication	Substrat		•••	. 10
		622	Device Output and Tran	sfor Characteristics			• • •	. 10
	63	Summ	Device Output and Iran	Sier Onaracteristics		• • • •	•••	. 13
	0.0	Summ	uy					. 01
7	CO	NCLU	SION AND FUTURE	WORK				89
	7.1	Concl	$\operatorname{sion}$					. 89
	7.2	Futur	Work and Research Reco	ommendations				. 90
		7.2.1	Ohmic Contacts					. 90
		7.2.2	All GaN Cascode Device	Configuration				. 90
Α	FAI	BRICA	<b>FION PROCESSES</b>					92
	A.1	Samp	Scribing $\ldots$					. 92
	A.2	Samp	Cleaning $\ldots$					. 92
	A.3	Ohmi	Lithography Recipe $1$ .					. 92
	A.4	Ohmi	Metallization Recipe $1$ .					. 93
	A.5	Gate	ithography Recipe 1					. 93
	A.6	Gate	Ietallization Recipe 1					. 94
	A.7	Ohmi	Lithography Recipe 2 .					. 94
	A.8	Ohmi	Metallization Recipe 2 .					. 94
	A.9	Gate	ithography Recipe 2					. 95
	A.1(	) Gate 1	Ietallization Recipe 2					. 95
	A.11	l Bondı	ad Formation Recipe $2$ .					. 96
	A.12	2 Bondi	ad Metallization Recipe 2					. 96
	A.13	3 E-mo	e AlGaN/GaN MISHEM	Γ with Silicon Dioxid	de Gate I	nsulator	c De-	-
		vice F	brication $\ldots$					. 96
	A.14	4 AlGal	/GaN MISHEMT with S	ilicon Nitride Gate l	[nsulator	Device	Fab-	
		ricatio	, 1					. 97
в	SO	ME SE	MICONDUCTOR DE	VICE BASICS				99

	B.1	Metal/Semiconductor Contacts	99
	B.2	Carrier Transport Mechanisms across Metal/Semiconductor Junctions .	101
		B.2.1 Thermionic Emission	101
		B.2.2 Field Emission	102
		B.2.3 Thermionic Field Emission	102
	B.3	Metal Insulator Semiconductor Field	
		Effect Transistor (MISFET)	103
		B.3.1 Operation of a MOSFET with No Gate Voltage	104
		B.3.2 Creating a Channel for Current Flow	104
		B.3.3 Operation at Small Drain-Source Voltage	105
		B.3.4 Operation at Higher Drain-Source Voltage	105
	<b>B.</b> 4	Summary	106
С	SIL FAC	ICON NITRIDE GATE DIELECTRIC MISHEMTs AND INTER- CE TRAPS	108
	C.1	AlGaN/GaN MISHEMT with SiN Gate Insulator	108
	C.2	Metal-Oxide-Semiconductor Capacitors	110
	C.3	Summary	114

#### Bibliography

115

# List of Figures

1.1	Roadmap of the life-cycle of competing semiconductor materials, Si, GaN and SiC, indicating a new generation every 20 years.		2
1.2	Summary of Si, SiC and GaN relevant material properties		2
1.3	Electronic applications of GaN-based devices.		11
1.4	Yole's report showing that automakers will begin to adopt GaN power		
	devices in inverters, dc-dc converters, and on-board chargers, generating		
	revenue of \$150 million by 2020.	•	12
2.1	Wurtzite unit cell of GaN showing the lattice constants $a_0$ and $c_0$		16
2.2	Atomic arrangement in Ga-face and N-face GaN crystals. Spontaneous		
	polarization vector is also shown.	•	18
2.3	Directions of spontaneous and piezoelectric polarization in Ga and N-face		
	AlGaN/GaN heterostructures		19
2.4	Conduction energy band of an AlGaN/GaN structure		22
2.5	Total sheet carrier concentration of the 2DEG formed at the AlGaN/GaN $$		
	interface versus different alloy compositions for various AlGaN barrier		
	layer thickness.	•	24
2.6	Variations of total polarization induced sheet carrier concentration of the		
	2DEG formed at the AlGaN/GaN interface versus various AlGaN barrier		
	thickness for different alloy compositions.	•	25
2.7	Typical IV characteristics of a HEMT. The inset shows a common source		~-
	configuration for a transistor.	•	27
2.8	A basic AlGaN/GaN HEMT.	•	28
2.9	A basic AlGaN/GaN HEMT with corresponding energy band diagram.	•	29
2.10	A semiconductor material with Ohmic contact pads for TLM characteri-		
0.11		·	36
2.11	A graph showing an example of plot of total resistance as a function of		97
0.10	pad spacing.	•	37
2.12	A cross-section and top view of AlGaN/GaN semiconductor material with Obmis contact pade for CTLM abaractorization		<b>9</b> 0
	Ommic contact pads for CTLM characterization.	•	30
3.1	A schematic of a reactive ion etching (RIE) tool. $\ldots$	•	41
3.2	A schematic of an inductively-couple plasma reactive ion etching (ICP-		
	RIE) tool	•	42
3.3	The optical system of a mask aligner used to replicate a mask pattern		
	during optical lithography.	•	44
3.4	An image of a negative photoresist (left) and positive photoresist (right)		
	after exposure, development and etch.	•	45

3.5	A schematic of a plasma enhanced chemical vapour deposition (PECVD) tool.	47
3.6	A schematic of an inductively-coupled plasma chemical vapour deposition (ICP-CVD tool).	48
3.7	A schematic of an atomic layer deposition (ALD) tool	50
4.1	Double barrier AlGaN/GaN HEMT structure with etch-stop layer	52
4.2	Achieving an e-mode GaN HEMT device using fluorine ion implantation.	53
$\begin{array}{c} 4.3\\ 4.4\end{array}$	P-N junction gate e-mode GaN HEMT	54
4 5	cap layer.	55
4.5	Energy band diagrams: a) with and b) without InGaN cap layer. $\ldots$	55
4.6 4.7	Cross section of a tunnel-junction AlGaN/GaN FE1	50
10	Cate recorded AlCoN/CoN/HEMT structure	50 57
4.0	Gate recessed Argan/Gan memi structure	57
5.1	A screen shot of a complete device layout design in L-Edit including alignment markers, test structures and DC transistors	60
5.2	Typical process flow used to fabricate e-mode HEMTs in this research.	
	PR stands for photoresist.	62
5.3	Photolithography process using recipe 1. a) sample is cleaned b) pho- toresist is spun onto sample c) sample is soaked in developer solution and exposed under UV light with mask plate in contact with sample d) pho- toresist is developed giving an undercut profile e) sample is metallized f) lift-off process is performed to remove photoresist and unwanted metal while leaving behind only the metal which is directly evaporated onto the	
	sample surface.	64
5.4	Photolithography process using recipe 2. a) clean sample b) spin and bake LOR 10A and photoresist c) expose sample under UV light d) develop resist and LOR 10A e) deposit metal f) lift-off the bi-layer stack and	
	unwanted metal from sample surface	65
5.5	Optical microscope image of Ohmic contacts for a gate wrap-around de- vice after metallization before and after rapid thermal annealing	67
5.6	Optical microscope image of gate wrap-around device before and after dielectric etch and Ohmic bondpad formation.	68
5.7	L-Edit layout of individual cells of a gate wrap-around transistor layers	60
5.8	L-Edit layout of a gate wrap-around HEMT (source and drain Ohmic	09
0.0	contacts are nominally overlapped by the gate contact)	69
5.9	L-Edit layout of Ohmic contacts on mask plate which was used for Ohmic lithography during fabrication of e-mode HEMTs.	70
5.10	L-Edit layout of gate contacts on mask plate which was used for gate	70
E 11	L Edit largert of hendrode on mode rists which we used for her here	70
5.11	lithography during fabrication of e-mode HEMTs.	71
5.12	L-Edit layout of the three lithography layers used during photolithograpy to fabricate e-mode HEMTs after successful alignment.	71

6.1	Cross-section of the proposed e-mode AlGaN/GaN device	73
6.2	Output characteristics of an e-mode $6 \mu m \times 100 \mu m$ AlGaN/GaN MOS- HEMT with 10 nm thick SiO <sub>2</sub> dielectric. Note the (near) zero drain cur-	
	rents for $V_{GS}$ of $0V$ , $1V$ , $2V$ , and $3V$ .	74
6.3	Output characteristics of an e-mode $6\mu\text{m} \times 100\mu\text{m}$ AlGaN/GaN MOS-	
	HEMT with 20 nm thick ${\rm SiO}_2$ dielectric. Drain current traces for ${\rm V}_{\rm GS}$ of	
	0V, 1V, and 2V are overlapping	75
6.4	Transfer characteristics and transconductance of e-mode AlGaN/GaN	76
6.5	MOS-HEM1s at low drain bias voltages of 1 v and 3 v	70
0.0	HEMTs with 10 nm and 20 nm thick SiO <sub>2</sub> gate dielectrics.	76
6.6	Variation of the threshold voltage of the proposed e-mode MOS-HEMT	
	employing 20 nm thick $SiO_2$ gate dielectric with stress time	77
6.7	Cross-section of the proposed e-mode AlGaN/GaN device on silicon sub-	
	strate	78
6.8	Scanning electron microscopy (SEM) image of normally-off AlGaN/GaN	
	device structure fabricated for DC measurements. Figure a) shows the	
	section which was cut in order to take transmission electron microscopy	
	(TEM) images of the cross-section of the device, and Figure c) shows a	
	zoomed in image of the section which was cut for TEM analysis	79
6.9	TEM image of cross-section of gate overlap device showing the dielectric	
	step coverage	80
6.10	Output characteristics of an e-mode $4 \mu m \times 100 \mu m$ AlGaN/GaN MOS-	
	HEMT with 20 nm thick $SiO_2$ dielectric. Note the (near) zero drain cur-	80
6 11	Output characteristics of an e-mode $6 \mu m \times 100 \mu m$ AlGaN/GaN MOS-	00
0.11	HEMT with 20 nm thick $SiO_2$ dielectric. Note the (near) zero drain cur-	
	rents for $V_{GS}$ of 0 V, 1 V and 2 V	81
6.12	Output characteristics of an e-mode $8\mu\mathrm{m}\times100\mu\mathrm{m}$ AlGaN/GaN MOS-	
	HEMT with 20 nm thick $SiO_2$ dielectric. Note the (near) zero drain cur-	01
C 19	rents for $V_{GS}$ of $0V$ , $1V$ and $2V$	81
0.13	Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode $4 \mu m \times 100 \mu m$ AlCaN/CaN MOS-HEMT with 20 nm thick	
	SiO <sub>2</sub> dielectric at drain bias voltages of 1 V and 10 V. $\dots \dots \dots \dots$	82
6.14	Transfer characteristics (left y-axis) and transconductance (right y-axis)	-
	of an e-mode $6\mu\text{m} \times 100\mu\text{m}$ AlGaN/GaN MOS-HEMT with 20 nm thick	
	$SiO_2$ dielectric at drain bias voltages of $1 V$ and $10 V$	83
6.15	Transfer characteristics (left y-axis) and transconductance (right y-axis)	
	of an e-mode $8 \mu\text{m} \times 100 \mu\text{m}$ AlGaN/GaN MOS-HEMT with 20 nm thick	0.0
6 16	$SiO_2$ dielectric at drain bias voltages of 1 V and 10 V	83
0.10	HEMT with 30 nm thick SiO <sub>2</sub> dielectric. Note the (near) zero drain cur-	
	rents for $V_{CS}$ of $0V$ and $1V$ .	84
6.17	Output characteristics of an e-mode $6 \mu\text{m} \times 100 \mu\text{m}$ AlGaN/GaN MOS-	
	HEMT with $30 \text{ nm}$ thick SiO <sub>2</sub> dielectric. Note the (near) zero drain cur-	
	rents for $V_{GS}$ of 0 V, 1 V and 2 V	85

6.18	Output characteristics of an e-mode $8 \mu m \times 100 \mu m$ AlGaN/GaN MOS- HEMT with 30 nm thick SiO <sub>2</sub> dielectric. Note the (near) zero drain cur-	<b>۲</b>
6.19	rents for $V_{GS}$ of $0 V$ , $1 V$ and $2 V$ . Transfer characteristics (left y-axis) and transconductance (right y-axis)	. 85
	of an e-mode $4 \mu\text{m} \times 100 \mu\text{m}$ AlGaN/GaN MOS-HEMT with 30 nm thick SiO <sub>2</sub> dielectric at drain bias voltages of 0.5 V and 10 V	86
6 20	Transfer characteristics (left y-axis) and transconductance (right y-axis)	. 80
0.20	of an e-mode $6\mu\text{m} \times 100\mu\text{m}$ AlGaN/GaN MOS-HEMT with 30 nm thick	
	$SiO_2$ dielectric at drain bias voltages of $0.5 V$ and $10 V$	. 86
6.21	Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode $8 \mu m \times 100 \mu m$ AlGaN/GaN MOS-HEMT with 30 nm thick	
	$SiO_2$ dielectric at drain bias voltages of $0.5 V$ and $10 V$	. 87
6.22	Gate leakage current as functions of gate bias of e-mode AlGaN/GaN MOS-HEMTs with $20 \text{ nm}$ thick SiO <sub>2</sub> dielectric.	. 87
7.1	A cascode configuration featuring a high-voltage normally-on GaN HEMT	
	with a low-voltage normally-off GaN HEMT.	. 91
B.1	A metal/semiconductor junction	. 100
B.2	Structure of an n-channel silicon MOSFET	. 103
B.3	A MOSFET with a channel for current flow	. 104
B.4	A MOSFET with small drain-source voltage applied	. 105
B.5	A MOSFET with high drain-source voltage applied.	. 106
B.6	IV Characteristic of an enhancement-type NMOS transistor	. 107
C.1	MISHEMT with $SiN_x$ gate insulator.	. 109
C.2	Output characteristics of a 3 nm thick barrier $4 \mu m \times 100 \mu m \text{ AlGaN/GaN}$ MOS-HEMT with 20 nm thick SiN dielectric.	. 109
C.3	Transfer characteristics and $G_m$ of a $3\mathrm{nm}$ thick barrier $4\mu\mathrm{m}\times100\mu\mathrm{m}$	
	AlGaN/GaN MOS-HEMT with 20 nm thick SiN dielectric	. 110
C.4	C-V graph of $250 \mu\text{m}$ MOSCAP structure with $20 \text{nm}$ SiO <sub>2</sub> gate dielectric without surface treatment prior to dielectric deposition (Sample 1)	. 112
C.5	C-V graph of 250 $\mu$ m MOSCAP structure with 20 nm SiO <sub>2</sub> gate dielectric	
	with surface treatment prior to dielectric deposition (Sample 2).	. 112
C.6	C-V graph of $250\mu\mathrm{m}$ MOSCAP structure with $20\mathrm{nm}$ SiO_2 gate dielec-	
	tric with surface treatment prior to dielectric deposition and with di-	
	electric treatment (annealing at 900 °C for 30 minutes in nitrogen ambi-	
	ent)(Sample 3). $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	. 113

## List of Tables

1.1	Properties of GaN and competing semiconductor materials in power elec-		
	tronics.		4
1.2	Figures of merit of GaN and competing semiconductor materials in power electronics.		6
2.1	Lattice parameters of wurtzite III-nitrides at 300 K.		17
2.2	Basic properties of substrates and III-N semiconductor materials	,	32
2.3	Various metals and their workfunctions	,	34

## Abbreviations

2DEG	Two Dimensional Electron Gas
ALD	Atomic Layer Deposition
AlGaN	Aluminium Gallium Nitride
AlN	Aluminium Nitride
BFOM	Baliga's Figure Of Merit
BHFFOM	Baliga's High Frequency Figure Of Merit
CFOM	Combined Figure Of Merit
CVD	Chemical Vapour Deposition
ECR-RIE	${\bf E} lectron {\bf - C} y clotron {\bf - R} esonance \ {\bf R} eactive \ {\bf I} on \ {\bf E} tching$
FET	Field Effect Transistor
FOM	Figure Of Merit
GaN	Gallium Nitride
HEMT	$\mathbf{H} igh \ \mathbf{E} lectron \ \mathbf{M} obility \ \mathbf{T} ransistor$
HVPE	$\mathbf{H}$ ydride $\mathbf{V}$ apour $\mathbf{P}$ hase $\mathbf{E}$ pitaxy
ICP-CVD	Inductively-Coupled-Plasma Chemical Vapour Deposition
ICP-RIE	$\mathbf{I} \mathbf{n} \mathbf{d} \mathbf{u} \mathbf{c} \mathbf{i} \mathbf{v} \mathbf{e} \mathbf{l} \mathbf{o} \mathbf{r} \mathbf{e} \mathbf{c} \mathbf{i} \mathbf{v} \mathbf{e} \mathbf{f} \mathbf{e} \mathbf{e} \mathbf{e} \mathbf{f} \mathbf{e} \mathbf{e} \mathbf{e} \mathbf{e} \mathbf{e} \mathbf{e} \mathbf{e} e$
InGaN	Indium Gallium Nitride
JFET	<b>J</b> unction <b>F</b> ield <b>E</b> effect <b>T</b> ransistor
JFOM	Johnson's Figure Of Merit
KFOM	Keyes Figure Of Merit
MBE	$\mathbf{M}$ olecular $\mathbf{B}$ eam $\mathbf{E}$ pitaxy
MISFET	$\mathbf{M}$ etal-Insulator Field Effect Transistor
MOCVD	Metal Organic Chemical Vapour Deposition
MOSFET	$\mathbf{M} etal\textbf{-}\mathbf{O} xide\textbf{-}\mathbf{S} emiconductor \ \mathbf{F} ield \ \mathbf{E} ffect \ \mathbf{T} ransistor$
MOSHEMT	Metal-Oxide-Semiconductor High Electron Mobility Transistor

PECVD	$\mathbf{P} \text{lasma-Enhanced } \mathbf{C} \text{hemical } \mathbf{V} \text{apour } \mathbf{D} \text{eposition}$
RIE	Reactive Ion Etching
Si	Silicon
SiC	Silicon Carbide

# **Physical Constants**

Constant Name	$\mathbf{Symbol}$	=	Constant Value (with units)
Speed of light	с	=	2.997 924 58 × 10 <sup>8</sup> (m/s) (exact)
Charge of electron	e	=	$1.602\times 10^{-19}~(Coulombs)$
Plank's constant	h	=	$6.626 \times 10^{-34} \ (Joules  sec)$
Mass of electron	$m_e$	=	$9.109 \times 10^{-31} \ (Kilograms)$
Reduced Planck's constant	$\hbar$	=	$1.055\times 10^{-34}~(Joulessec)$

## Symbols

A	Area	$\mathrm{m}^2$
Р	Power	$W (Js^{-1})$
$E_C$	Conduction band energy	eV
$E_c$	Critical electric field	$V/m^{-1}$
$E_f$	Fermi level energy	eV
$E_g$	Bandgap	eV
$E_v$	Valence band energy	eV
$\Delta E_v, \Delta E_C$	Discontinuity of the valence/conduction band	
	at a heterointerface	eV
$\Delta E_g$	Total difference of the bandgaps	
	at a heterointerface	eV
$f_{MAX}$	Maximum frequency of operation	Hz
$\Delta f$	Frequency interval	Hz
$f_T$	Unity current gain cut-off frequency	Hz
$Ga_2O_3$	Gallium oxide	
$G_M$	Transconductance	$\mathrm{mS/mm}$
$H_2$	Hydrogen	
$H_2O$	Water	
$I_D$	Drain current	А
$I_{DS}$	Drain-source current	А
$I_S$	Source current	А
$L_{GD}$	Gate-drain length	m
$L_{GS}$	Gate-source length	m
$L_{SD}$	Source-drain length	m
$L_T$	Transfer length of an Ohmic contact	m

$N_2$	Nitrogen	
$NH_3$	Ammonia	
$P_{PE}$	Piezoelectric polarization	$C/m^2$
$P_{SP}$	Spontaneous polarization	$C/m^2$
$R_C$	Ohmic contact resistance	$\Omega.\mathrm{mm}$
$R_{CH}$	Resistance of the channel region	
	under the gate-foot	Ω
$R_{DS}$	Drain-source resistance	Ω
$R_{on}$	On-resistance of a device	$\Omega.\mathrm{mm}~\mathrm{or}~\Omega.\mathrm{mm}^2$
$R_{SH}$	Sheet resistance	$\Omega/\Box$
$V_{BR}$	Breakdown voltage	V
$V_{DS}$	Drain-source Voltage	V
$V_{FB}$	Flat band voltage	V
$V_{GS}$	Gate-source voltage	V
$V_P$	Pinch-off voltage	V
$V_{TH}$	Threshold voltage	V
$V_x$	Effective channel voltage	V
$SF_6$	Sulphur hexafluoride	
$Si_3N_4, SiN$	Silicon nitride	
$t_{OX}$	Thickness of oxide	m
$T_{OX}$	Oxidation time	S
$W_g$	Gate width	m
$x_0$	Initial thickness of oxide film	m
$v_{sat}$	Saturation electron velocity	$\rm m/s^{-1}$
ω	angular frequency	$rads^{-1}$
ε	Permittivity	F/m
$\varepsilon_o$	Vacuum permittivity	F/m
$\varepsilon_r$	Dielectric constant	
$\chi, \chi_L$	Thermal conductivity	W/cm.K
$\mu$	Electron mobility	$\mathrm{cm}^2/\mathrm{V.s}$

This thesis is dedicated to my late grandmother, Mary Aba Tetteh and my son, Raphael Ilan Brown

### CHAPTER 1

### INTRODUCTION

Since the production of the first silicon transistor by Gordon Teal in 1954 [1], silicon (Si) has assumed a central role in the development of semiconductor devices. Silicon has been the dominant semiconductor of choice for high-voltage switching applications to date. Recently, however, silicon technology is fast approaching its theoretical limits imposed by the intrinsic material properties such as low saturation velocity, low breakdown voltage, low inversion layer mobility and high device resistance [2]. Due to these limitations, there is growing interest in research activities in modern electronics towards new materials which are able to satisfy the specific needs of higher operating frequencies, higher output power and higher operating voltages. Wide-bandgap semiconductors such as gallium nitride (GaN) and silicon carbide (SiC) have demonstrated the potential for meeting the requirements for high-temperature and high-power switching applications. Figure 1.1 shows the roadmap of the life-cycle of the main semiconductors used in power device technologies, Si, GaN and SiC [3]. It can be seen that wide-bandgap semiconductors such as GaN and SiC are expected to dominate the modern era of power device technology for the foreseeable future. Figure 1.2 shows the relevant material properties for Si, SiC and GaN which include the energy gap, (breakdown) electric field, thermal conductivity, electron velocity, and melting point. This figure gives a good indication of the advantages of GaN over its main competitors in the semiconductor market. The large bandgap energy results in a high electric breakdown field, which is an order of magnitude higher than that of Si. The high electric breakdown field enables the possibility to sustain the application of high bias voltages, thus making GaN suitable for the fabrication of high-voltage devices.

Table 1.1 shows the values of the fundamental material properties of GaN and competing semiconductor materials in power electronics [4-6]. Due to its superior intrinsic



FIGURE 1.1: Roadmap of the life-cycle of competing semiconductor materials, Si, GaN and SiC, indicating a new generation every 20 years.



FIGURE 1.2: Summary of Si, SiC and GaN relevant material properties.

physical properties including wide bandgap  $(E_q)$  of around 3.4 eV, high breakdown electric field ( $E_c$ ) of 3.3 MV/cm, high electron saturation velocity ( $v_{sat}$ ) of  $2.5 \times 10^7$  cm/s and high density carriers in the form of two-dimensional electron gas (2DEG) with high mobility ( $\mu$ ) of around 2000 cm<sup>2</sup>/Vs [5], GaN is considered an outstanding material for high-frequency and high-power devices as well as in the area of opto-electronics. Widebandgap semiconductor power devices offer great performance improvements and can work in harsh environments where silicon power devices cannot function. One of the main advantages of III-nitride materials such as gallium nitride (GaN) is the ability to form a heterojunction with a ternary alloy made from another III-nitride semiconductor material such as aluminium gallium nitride (AlGaN). Due to the strong polar material properties, the modification of the material composition results in dramatic modifications of the polar crystal properties and thus of available carrier concentrations obtained at the heterojunction interfaces in the devices. The huge success of III-N materials is not mainly due to the intrinsic bulk material properties, but due to the heterojunction interface properties. In III-N heterostructures, the interfaces allow for the formation of n-channels and intrinsically provide extremely high carrier concentrations  $\geq 10^{13} \, cm^{-2}$  through polarization engineering without impurity doping [7]. The resulting two-dimensional electron gas (2DEG) at the heterojunction serves as the conductive channel.

The high electric breakdown field of GaN is a result of the wide bandgap of 3.44 eV at room temperature of the material and enables the application of high supply voltages on GaN-based devices, which is one of the two requirements for high-power device performance. The wide bandgap of the material also allows it to withstand high operating temperatures of  $300 \,^{\circ}$ C to  $700 \,^{\circ}$ C. Large drain currents of greater than  $1 \,\text{A/mm}$ , which are the secondary requirement for a power device, can be achieved due to the high electron mobilities of around  $2000 \,\text{cm}^2/\text{Vs}$  and high electron sheet densities of around  $1 \times 10^{13} \,\text{cm}^{-2}$ . These material properties clearly indicate why GaN is a serious candidate for next-generation microwave high-power/high-temperature applications.

Other important parameters that describe the quality of a semiconductor material are the relative permittivity or dielectric constant ( $\varepsilon_r$ ) and the thermal conductivity ( $\chi$ ). The thermal conductivity describes the ease of which heat is conducted in the material, and hence, the possibility to efficiently extract the dissipated power from the device. Materials with lower thermal conductivity typically lead to device degradation at elevated temperatures. Although III-V semiconductors have relatively moderate value of thermal conductivity ( $\chi$ ), GaN has a thermal conductivity of  $(1.3 W cm^{-1} K^{-1})$  which is comparable to that of Si  $(1.5 W cm^{-1} K^{-1})$ , which is the dominating semiconductor material of choice in the present electronics industry.

Property	GaN	Si	GaAs	$\mathbf{SiC}$	InP
Electron mobility, $\mu \ (cm^2/V.s)$	2000	1300	5000	260	5400
Dielectric constant, $\varepsilon_r$	9.5	11.4	13.1	9.7	12.5
Bandgap, $E_g \ (eV)$	3.4	1.1	1.4	2.9	1.35
Electric breakdown field, $E_c \ (kV/cm)$	3300	300	400	2500	500
Saturated electron drift velocity, $v_{sat} \ (\times 10^7 \ cm/s)$	2.2	1.0	1.0	2	1.0
$\frac{1}{\chi (W/cmK)}$	1.3	1.5	0.46	4.9	0.7

TABLE 1.1: Properties of GaN and competing semiconductor materials in power electronics.

Beyond the individual material properties, several figures of merit (FOM) have been proposed to benchmark the merit of different semiconductor materials for a given application, with the higher the figure of merit number the better the performance is likely to be. These figures of merit combine the most relevant material properties with respect to high-power and high-frequency applications into one number that represents a rough measure of the relative strengths of the other competing materials. One of the most commonly used is Johnson's Figure of Merit (JFOM), which gives an idea of material suitability for high-power applications at high frequencies and can be calculated using Equation (1.1) [8].

$$JFOM = \left(\frac{E_c v_{sat}}{2\pi}\right)^2 \tag{1.1}$$

where  $E_c$  is critical electric field and  $v_{sat}$  is saturation electron velocity.

Baliga's Figure of Merit (BFOM), on the other hand, was developed to compare the ultimate performance of field-effect transistors for low-frequency power-switching applications where conduction losses dominate [9]:

$$BFOM = \varepsilon_r \mu E_q^3 \tag{1.2}$$

where  $\mu$  is the electron mobility and  $\varepsilon_r$  is dielectric constant of the semiconductor.

Baliga's High-Frequency FOM (BHFOM) benchmarks these devices at high-frequency, where switching losses dominate [9]:

$$BHFFOM = \mu E_c^2 V_G^{1/2} / 2V_{BR}^{3/2}$$
(1.3)

where  $V_G$  is the gate drive voltage and  $V_{BR}$  is the breakdown voltage. The latter can be related to an experimental figure of merit involving the on-state specific resistance  $R_{on}$ of the device and its critical electric field,  $E_c$ :

$$FOM = V_{BR}^2 / R_{on} \equiv \varepsilon_r \mu E_c^3 / 4 \tag{1.4}$$

Other FOMs have been proposed for more specific cases. Amongst these, Keyes FOM considers thermal limitations due to transistor switching [9]:

$$KFOM = \chi \left(\frac{cv_{sat}}{4\pi\varepsilon_r}\right)^{0.5} \tag{1.5}$$

where  $\chi$  is thermal conductivity and c is the speed of light.

Finally, a Combined FOM (CFOM) was developed to simultaneously account for high-frequency, high-power and high-temperature performance [10]:

$$CFOM = \chi \varepsilon_r \mu v_{sat} E_c^2 \tag{1.6}$$

Table 1.2 compares several figures of merit for the possible high-power and high-frequency performance of GaN relative to other competing semiconductor materials. This table shows that GaN is an excellent candidate for high-frequency power applications. For GaN, the JFOM is about 270 to 480 times that of silicon (Si), about 135 to 240 times that of gallium arsenide (GaAs) and about 1.5 times that of silicon carbide (SiC) [9]. From these figures, it is clear that GaN offers much better high-power/high-frequency performance possibilities than Si, GaAs and SiC. The BFOM of GaN is about 17 to 34 times that of Si, about 1.5 to 2.5 times that of GaAs and about 3 times that of SiC [9]. No matter what FOM is used, wide-bandgap semiconductors offer better performance characteristics for high-frequency, high-power applications. The use

of wide-bandgap materials can be considered amongst one of the best solutions that can meet the requirements of modern day power electronics. Amongst wide-bandgap semiconductors, GaN-based materials are arguably the most matured ones in terms of wafer-size availability, technology and market prospects.

TABLE 1.2: Figures of merit of GaN and competing semiconductor materials in power electronics.

Property	GaN	Si	GaAs	SiC
Johnson's Figure of Merit, (JFOM)	270-480	1.0	2	324-400
Keyes Figure of Merit, (KFOM)	1.4	1.0	0.4	4.5-4.8
Baliga's Figure of Merit, (BFOM)	17-34	1.0	13	6-12
Baliga's High Frequency Figure of Merit, $(BHFFOM),$	86-172	1.0	10	57-76
Combined Figure of Merit, (CFOM)	108-290	1.0	4	275-310
$\begin{array}{c} T_{max}, \\ (^{\circ}C) \end{array}$	700	300	300	600

Other advantages of wide-bandgap semiconductor-based power devices compared with silicon-based power devices are as follows:

- Wide-bandgap semiconductor-based power devices can operate at high temperatures of up to 600 °C whilst silicon devices can only operate at a maximum junction temperature of 150 °C although a silicon on insulator (SOI) gate drive integrated circuit (IC) has been demonstrate to operate up to a temperature of around 200 °C [11].
- Due to their higher electric breakdown field, wide-bandgap semiconductor-based power devices have higher breakdown voltages; thus the first commercial SiC schottky diodes are already rated at 600 V compared to silicon schottky diodes which are commercially available typically at 300 V.

• Wide-bandgap semiconductor-based unipolar devices have lower on-resistances, as the on-resistance of a power device is inversely proportional to the cube of the electric breakdown as given by Equation (1.7) [9, 12]. Lower  $R_{on}$  means higher overall converter efficiency is attainable.

$$R_{on} = \frac{4(V_{BR})^2}{\varepsilon_r \mu(E_c)^3} \tag{1.7}$$

• Wide-bandgap devices such as SiC have a higher thermal conductivity (4.9 W/cm-K for SiC, as opposed to 1.5 W/cm-K for silicon). Therefore, wide-bandgap semiconductor-based power devices have a lower junction-to-case thermal resistance  $R_{th-jc}$ . This means that heat is more easily transferred out of the device, and thus the device temperature increase is slower.

#### 1.1 Brief History of GaN Technology

The history of GaN began during the early decades of the last century, when in 1930, Johnson et al. prepared gallium nitride by passing ammonia gas on metallic gallium at high temperatures of 900 °C - 1000 °C [13]. In 1938, Juza and Hahn synthesized GaN by passing ammonia (NH<sub>3</sub>) over liquid gallium at elevated temperatures. This method resulted in a powder consisting of small needles and platelets. Their purpose was to investigate the crystal structure and lattice constant of GaN [14]. However, it was only in 1968 that the first large area GaN epitaxially grown on sapphire substrates by Hydride Vapour Phase Epitaxy (HVPE) was demonstrated [15, 16].

After that discovery, a rapid progress was made in GaN technology leading to 1991 when Khan et al. reported the first evidence of two-dimensional electron gas (2DEG) formation at an  $Al_xGa_{1-x}N/GaN$  heterojunction grown by metal organic chemical vapour deposition (MOCVD) on sapphire [17]. The first GaN metal semiconductor field-effect transistor (MESFET) and heterostructure field-effect transistor (HFET) grown by metal organic chemical vapour deposition (MOCVD) on sapphire substrates were reported in 1993 and 1994, respectively by Khan et al. [18, 19].

Since the discovery by Khan et al. in 1994, excellent progress has been made in the development of GaN technology. In 2000, AlGaN/GaN HEMT technology was successfully transferred to silicon substrates by Kaiser et al. by growing the heterostructure using Metal Organic Chemical Vapour Deposition (MOCVD) [20].

Another important step in GaN technology was achieved by Tripathy et al. in 2012, by reporting the epitaxial growth, characterization, and device characteristics of crack-free AlGaN/GaN heterostructures on a 200 mm (8 inch) diameter Si(111) substrate [21].

Some of the most important highlights of the scientific research on GaN technology are listed below:

- 1. In 1930, Johnson et al. demonstrated the first synthesis of GaN [13].
- 2. In 1938, Juza and Hahn powdered GaN consisting of small needles and platelets in order to study the crystal structure and lattice parameters of GaN [14].
- 3. In 1959, Grimmeiss at al. conducted the first photo-luminescence measurement on small GaN crystals [22].
- 4. In 1969, Maruska et al. grew the first single crystal film of GaN, epitaxially grown on a centimetre-sized sapphire substrate using hydride vapour phase epitaxy (HVPE) [15].
- 5. In 1971, Pankove et al. fabricated the first GaN light emitting diode (LED) [23, 24].
- 6. In 1972, Maruska et al. fabricated the first hydride vapour phase epitaxy (HVPE) GaN LED using magnesium (Mg) as a p-type dopant, which emits at a wavelength of 430 nm (violet) [25].
- 7. In 1986, Amano et al. fabricated improved GaN films, grown by metal organic chemical vapour deposition (MOCVD) on sapphire substrates [26].
- 8. In 1989, Amano et al. was also the first to grow p-type conducting GaN films [27].
- 9. In 1991, Nakamura grew the first GaN on AlN buffer layer, and observed an improvement in surface morphology which was attributed to the prior deposition of the AlN buffer layer [28].
- 10. In 1991, Nakamura et al. also grew highly p-typed Mg-doped GaN films with GaN buffer layers which demonstrated the highest hole concentration and lowest resistivity reported to that date [29]. In the same year, Khan et al. made one of the most important discoveries in AlGaN/GaN heterojunctions to date when they reported the first evidence enhanced electron mobility in the form of two-dimensional electron gas (2DEG) formation at an Al<sub>x</sub>Ga<sub>1-x</sub>N/GaN heterojunction grown by MOCVD on sapphire [17].
- 11. In 1993, Khan et al. reported the first fabrication and characterization of a metal semiconductor field effect transistor (MESFET) based on single crystal GaN. The GaN layer was deposited over sapphire substrate using low pressure metal organic chemical vapour deposition [18]. Bykhovski et al. also showed during that same year that, strongly pronounced piezoelectric properties play a key role in GaN-AlN-GaN semiconductor-insulator-semiconductor (SIS) and related structures [30].

- 12. In 1994, Khan et al. also reported the first microwave performance of an Al-GaN/GaN heterostructure field-effect transistor(HFET) grown by MOCVD on a sapphire substrate [19].
- 13. In 1995, Ozgur et al. fabricated the first high transconductance normally-off GaN based modulation-doped field-effect transistors [31].
- 14. In 1997, Asbeck et al. quantified the influence of piezoelectric effect on the design and behaviour of III-V nitride heterostructure field-effect transistors (HFETs) grown by both MOCVD and molecular beam epitaxy (MBE) on sapphire and SiC substrates [32].
- 15. In 1998, Ren et al. fabricated and characterized a GaN metal-oxide semiconductor field-effect transistor (MOSFET) which demonstrates significantly reduced gate leakage and improved I-V characteristics at elevated temperatures [33].
- 16. In 1999, Yoshida et al. grew the first bipolar junction transistor using only GaN [34].
- 17. In 2000, Zhang et al. reported on the DC characteristics of the first *pnp* AlGaN/-GaN heterojunction bipolar transistor (HBT) [35].
- In 2001, Semond et al. reported on the growth of high-electron-mobility AlGaN/-GaN heterostructures on silicon (111) substrates by MBE using ammonia as the nitrogen source [36].
- 19. In 2006, Cai et al. fabricated a normally-off AlGaN/GaN HEMT which uses a fluoride-based plasma treatment to control the threshold voltages ( $V_{th}$ ) of the device [37].
- 20. In 2009, Uemoto et al. developed a normally-off GaN-based transistor using conductivity modulation, called a gate injection transistor (GIT). This device principle utilizes hole-injection from the p-AlGaN to the AlGaN/GaN heterojunction, which simultaneously increases the electron density in the channel, resulting in a dramatic increase of the drain current owing to the conductivity modulation [38].
- 21. In 2012, Tripathy et al. reported crack-free AlGaN/GaN heterostructures grown on 200 mm diameter (111)-oriented silicon substrates and presented an in-depth characterization study of epilayers grown by MOCVD [21].
- 22. In 2013, Zhang et al. demonstrated E-mode fluorinated MOS-HEMTs with a  $V_{th}$  higher than 3 V, no current collapse and long-term stability at 250°C using a dual-gate structure and an Al<sub>2</sub>O<sub>3</sub> gate oxide deposited by atomic layer deposition [39].

#### **1.2** Application Areas of GaN

The direct bandgap of GaN and its alloys enables the material to be used for both optical and electronics applications. At a temperature of 300 K, the bandgap of GaN is 3.44 eVwhich corresponds to a wavelength of around 360 nm in the near ultraviolet (UV) region of the optical spectrum [40, 41]. The relationship between the bandgap energy (E<sub>g</sub>) and wavelength is given by Equation (1.8):

$$Bandgap \ Energy \ (E_g) = \frac{hc}{\lambda} \tag{1.8}$$

where h is the Planck's constant, c is the speed of light,  $\lambda$  is the wavelength and 1  $eV = 1.6 \times 10^{-19} Joules$ 

Optical applications of GaN include high-brightness blue light emitting diodes (LEDs), backlighting for mobile phones and personal digital assistants (PDAs), and blue-violet laser diodes (LDs).

Figure 1.3 shows an overview of GaN-based micro-electronics applications. High frequency applications for GaN-based devices include microwave and millimetre wave amplifiers, RF amplifiers, space and radar electronics including local multipoint distribution systems, military applications inverter systems, digital radio and base station transmitters. Due to the high breakdown voltage of a HEMT, it can work at medium operating voltages like 42 V, hence reducing the cost of circuitry by eliminating the voltage stepdown circuit [5]. Power electronics applications for GaN-based HEMTs include electric and hybrid vehicles, electrical inverters, switched mode power supplies and motor drive circuits which operate in the voltage range of 600 V to around 1200 V [4, 42, 43].

#### **1.3** Gallium Nitride Power Semiconductors Market

Due to its vast addressable market, the GaN power semiconductors market is quickly expanding. Several transistors, diodes and rectifiers have been in the market since 2008, with extraordinary growth in the volume of power discretes (HEMTs, diodes, rectifiers and FETs) boosting the total revenue of the GaN power semiconductors market. The complete GaN power semiconductors industry shifted to a mass-production scenario in 2011 with the success and revenue potential drawing the focus of several power semiconductor market giants [44].

In 2012, the overall GaN market revenue (including both power and opto segments) accounted for less than 1% (\$340 million) of the total semiconductors market (\$34 billion). The GaN power semiconductors market reached \$12.60 million by the end of 2012. The



FIGURE 1.3: Electronic applications of GaN-based devices.

phenomenal growth rate of approximately 60 to 80% year-on-year is expected for the following years. The forecasted revenue for GaN power semiconductors is \$1.75 billion by end of 2022 at a compound annual growth rate (CAGR) of 63.78%. According to the Lyon, France-based market research and strategy consulting firm Yole développement, automakers will begin to adopt GaN power devices in inverters, dc-dc converters, and on-board chargers, generating a revenue of around \$150 million by 2020 as shown in Figure 1.4 [45]. Written in the bubbles is the main device voltage target for GaN, whilst the bubble size is related to Si device market as of 2013, most likely accessible to GaN. Apart from power semiconductors, GaN is predominantly used in optoelectronics for LEDs and laser diodes. The total GaN semiconductors (including both, power and optoelectronics) market revenue is expected to reach \$2.6 billion at a CAGR of 21.92% by 2022 [44].



FIGURE 1.4: Yole's report showing that automakers will begin to adopt GaN power devices in inverters, dc-dc converters, and on-board chargers, generating revenue of \$150 million by 2020.

#### 1.4 Research Aims and Objectives

Conventional GaN-based HEMTs are of depletion mode (D-mode) type but enhancement mode (E-mode) or normally-off devices would be preferable for power electronics since they simplify circuit design by enabling the exclusion of negative-polarity power supply and they also provide the highly desirable essential feature of fail-safe operation [46, 47].

Although a great deal of effort has been made to achieve normally-off AlGaN/GaN HEMTs, the current E-mode HEMT devices do not exhibit high enough threshold voltages for power switching applications (V<sub>th</sub> in the range of 3-5 V is preferred in order to prevent the mis-operation caused by noise) [48, 49]. Also, current techniques used to realise normally-off operation of these devices have their drawbacks, for example, gate recessed E-mode devices are not easily reproducible and repeatable due to the difficulties in controlling the etch rate to achieve uniform threshold voltage on all devices on a wafer. It has also been proved that physical etching damages the sample surfaces and therefore affects the device performance [50]. For E-mode devices realised using fluorine ion implantation, the plasma treatment normally induces damages and creates defects in the semiconductor material, and consequently degrades the carrier mobility.

This technique is also known to be unreliable as the implanted fluorine ions have the tendency to move around the GaN crystal lattice making the device very unstable [51].

The main aim of this work is therefore to develop an enhancement-mode AlGaN/GaN transistor technology with the following attributes: a positive threshold voltage (3 to 5 V), a low specific on-resistance ( $R_{ON}$ , approximately between  $0.5 \,\Omega mm^2$  to  $1 \,\Omega mm^2$ ), high drain current ( $I_{DS}$ , greater than  $1 \,A/mm$ ) and high breakdown voltage ( $V_{BR}$ , approximately between  $100 \,V$  to  $1500 \,V$ ).

The specific objectives of this research are to:

- Assess the suitability of a new AlGaN/GaN HEMT with a thin (3 nm) AlGaN barrier layer for normally off operation,
- Develop process modules for fabricating the normally-off AlGaN/GaN HEMT,
- Fabricate the normally-off AlGaN/GaN HEMTs using the developed process flow,
- Characterize the performance of the fabricated devices,
- Optimize the performance of the fabricated transistors.

### 1.5 Organization of Thesis

This thesis is divided into 7 chapters. This chapter, Chapter 1 gives a brief introduction of why wide-bandgap materials and gallium nitride (GaN) in particular have attracted so much attention in the power electronics industry. The chapter also gives a brief history of GaN technology and states some of its application areas. It then goes on to give a brief overview of the impact of GaN in the power semiconductor market. The main aims and objectives of this research project are then presented and discussed briefly.

Chapter 2 describes GaN material properties and polarization effects in the AlGaN/GaN heterostructure. In this chapter, a simulation of the sheet carrier concentration present in the conduction channel at an AlGaN/GaN interface depending on the thickness of AlGaN barrier layer or the aluminium content (x) in the AlGaN barrier is then carried out. The layer structure of basic AlGaN/GaN HEMTs and their fundamental operation is introduced. The growth techniques for gallium-based binary and ternary compounds and the choices of substrates for growing GaN are also described. The two main metal contacts, rectifying and Ohmic contacts are then introduced.

Chapter 3 gives an overview of the processing techniques used for fabricating GaN-based devices including etching techniques, lithography and surface passivation.

Chapter 4 reviews some of the current techniques which have been proposed for fabricating enhancement-mode AlGaN/GaN HEMTs and introduces some of their advantages and disadvantages.

Chapter 5 focuses on the development of the fabrication processes used to realise the normally-off GaN transistors in this project. It describes the fabrication processes used for single finger normally-off transistors with focus on the mask design and layout of the transistors. A schematic overview of the process flows used to fabricate the normally-off devices is provided.

Chapter 6 focuses on the fabrication and characterization of the enhancement-mode AlGaN/GaN devices which employ an ultrathin sub-critical 3 nm barrier layer and rely on an induced two-dimensional electron gas for operation. Devices on AlGaN/GaN materials grown on sapphire and silicon substrates are presented here. The results of the performance of these devices are analysed and discussed in this chapter.

Chapter 7 provides a summary of the research project as well as the discussion of potential future work that can be carried out to further improve and enhance the devices studied in this work.

The thesis concludes with appendices A, B and C which provide process details for the devices fabricated in this work, some semiconductor basics and GaN devices with silicon nitride gate dielectric, respectively.

### **1.6** Original Contribution to Research Field

During this research, the author of this thesis was able to:

- Assess the suitability of a new AlGaN/GaN HEMT with a thin (3 nm) AlGaN barrier layer for normally off operation by developing an analytical model of the dependence of polarization induced sheet carrier concentration on the thickness of AlGaN barrier layer or the aluminium content (x) in the AlGaN barrier,
- Develop process modules for fabricating a normally-off AlGaN/GaN HEMT,
- Fabricate the normally-off AlGaN/GaN HEMTs using the developed process flow,
- Characterize the performance of the fabricated devices by utilizing test and measurement tools including a probe station and a device parameter analyzer,
- Analyse and discuss the performance of the transistors,
- Make suggestions for future research work that could be carried out to improve the device performance.

### CHAPTER 2

# GaN HIGH ELECTRON MOBILITY TRANSISTORS

As mentioned in Chapter 1, one of the main advantages of GaN-based devices such as AlGaN/GaN high electron mobility transistors (HEMTs) is the formation of a twodimensional electron gas (2DEG) channel at the interface of the heterostructure. This 2DEG at the heterojunction serves as the conductive channel for large drain currents due to high electron mobility and high electron sheet charge density. In this chapter, the material properties and crystal structure of GaN is discussed, with focus on the polarization effects in the AlGaN/GaN heterostructure. The theory behind the formation of the 2DEG channel in the AlGaN/GaN interface is also discussed and an analytical model of the dependence of polarization induced sheet carrier concentration on the thickness of AlGaN barrier layer or the aluminium content (x) in the AlGaN barrier is presented. The typical structure of a GaN-based HEMT is also introduced including the fundamental operation of an AlGaN/GaN HEMT. Some of the techniques used to grow gallium-based compounds, and some substrate choices for growing GaN-based devices are introduced, followed by a brief discussion of the two main metal-semiconductors used in AlGaN/GaN HEMTs. The last section of the chapter gives a brief introduction to the transfer length method (TLM), also known as transmission line model, a technique used to characterize the quality of Ohmic contacts.

### 2.1 GaN Material System

In this section, the material properties and crystal structure of III-Nitride semiconductors, in particular for GaN, are discussed. The effects of polarization on the 2DEG channel formed in the AlGaN/GaN heterostructure will also be briefly described.

### 2.1.1 Material Properties and Crystal Structure of III-Nitride Semiconductors

Group III-Nitride semiconductors can be found in three common crystal structures; wurtzite, zincblende, and rock salt. The zincblende structure can also be found for GaN for thin films [52]. The rock salt phase is of no importance to electronic devices so far. At room temperature and atmospheric pressure, the wurtzite structure is the thermodynamically stable phase for GaN and consists of two inter-penetrating hexagonal close-packed lattices (HCP), which are shifted with respect to each other by  $(3/8)c_0$ where  $c_0$  is the height of the hexagonal lattice cell. The unit cell of the wurtzite lattice is hexagonal with a basis of four atoms, two nitrogen and two gallium atoms, see Figure 2.1. It contains six atoms and it is characterized by two lattice constants,  $a_0$  (3.18 Å) and  $c_0$  (5.18 Å) [53]. In this structure, growth is typically performed along the c-axis.



FIGURE 2.1: Wurtzite unit cell of GaN showing the lattice constants  $a_0$  and  $c_0$ .

The chemical bonds of III-nitride compounds such as GaN are predominantly covalent, which means that, each atom is tetrahedrally bonded to four atoms of the other type. There is also an ionic contribution of the bond which determines the stability of the structural phase due to the large difference in electronegativity of Ga and N atoms. There is no inversion symmetry in this lattice along the [0001] direction or *c*-axis, which by convention is the direction shown by the vector pointing from a Ga atom to the nearest neighbouring N atom. Lack of inversion symmetry means that, when defining an atom position on a close-packed plane with coordinates (x, y, z), it is not invariant
to the position (-x, -y, -z) since inversion results in replacement of group III atoms by nitrogen atoms and vice versa. As a result of the lack of symmetry, all atoms on the same plane at each side of a bond are the same. Hence, wurtzite GaN crystals have two distinct faces, commonly known as Ga-face and N-face, which correspond to the (0001) and  $(000\bar{1})$  crystalline faces as shown in Figure 2.2.

Figure 2.1 also shows the three parameters that define the wurtzite lattice. These are the edge length of the basal hexagon  $(a_0)$ , the height of the hexagonal lattice cell  $(c_0)$ , and the cation-anion bond length ratio  $(u_0)$  along the [0001] direction in units of  $c_0$ . The subscript "0" indicates that these values are those of the equilibrium lattice. In an ideal wurtzite crystal, the  $c_0/a_0$  ratio equals  $\sqrt{\frac{8}{3}} = 1.633$  and the value for  $u_0$  is 0.375. Because of the different metal cations, the bond lengths and the resultant  $c_0/a_0$ ratios of AlN, GaN, and InN are different. Table 2.1 shows an overview of these lattice parameters of wurtzite III-nitrides at 300 K [54]. From Table 2.1, it is clear that GaN is closest to the ideal wurtzite structure, followed by InN and AlN. This fact is very important because the degree of non-ideality is a significant factor in determining the strength of polarization in III-nitrides, which will be discussed further in Section 2.1.2.

Parameter	Ideal	AlN	$\operatorname{GaN}$	InN
$a_0,$ (Å)	_	3.112	3.189	3.54
$a_0,({ m \AA})$	_	4.982	5.185	5.705
$c_0/a_0$ [expected]	_	1.6010	1.6259	1.6116
$c_0/a_0$ [calculated]	1.633	1.6190	1.6336	1.6270
$u_0$	0.375	0.380	0.376	0.377

TABLE 2.1: Lattice parameters of wurtzite III-nitrides at 300 K.

### 2.1.2 Polarization Effects in the AlGaN/GaN Heterostructure

In the absence of external electric fields, the total polarization (P) of a GaN or AlGaN layer is the sum of spontaneous polarization  $P_{SP}$  and piezoelectric polarization  $P_{PE}$ . Spontaneous polarization arises from the lack of symmetry of the wurtzite crystal, whereas piezoelectric polarization, is a result of the stress and strain created due to the lattice mismatch of the GaN and AlN layers. Structural parameters play a significant role in determining spontaneous polarization, which explains the differences in



FIGURE 2.2: Atomic arrangement in Ga-face and N-face GaN crystals. Spontaneous polarization vector is also shown.

polarization of GaN and AlN [55]. The increasing non-ideality of the crystal structure from GaN to AlN increases the spontaneous polarization.

A standard AlGaN/GaN epitaxial film is grown along the [0001] axis as described in Section 2.1.1 and hence polarizations along that axis are the only ones considered. The spontaneous polarization along the c-axis is given by  $\mathbf{P}_{SP} = P_{sp} \hat{z}$ , whereas the piezoelectric polarization is given by the piezoelectric coefficients  $e_{33}$  and  $e_{31}$  as:

$$P_{PE} = e_{33}\varepsilon_z + e_{31}(\varepsilon_x + \varepsilon_y) \tag{2.1}$$

and  $\varepsilon_z = (c - c_0)/c_0$ , which is the strain along the c-axis. The in-plane strain is assumed to be isotropic and is given by  $\varepsilon_x = \varepsilon_y = (a - a_0)/a_0$ , where  $a_0$  and  $c_0$  are the equilibrium values of the lattice parameters. The relationship of the lattice constants of the wurtzite GaN is given by Equation (2.2).

$$\frac{c-c_0}{c_0} = 2\frac{C_{13}}{C_{33}}\frac{a-a_0}{a_0} \tag{2.2}$$

where  $C_{13}$  and  $C_{33}$  are the elastic constants.

Combining Equations (2.1) and (2.2), the net polarization along the c-axis can be calculated as:

$$P_{PE} = 2\frac{a - a_0}{a_0}(e_{31} - e_{33})\frac{C_{13}}{C_{33}}$$
(2.3)

Since  $[e_{31}-e_{33}(C_{13}/C_{33})] < 0$  for AlGaN over the whole range of compositions, the piezoelectric polarization is negative for tensile and positive for compressive strain barriers respectively [54].

The spontaneous polarization for Ga-face GaN and AlGaN are negative because the direction vector points towards the substrate as shown in Figure 2.3 [55]. Similarly for N-face materials, spontaneous polarization direction vector points away from the substrate. For both Ga-face and N-face materials, the piezoelectric and spontaneous polarizations are parallel in case of tensile strain and anti-parallel for compressively strained barrier layers which is also shown in Figure 2.3.



FIGURE 2.3: Directions of spontaneous and piezoelectric polarization in Ga and N-face AlGaN/GaN heterostructures.

The polarization induced charge density in space is given by:

$$\rho_p = \nabla.\mathbf{P} \tag{2.4}$$

The polarization sheet charge density( $\sigma$ ) at an abrupt interface of an AlGaN/GaN or GaN/AlGaN heterostructure is given by [55, 56]:

$$\sigma = P(bottom) - P(top) \tag{2.5}$$

$$\sigma = \{P_{SP}(bottom) + P_{PE}(bottom)\} - \{P_{SP}(top) + P_{PE}(top)\}$$
$$= \{P_{PE}(bottom) - P_{PE}(top)\} + \{P_{SP}(bottom) - P_{SP}(top)\}$$
$$= \sigma(P_{PE}) + \sigma(P_{SP})$$
(2.6)

However, if the polarization induced sheet charge density is positive  $(+\sigma)$ , free electrons will try to compensate for the charges induced by polarization and vice versa. These charges will accumulate at the heterojunction of the barrier and the buffer layer to create a Two Dimensional Electron Gas (2DEG). In the case of Ga-face structures with AlGaN on top of GaN, the polarization induced sheet charge density is positive as shown in Figure 2.3a.

If the heterostructure is grown pseudomorphically, the piezoelectric polarization of the tensile strained AlGaN barrier layer will increase the difference between the AlGaN polarization and the GaN polarization and in turn increase the sheet charge density  $(+\sigma)$  and the electron confinement in the 2DEG. Alternatively for N-face AlGaN/GaN heterostructures, the spontaneous and piezoelectric polarizations have opposite directions (points away from the substrate) to Ga-face AlGaN/GaN structures (points towards the substrate) resulting in a negative polarization induced sheet charge density  $(-\sigma)$  as shown in Figure 2.3d and 2.3e. The negative sheet charge density assists to confine holes at the heterojunction, creating a "Two Dimensional Hole Gas" (2DHG).

In order to calculate the amount of the polarization induced sheet charge density ( $\sigma$ ) at the AlGaN/GaN and GaN/AlGaN interfaces in dependence of the Al-content x of the  $Al_xGa_{1-x}N$  barrier, the following set of linear interpolations between the physical properties of GaN and AlN are used [55, 56]:

Lattice constant:

$$a(x) = (-0.077x + 3.189)10^{-10}$$
 m, (2.7)

Elastic constants:

$$C_{13}(x) = (5x + 103)$$
 GPa, (2.8)

$$C_{33}(x) = (-32x + 405)$$
 GPa, (2.9)

Piezoelectric constants:

$$e_{31}(x) = (-0.11x - 0.49) \quad C/m^2,$$
 (2.10)

$$e_{33}(x) = (0.73x + 0.73)$$
 C/m<sup>2</sup>, (2.11)

Spontaneous polarization:

$$P_{SP}(x) = (-0.052x - 0.029)$$
 C/m<sup>2</sup>. (2.12)

The total amount of polarization induced sheet charge density for an undoped  $Al_xGa_{1-x}N/GaN$  heterostructure can then be calculated by using Equations 2.3, 2.6 and 2.12, which is given by Ambacher et al. [55, 56]:

$$|\sigma(x)| = |P_{PE}(Al_x Ga_{1-x}N) + P_{SP}(Al_x Ga_{1-x}N) - P_{SP}(GaN)|$$
(2.13)

$$|\sigma(x)| = \left| 2\frac{a(0) - a(x)}{a(x)} \left\{ e_{31}(x) - e_{33}(x)\frac{C_{13}(x)}{C_{33}(x)} + P_{SP}(x) - P_{SP}(0) \right\} \right|$$
(2.14)

The magnitude of the polarization induced sheet charge density ( $\sigma$ ) for both Ga-face or N-face heterostructures is always the same but the sign changes. All of the devices used in this research employ Ga-face GaN.

### 2.1.2.1 Simulation of 2DEG Sheet Density in GaN-Based HEMTs

Figure 2.4 shows the conduction energy band diagram of an AlGaN/GaN structure. The bandgap offset at the AlGaN/GaN interface,  $\Delta E_c$ , forms a quantum well in the GaN under the AlGaN barrier layer as shown in Figure 2.4. It is also shown in the diagram that the electrons in the 2DEG channel are provided by the unintentionally doped barrier layer. The electrons confined in a 2-dimensional conduction channel have high mobility and high electron density, which make HEMTs promising for high frequency and high power applications.



FIGURE 2.4: Conduction energy band of an AlGaN/GaN structure.

Since the total charge must be neutral, the free electrons accumulate in the quantum well to compensate the high positive polarization induced sheet charge at the AlGaN/GaN interface for Ga-face material. The electrons in the quantum well form a two dimensional electron gas (2DEG) channel. The density of the 2DEG is usually around  $1 \times 10^{13} \text{ cm}^{-2}$  with mobility of around  $2000 \text{ cm}^2/\text{V.s}$  at room temperature [30, 57]. The 2DEG sheet carrier concentration is given by [55, 56]:

$$n_s(x) = \frac{+\sigma(x)}{e} - \left(\frac{\epsilon_0 \epsilon(x)}{de^2}\right) \left[e\phi_b(x) + E_F(x) - \Delta E_c(x)\right]$$
(2.15)

where, d is the width of the  $Al_xGa_{1-x}N$  barrier,  $e\phi_b$  is the schottky-barrier of the gate contact,  $E_F$  is the Fermi level with respect to the GaN conduction-band-edge energy, and  $\Delta E_c$  is the conduction band offset at the AlGaN/GaN interface.

To determine the sheet carrier concentration from the polarization induced sheet charge from Equation 2.15, the following approximations were used [56]:

Dielectric constant:

$$\epsilon(x) = -0.3x + 10.4\tag{2.16}$$

Schottky barrier:

$$e\phi_b = (1.3x + 0.84)$$
 eV (2.17)

Fermi energy:

$$E_F = E_0(x) + \frac{\pi\hbar^2}{m^*(x)} n_s(x)$$
(2.18)

where the ground sub-band level of 2DEG is given by [56]:

$$E_0(x) = \left\{ \frac{9\pi\hbar e^2}{8\epsilon_0 \sqrt{8m^{*(x)}}} \frac{n_s(x)}{\epsilon(x)} \right\}^{\frac{2}{3}}$$
(2.19)

with the effective electron mass,  $m^*(x) \approx 0.22m_e$ 

Band offset:

$$\Delta E_C = 0.7[E_q(x) - E_q(0)] \tag{2.20}$$

where the band gap of  $Al_xGa_{1-x}N$  is given by:

$$E_g(x) = xE_g(AIN) + (1-x)E_g(GaN) - x(1-x)1.0$$
 eV, (2.21)

$$= x6.13 \text{ eV} + (1-x)3.42 \text{ eV} - x(1-x)1.0 \text{ eV}$$
(2.22)

From Equation (2.15), it can be seen that the sheet concentration is dependent on the percentage of aluminium (Al) composition in the  $Al_xGa_{1-x}N$  barrier layer and/or the thickness of the barrier layer. To further understand how the 2DEG sheet carrier concentration can be controlled by the percentage of aluminium in the AlGaN alloy and the thickness of the AlGaN barrier, we used the Wolfram Mathematica 9 software to compute Equation (2.15), and generated an analytical model of the expected sheet charge concentration against the thickness of AlGaN barrier layer and the aluminium content in the  $Al_xGa_{1-x}N$  alloy.

Figure 2.5 shows a graphical representation of the calculated sheet carrier concentration of the 2DEG sheet carrier density against different alloy compositions for various Al-GaN barrier thickness. It can be seen from the graph that, if for example, a constant barrier thickness of 30 nm is considered, the sheet carrier concentration,  $n_s$  will be approximately  $0.85, 1.5, 2 \times 10^{13} \, cm^{-2}$  for alloy compositions of x = 20%, 30% and 40%, respectively. If the thickness of an AlGaN barrier of alloy composition of x = 25% is increased from 10 nm over 20 nm to 30 nm, the sheet carrier concentration is increased from approximately 0.75 over 1.1 to  $1.2 \times 10^{13} \, cm^{-2}$ .

Figure 2.6 shows the variations of total polarization induced sheet charge carrier concentration of the 2DEG sheet carrier density when the AlGaN barrier thickness is varied for several alloy contents. From the graph, it can be seen that, for example, to achieve a sheet carrier concentration of  $1 \times 10^{13} \, cm^{-2}$ , only approximately  $0.5 \, \text{nm}, 2.2 \, \text{nm}, 4 \, \text{nm}, 7 \, \text{nm}$ , and 20 nm of AlGaN barrier thickness with alloy contents of x = 100 %, 60 %, 45 %, 35 %, and 25 \%, respectively, is required.



FIGURE 2.5: Total sheet carrier concentration of the 2DEG formed at the AlGaN/-GaN interface versus different alloy compositions for various AlGaN barrier layer thickness.

Conventional AlGaN/GaN devices employ approximately 20 - 25% Al content in the AlGaN barrier of thickness of approximately 20 nm [5, 58, 59]. Due to the presence of 2DEG at the AlGaN/GaN interface to allow for conduction even when no gate voltage is applied, these AlGaN/GaN HEMTs are therefore depletion-mode devices. The graph in Figure 2.6 also shows that there is no 2DEG formation in the GaN channel below a certain AlGaN barrier layer thickness. For example, at a barrier thickness of less than approximately 6 nm for x = 15%, 4 nm for x = 25%, and 3 nm for x = 35%, there is no presence of 2DEG in the GaN channel to allow for conduction. This barrier thickness at which there is very little or no electrons present to form a 2DEG channel is known as the "critical barrier thickness" of the barrier layer. Therefore, in order to realize enhancement-mode devices, it is crucial for the AlGaN barrier layer to be below the critical barrier thickness required for the formation of a 2DEG channel in the GaN channel. It will be demonstrated in subsequent chapters that this is indeed one of the main methods of achieving enhancement-mode operation.

The aluminium content in AlGaN barrier layers to date usually range between 15% to 40% for high quality HFETs. The reasons for this limitation are because, for barrier layers with x < 15%, the conduction band offset becomes too small ( $\Delta E_c < 0.28 \text{ eV}$ ),



FIGURE 2.6: Variations of total polarization induced sheet carrier concentration of the 2DEG formed at the AlGaN/GaN interface versus various AlGaN barrier thickness for different alloy compositions.

resulting in bad confinement of the polarization induced electron sheet charge. For barrier layers with x > 40%, the high lattice and thermal mismatch between the GaN buffer and the barrier layer cause a high density of structural defects in the AlGaN, and rough interfaces which limit the mobility of the 2DEG [55, 56, 60, 61]. All materials used in this project have an AlGaN barrier layer with an aluminium content of x = 25%.

# 2.2 The AlGaN/GaN HEMT

### 2.2.1 Fundamentals of HEMTs

As already defined earlier, HEMT stands for High Electron Mobility Transistor. There are similar forms of this structure including HFET

(Heterostructure Field Effect Transistor), MODFET (Modulation-Doped Field Effect Transistor), MISHEMT (Metal Insulator High Electron Mobility Transistor) amongst others. The first generation of HEMT structure was constructed in a lattice-matched GaAs-based AlGaAs/GaAs system [62] which has been widely studied and used in RF, microwave and millimetre wave applications.

The fundamental characteristic of the HEMT structure is the conduction band offset between the materials which construct the barrier and channel layers. The barrier layer has a higher conduction band than the channel layer. Due to the large conduction band discontinuity, the electrons diffusing from the large bandgap AlGaN into the smaller bandgap GaN form a Two-Dimensional Electron Gas (2DEG) in the triangular quantum well at the interface which is the hallmark of a HEMT. In a HEMT, by placing a Schottky barrier gate on the barrier layer, the 2DEG sheet charge concentration can be controlled by applying an appropriate bias.

In 1993, Khan et al. demonstrated the first AlGaN/GaN MODFET (with n-doped barriers), with a transconductance of 23 mS/mm and 2DEG mobility of 563 cm<sup>2</sup>/Vs at 300 K [63]. They also reported the first microwave results with current gain cut-off frequency  $(f_T)$  of 11 GHz and maximum frequency of oscillation  $(f_{MAX})$  of 14 GHz [19]. In the early stages, the MODFETs exhibited very low transconductances and relatively poor frequency response. With improvements in the materials quality, however, the transconductance, current capacity and drain breakdown voltage were all increased to the point that GaN based MODFETs were used in the arena of high-power RF devices. One of the highest power density achieved for  $0.45 \times 125 \,\mu$ m GaN MODFET was 6.8 W/mm at 10 GHz and associated gain of 10.65 dB [64]. The operation temperature was pushed to 750 °C by using a thermally stable Pt/Au gate contact [65].

### 2.2.2 Operation of an AlGaN/GaN HEMT

An AlGaN/GaN HEMT is used to either process data or power (switch/amplifier) as a discrete device or part of an integrated circuit. Figure 2.7 illustrates the output characteristics of a depletion mode HEMT. It should be noted that this figure is for illustrative purposes only and will only be used to describe the operation of the device. The most common way to bias a HEMT is in the common source configuration as illustrated in the inset of Figure 2.7. The gate electrode is the input whilst the drain electrode is the output and the source is the common terminal. The gate's input signal acts as the control signal of the device and has the ability to switch the device on and off. Switching off a depletion mode device requires the application of a negative voltage to deplete the channel of electrons which will result in a highly resistive channel where no current will flow, this is known as "pinch-off". The expression which relates the number of carriers in the channel to the applied bias voltage is given by (modelling the 2DEG metal gate as a capacitor) [66]:

$$n_s = \frac{\varepsilon_{AlGaN}}{q(d_{AlGaN} + \Delta d)} (V_{GS} - V_{TH})$$
(2.23)

where,  $n_s$  is the charge density per unit area of the 2DEG,  $d_{AlGaN}$  is the thickness of the AlGaN barrier layer,  $\Delta d$  is the effective distance of the 2DEG from the heterointerface,  $V_{GS}$  is the gate bias voltage, and  $V_{TH}$  is the threshold voltage



FIGURE 2.7: Typical IV characteristics of a HEMT. The inset shows a common source configuration for a transistor.

When  $V_{GS} = V_{TH}$ , the term  $(V_{GS} - V_{TH})$  in Equation (2.23) will be equal to zero, resulting in n<sub>s</sub> being zero and the device shall be off. When  $V_{GS} = 0 V$ , the channel will naturally be very densely populated with electrons and the application of a drain to source voltage will cause current flow between the source and drain of the HEMT.

For low drain voltages, i.e.  $V_{DS} < V_{GS} - V_{TH}$ , the device is said to be operating in the linear regime where the electron velocity in the channel is proportional to the applied electric field and so the current will increase with this field. The current flowing between the source and drain at this point is given by:

$$I_{DS} = qn_s v_{eff} W_G \tag{2.24}$$

where,  $v_{\rm eff}$  is the effective velocity of the electrons in the channel, and  $W_{\rm G}$  is the gate width

The velocity of electrons in the channel depends on their mobility and the electric field applied and is given by the relationship [67]:

$$v = \mu_n E \tag{2.25}$$

where,  $\mu_n$  is the electron mobility, and E is the applied electric field

The mobility of electrons in the 2DEG of an AlGaN/GaN HEMT is affected by carrier scattering through imperfections and dislocations in the semiconductor crystal and alloy disorder scattering [68]. According to Equation (2.25), the electron velocity increases with the applied electric field and so using this value in Equation (2.24), the current is shown to increase linearly for low fields when  $V_{DS} < V_{GS} - V_{TH}$ . Increasing the applied electric field so that the drain bias is  $V_{DS} > V_{GS} - V_{TH}$ , the velocity begins to saturate and becomes independent of the applied electric field. As the drain bias is increased such that  $V_{DS} > V_{GS} - V_{TH}$ , the lateral bias beneath the gate (due to drain bias) begins to pinch the channel off at the drain end of the gate. This continues until a point where the flow of electrons in the channel is constricted and limits the amount of electrons which can flow to the drain contact. At this point the device moves into what is known as the saturation regime and any further increase in the drain bias does not result in an increase of current. This region of device operation is analogous to a junction field-effect transistor (JFET) where the depletion region intrudes into the channel near the drain when the drain bias is increased. The drain current here can be expressed by:

$$I_{DS} = \frac{\varepsilon_{AlGaN} v_{sat} W_G}{(d_{AlGaN} + \Delta d)} (V_{GS} - V_{TH})$$
(2.26)

where,  $v_{sat}$  is the saturated electron velocity

An increase/decrease of the gate width  $W_G$  would directly result in an increase/decrease of current since more/less electrons pass through a specific region at any given time.

### 2.2.3 GaN-Based HEMT Layer Structures



FIGURE 2.8: A basic AlGaN/GaN HEMT.

A basic AlGaN/GaN HEMT shown in Figure 2.8 consists of an AlGaN barrier layer usually around 20 nm thick grown on a GaN channel layer usually around  $3 \mu m$  thick. These device structures are usually grown on sapphire, semi-insulating silicon Carbide (SiC), or Si substrates.

The substrates generally have different lattice constants from the GaN layer (more details is given in Section 2.4), which means that there is a lattice mismatch between the GaN epilayers and the substrate [59, 69, 70]. This mismatch induces strain which results in dislocation or defect areas. Due to this, a thin neucleation layer of GaN or AlN is grown on top of the substrate, on top of which the channel and barrier layers are subsequently grown [5].

A device structure of the typical dimensions given here are of depletion-mode or normally-on mode, meaning this device requires a negative bias voltage in order to be turned off. The band diagram for this normally-on AlGaN/GaN device structure is shown in Figure 2.9.



FIGURE 2.9: A basic AlGaN/GaN HEMT with corresponding energy band diagram.

# 2.3 Growth Techniques for Gallium-Based Binary and Ternary Compounds

The growth of good quality GaN based binary and ternary heterostructures is important for the production of high performance HEMT devices. Historically, hydride vapour phase epitaxy (HVPE) and its closely related technique, halide vapour phase epitaxy, have played an important role in the development of semiconductor material systems. The most popular and common procedures used to grow high quality GaN epitaxy are Metal Organic Chemical Vapor Deposition (MOCVD) and Molecular Beam Epitaxy (MBE).

### 2.3.1 Hydride Vapour Phase Epitaxy

The first technique used to grow epitaxial GaN layers was the hydride vapour phase epitaxy (HVPE). This technique remained the most commonly used method until the early 1980s. During this technique, the gallium monochloride (GaCl) precursor is synthesized inside a reactor by the reaction of hydrochloric acid (HCl) with liquid gallium (Ga) at temperatures between 750 °C and 900 °C. Then the GaCl, which is transported to usually foreign substrates, e.g. sapphire, silicon, gallium arsenide, or silicon carbide, reacts with ammonia (NH<sub>3</sub>) at 900 °C – 1100 °C to form a GaN film. Finally, the foreign substrate can be removed by techniques such as laser ablation. HVPE has a really high growth rate (typically 100  $\mu$ m/hr), which has the potential to facilitate large area, thick and low defect density GaN quasi-substrates for subsequent growth by other epitaxial techniques such as MOCVD or MBE.

### 2.3.2 Metal Organic Chemical Vapour Deposition

Metal organic chemical vapour deposition (MOCVD) is a technique that involves a dynamic flow in which gaseous reactants pass over a heated substrate and react chemically to form a semiconductor layer. The metal organic compounds are transported by a carrier gas, H<sub>2</sub>, to a heated substrate which reacts with Hydrides to form semiconductor films. The procedure involves the introduction of TriMethyl-Gallium (TMGa) and Ammonia gas simultaneously into the reaction chamber as shown in Equation (2.27) with a substrate heated to temperatures usually in the range of 800 °C ~ 1000 °C [71].

$$(CH_3)Ga(g) + NH_3(g) = GaN(s) + 3CH_3H(g)$$
(2.27)

However, for the growth of ternary compounds like AlGaN, InGaN, etc., Trimethyl-Aluminium or Trimethyl-Indium simultaneously are generally reacted with Trimethyl-Gallium and Ammonia at elevated temperatures as shown in Equation (2.28).

$$x(CH_3)Al + (1-x)(CH_3)_3Ga + NH_3 \rightarrow Al_xGa_{1-x}N + \dots$$
 (2.28)

MOCVD is currently used to grow high quality GaN epitaxial wafers for commercial purposes such as HEMTs, injection laser diodes and high brightness blue LEDs.

### 2.3.3 Molecular Beam Epitaxy

Molecular beam epitaxy (MBE) is one of the modern techniques used to grow single crystals. The main advantages of this technique is that it makes it possible to grow high quality layers, and it is easy to control the doping, concentration and thickness of the layer grown due to its slow deposition rate. In conventional solid-source MBE, ultra-pure elements like Aluminium, Gallium, Indium, etc. are heated in effusion cells to sublimate and allowed to condense on a substrate where the elements react with each other to form thin film layers. MBE growth chamber temperatures vary in the range between 600 °C and 900 °C. A computer controls the effusion cell shutters allowing precise control of the thickness of each layer [72, 73].

# 2.4 Substrate Choice for AlGaN/GaN HEMTs

The most important physical properties of the substrate materials under consideration for III-N devices include lattice-mismatch, thermal conductivity, coefficient of thermal expansion (CTE) and isolation. Table 2.2 presents an overview of the most important physical properties of the substrate materials under consideration for III-N devices.

### 2.4.1 Silicon Carbide Substrates (SiC)

Semi-insulating SiC is one of the most attractive substrate materials for electronic applications, due to favourable combination of lattice-mismatch, isolation, and thermal conductivity [69, 70, 74, 75]. It is available with increasing quality and diameters of up to 4 inch, although recently, both IQE and RFMD have both produced 6 inch GaN on SiC wafers [76–78]. Both polytypes 4H- and 6H-semi-insulating SiC are used. Most of the outstanding electronic device results have been reported on semi-insulating SiC materials, e.g. [79]. Currently, the greatest amount semi-insulating SiC material is produced by Cree [74]. There is a number of other vendors such as II-VI and Toyota. Despite the great advantages, SiC is a relatively expensive substrate on the market and as a result of this, cheaper alternatives are preferred.

### 2.4.2 Sapphire Substrates

Sapphire is a promising material due to the availability of cost-effective 2-4 inch diameter substrates. Recently, however, 6 and 8 inch substrates have been successfully produced by Rubicon Technology [80, 81]. NTT Advanced Technology Corporation have also recently grown 6 inch sapphire substrates [82]. It is widely used for electronic devices and despite the lattice-mismatch, it has displayed very impressive power densities of  $\geq$  $12 Wmm^{-1}$  and high pulsed output powers have been achieved [83, 84]. These substrates

Property	GaN	6H SiC	6H s.i. SiC	Sapphire	Silicon	AlN
$\begin{array}{c} \text{Lattice} \\ \text{Constant} \\ (\mathring{A}) \end{array}$	3.189	3.08	3.08	$4.758/\sqrt{3}$	5.4301	3.112
Mismatch to GaN (%)	0	3.4	3.4	13	17	1
Thermal Conductivity, $\kappa_L$ $(Wcm^{-1}K^{-1})$	1.3	4.9	3.7	0.5	1.5	2.0
Isolation $(\Omega cm)$	$\geq 10^{9}$	$\geq 10^{11}$	$\geq 10^{11}$	-	$1-3 \times 10^4$	$\geq 10^{12}$
Coefficient of Thermal Expansion, CTE $(10^6 K^{-1})$	5.9	4.2	4.2	7.5	3.59	4.2

TABLE 2.2: Basic properties of substrates and III-N semiconductor materials

have poor thermal conductivity but have been commonly used in device prototyping and also for GaN-based LEDs.

### 2.4.3 Silicon Substrates

Due its economic importance and widespread use, silicon is the most important semiconductor and substrate material; thereby making the use of silicon as a substrate desirable, mostly for cost efficiency reasons. Silicon wafers are available for large wafer diameters (from 8 inch up to 12 inch), whilst there is also a large pre-existing siliconbased infrastructure (foundries) for manufacturing silicon wafers [85]. The use of silicon however, introduces some critical issues and engineering challenges for III-N devices. Some difficulties of using silicon substrates include the difference in the thermal expansion coefficients between the substrate and the GaN semiconductor material, especially for high-power operation and the impact of the lattice mismatch relative to GaN with respect to material quality, defects and device reliability [86].

### 2.4.4 GaN and AlN Substrates

Native GaN and AlN substrates have been recently developed and used for electronic applications, although they are most important for optoelectronic applications. These substrates face increasing attention due to the expected improvement in electronic device reliability [87]. AlN is highly resistive and has a better thermal conductivity than GaN. Also, AlN compared to all other non-native substrates has the lowest lattice mismatch.

# 2.5 Metal-Semiconductor Contacts

Metal-semiconductor contacts are a vital part of a HEMT as they provide a mechanism of carrier transport to and from the semiconductor and metal contact of the device. There are generally two types of metal-semiconductors which are used in a HEMT. These are Schottky contact; which is mainly used as the gate electrode, and Ohmic contact; which is generally used as the source and drain electrodes. Both Ohmic and Schottky contacts are of critical importance with respect to device functionality, performance and long term reliability. This section gives a brief description of the two types of metal-semiconductor contacts used in an AlGaN/GaN HEMT.

### 2.5.1 Schottky Contact

The main transport mechanism in rectifying contacts is thermionic emission which allows electrons to move over the schottky barrier. For wide bandgap materials such as GaN, the barrier height is strongly influenced by the work function of the contact metal. For GaN based semiconductors, metals with high workfunctions are generally used to fabricate Schottky contacts [88, 89].

Table 2.3 shows various metals and their corresponding workfunctions. It can be seen from this table that Nickel, Platinum and Gold are the best metals for Schottky contacts because they have very high workfunctions compared to other metals on the list. However, Platinum suffers from adhesion problems when in contact with GaN and the presence of Gold at the semiconductor interface is responsible for creating high leakage current at the gate [90]. Additionally, it is known from experiments that the adhesion of Gold (Au) on GaN and on SiN isolation layers can be critical [91]. Based on that, Nickel (Ni) is one of the most widely used Schottky contact metals for GaN based semiconductors. The metal scheme generally used for gate electrode is Ni/Au 50/150 nm. The Au layer prevents the oxidation of the Ni layer.

In a HEMT, the gate contact is a Schottky contact. A voltage applied to the gate modulates the 2DEG carrier density at the AlGaN/GaN interface, and so modulates

Metal	Workfunction $(eV)$		
Gallium (Ga)	4.20		
Silver (Ag)	4.26		
Aluminium (Al)	4.28		
Titanium (Ti)	4.33		
Silicon (Si)	4.52		
Molybdenum (Mo)	4.60		
Gold (Au)	5.10		
Nickel (Ni)	5.15		
Platinum (Pt)	5.65		

TABLE 2.3: Various metals and their workfunctions

the drain to source current. For conventional HEMTs, the gate has a negative bias (is not forward biased).

### 2.5.2 Ohmic Contact

An Ohmic contact is defined as one in which there is unimpeded transfer of majority carriers from one material to another, i.e., the contacts do not limit the current. The contact resistance between the metal and semiconductor is dependent on the doping concentration of the semiconductor, engineering of the interface structures and unintentionally formed interface states. It is very important to form very low contact resistance,  $R_c$ , for Field-Effect Transistors (FETs) in order to achieve excellent DC and RF performances. It is generally very difficult to realize Ohmic contacts with very low  $R_c$  in wide bandgap materials due to the low workfunction and hence small electron affinity of such materials; meaning that the barrier height ( $\Phi_B$ ) of the metal/semiconductor is generally higher [92]. This is usually the case in  $Al_xGa_{1-x}N$  (3.4 eV at x = 0 to 6.2 eV at x = 1) especially as the value of x is increased.

In general, there are two ways of overcoming the Ohmic contact issues on  $Al_xGa_{1-x}N/GaN$ :

- 1. Forming a low barrier height at the metal/semiconductor interface in order to allow the thermionic emission effect, and/or
- 2. Forming a very thin barrier at or near the bottom of the conduction band to allow thermally excited electrons to be tunnelled directly.

In order to realize high performance  $Al_xGa_{1-x}N/GaN$  HEMTs, the Ohmic contacts must have very low resistance, excellent edge definition, smooth surface morphology, as well as high mechanical and thermal stability. Most  $Al_xGa_{1-x}N/GaN$  heterostructures use Ti/Al metallization based Ohmic contacts [93]. The most frequently reported schemes are the Ti/Al/Ni/Au, Ti/Al/Pt/Au and Ti/Al/Ti/Au metal schemes. Quite a number of researchers also tend to use Molybdenum (Mo) instead of Nickel (Ni), Platinum (Pt) or Titanium (Ti) as a diffusion layer because it can give low contact resistance of around 0.2  $\Omega$ mm with excellent edge acuity and improved surface morphology [94–97]. Each metal in the Ohmic contact stack has its own purpose [95]:

- 1. Titanium serves as an adhesion layer to provide good stability. It dissolves the native oxide on the AlGaN surface and creates nitrogen vacancies by reacting with nitrogen atoms to form TiN in the AlGaN layer, which in turn acts as n-type donor sites, increasing the doping concentration, and thus, enabling electrons to tunnel through the thinner metal/semiconductor barrier with very little/no resistance.
- 2. Aluminium reacts with Ti to form an  $Al_3Ti$  layer which prevents the underlying Ti layer from oxidizing. It also serves as a diffusion barrier for the metals on top as they form schottky barriers.
- 3. Nickel, Platinum, Titanium, or Molybdenum then prevents the Al from mixing with the Au as the reaction of such a mix forms a highly resistive layer also known as 'purple plague'.
- 4. Gold is used as contact with the outside world due to its high conductivity.

The preferred transport mechanism in Ohmic contacts is known as field emission [98, 99]. During field emission, the carriers tunnel through the potential barrier. Field emission takes place when the depletion layer is sufficiently narrow as a result of high doping concentration in the semiconductor. It occurs when carriers do not have enough energy to cross the barrier thermionically, but are able to tunnel through the barrier allowing current flow from either the semiconductor to metal or vice versa and hence it is independent of the system temperature. More basics on metal semiconductor junctions are described in Appendix B for completeness.

#### 2.5.3 Transfer Length Method Structures

Transfer Length Method (TLM), also known as Transmission Line Model [100–102], is used for assessing the quality and performance of Ohmic contacts. This technique was introduced by Reeves and Harrison and a detailed analysis is available in Ref. [100]. The two main methods of performing these measurements use either linear TLM structures or circular TLM structures.

### 2.5.3.1 Linear TLMs

Figure 2.10 shows a schematic diagram of a TLM pattern on a doped semiconductor material with Ohmic pads prepared for TLM analysis. In this case, metal pads, of finite width, w, and length, s, are deposited at linearly increasing pad spacing, L, such that  $L_1 < L_2 < L_3$ . The sample is then mesa-etched usually to the semi-insulating substrate. This is done in order to isolate columns of the conductive epitaxial layer, thereby restricting current flow within the mesa height, d.



FIGURE 2.10: A semiconductor material with Ohmic contact pads for TLM characterization.

The most common method for characterizing TLMs is the four-point probe method as shown in Figure 2.10. This method is also suitable for measuring the resistivity of grown layers that are electrically isolated from the substrate. The four probes are equally separated by a distance,  $L_p$ . A constant current is passed between two adjacent probes; a second set of probes is then used to measure the voltage drop using a voltmeter, enabling the total resistance between the pads to be obtained. A separate current source and voltmeter are preferred to single ohm meter because of the latter's relatively low impedance which may give rise to inaccuracies. The current flows from the measuring probes through the metal contact, across the metal-semiconductor region, through the semiconductor transmission line ( $L_1, L_2$ , etc.) and into the second probe through the second metal-semiconductor region and metal. The net resistance obtained is the sum of the two metal contacts and the sheet resistance of the semiconductor. Several such measurements are repeated and the total resistance is plotted against the pad spacing as illustrated in Figure 2.11. The resistance,  $R_{TOTAL}$ , between two adjacent pads, is given by:

$$R_{TOTAL} = 2R_c + R_s \tag{2.29}$$

where,  $R_c$  is the resistance due to the contact, and  $R_s$  is the resistance due to the semiconductor material

 $R_s$  is given by:

$$R_s = \rho L/dW \tag{2.30}$$

where,  $\rho$  is the resistivity of the semiconductor material, W is the width of the metal pads, L is the pad spacing (L<sub>1</sub>, L<sub>2</sub>, etc.), and d is the mesa height.

But, since the sheet resistance,  $R_{sh}$ , of the semiconductor is given by  $\rho/d$ , Equation (2.29) can be re-written as:

$$R_{TOTAL} = 2R_c + R_{sh}(L/W) \tag{2.31}$$

Equation (2.31) has a gradient of  $R_{\rm sh}/W$ , and x and y axis intercepts at  $L_{\rm x}$  and  $2R_{\rm c}$  respectively. This is illustrated in Figure 2.11. Therefore, by using this procedure, both the metal-semiconductor contact resistance,  $R_{\rm c}$  and the sheet resistance of the semiconductor,  $R_{\rm sh}$  can be calculated [100].



FIGURE 2.11: A graph showing an example of plot of total resistance as a function of pad spacing.

### 2.5.3.2 Circular TLMs (CTLMs)

Circular TLM structures have been proposed to eliminate the drawbacks of linear TLM technique which tends to suffer from unwanted current flow patterns from contact edges unless mesa structures are fabricated [100, 103–106]. A typical CTLM layout is shown in Figure 2.12. Marlow and Das proposed this structure which consists of a circular contact, which is separated by a ring-shaped gap [106]. This structure is easier to fabricate and requires only one level of photolithography and does not require mesa-isolation.



FIGURE 2.12: A cross-section and top view of AlGaN/GaN semiconductor material with Ohmic contact pads for CTLM characterization.

The CTLM structures are characterized in the same way as linear TLMs. The measured resistance changes with varying gap spacing. The total resistance,  $(R_{TOTAL})$  between the contacts for the circular configuration [107], assuming  $R_1 >> L$  and  $R_2 = R_1 + L$ , follows the equation:

$$R_{TOTAL} = \frac{R_{sh}}{2\pi} \left[ ln \frac{R_2}{R_2 - L} + L_T \left( \frac{1}{R_2 - L} + \frac{1}{R_2} \right) \right]$$
(2.32)

where,  $R_{sh}$  is the sheet resistance of the semiconductor material, ln is the natural logarithm (log<sub>e</sub>),  $R_1$  is the radius of the inner contact,  $R_2$  is the inner radius of the outer contact, L is the gap spacing, and  $L_T$  is the transfer length, which is the length of the contact pad required for the current to flow into or out of the contact

The "ln" or "log<sub>e</sub>" term in Equation (2.32) can be evaluated using a Taylor expansion and so Equation (2.32) can be rewritten as shown in Equation (2.33). This relation can be modified into linear approximation by using correction factor c, that accounts for the circular geometry as shown in Equation (2.34) [108].

$$R_{TOTAL} = \frac{R_{sh}}{2\pi(R_1)} \left[L + 2L_T\right] c$$
 (2.33)

$$c = \frac{R_1}{L} ln \left[ \frac{R_1 + L}{R_1} \right] \tag{2.34}$$

where, c is the correction factor

Factors that cause errors in the measurement of contact resistance include the effect of the size of the contact areas and the distance between them, and the effect of a finite spreading resistance in the material on the contact metal [108]. The normalized contact resistance, ( $R_{c-norm}$ ) is obtained by multiplying  $R_c$  with the length of the contact through which the current enters or leaves the inner contact, which is:

$$R_{c-norm} = R_c W_c \quad [\Omega mm] \tag{2.35}$$

where,  $W_c$  is the circumference of inner contact for CTLMs, and  $W_c = W$  which is the pad width for linear TLMs.

### 2.6 Summary

This chapter provided a description of the GaN material system, with an overview of the material properties and its crystal structure. The formation of the 2DEG channel at the AlGaN/GaN interface was discussed and an analytical model of the dependence of its density on the thickness of AlGaN barrier layer or the aluminium content (x) in the AlGaN barrier was presented. The principle of operation of GaN-based HEMTs was described as well as the metal semiconductor contacts used on these devices.

# CHAPTER 3

# GaN DEVICE PROCESSING TECHNOLOGY

In this chapter, various etching techniques used during GaN-based device fabrication including both dry etch techniques and the two main lithography techniques, optical and electron beam lithography will be introduced. The three popular deposition techniques for dielectric films used to form the device gate dielectric or for device passivation will also be described. The techniques include plasma enhanced chemical vapour deposition (PECVD), inductively-coupled plasma chemical vapour deposition (ICP-CVD) and atomic layer deposition (ALD).

## 3.1 Etching Techniques for GaN-Based Devices

Improvements in the performance of AlGaN/GaN heterostructure devices depend on the quality of epitaxial materials and the development of device processing technologies. In particular, effective etching techniques are very essential for defining mesas, recess etching (Ohmic and gate recessing) and dielectric etching. Group III-nitrides have high bond energies (8.9 eV/atom for GaN and 11.5 eV/atom for AlN) compared to conventional III-V semiconductors [109]. The wide bandgaps and high bond strengths of these materials make them essentially chemically inert and highly resistant to bases and acids at room temperature. Therefore, dry etching is the main technique for the processing of GaN-based devices and it will be described further in the following sections.

### 3.1.1 Dry Etching

Due to the chemical stability of group III-nitride semiconductors, the difficulty in etching such materials using wet etch makes dry etching the dominant technique for etching AlGaN/GaN heterostructures, especially for mesa and recess etching. Amongst several techniques used for dry etching, reactive ion etching (RIE) and inductively-coupled-plasma reactive ion etching (ICP-RIE) have been widely studied and used to etch III-nitride materials and will be our main focus of attention in this section.

### 3.1.1.1 Reactive Ion Etching

Reactive ion etching (RIE) utilizes both the chemical and physical components of an etch mechanism to achieve anisotropic profiles, fast etch rates and dimensional control. Figure 3.1 shows a schematic of a RIE system [110]. RIE plasmas are typically generated by applying RF power of around 13.56 MHz between two parallel electrodes in a reactive gas. The substrate is placed on the powered electrone, where a potential is applied and ion energies (typically a few hundred electron-volts) are defined as they cross the plasma sheath. RIE is operated at low pressures, ranging from a few millitorr up to 200 mTorr, which promotes anisotropic etching due to increased mean free paths and reduced collisional scattering of ions during acceleration in the sheath [111]. The Oxford Instruments RIE 80 Plus tool and  $CHF_3/Ar$  gases are used for RIE in the James Watt Nanofabrication Centre (JWNC) of the University of Glasgow. Further details on how dry etching was performed in the processing of devices on this project will be presented in Section 5.1.2.6.



FIGURE 3.1: A schematic of a reactive ion etching (RIE) tool.

The chemistries for dry etching III-nitride semiconductors are mostly halogen-based with the most prevalent being chlorine-based. The etch rates reported for GaN using RIE with various etch chemistries range from 17 to 100 nm/min [112–115]. These etch rates were found to depend strongly on the plasma self-bias voltage, and essentially

independent of the chamber pressure for pressures less than 80 mTorr [114]. Higher etch rates were obtained at high plasma dc biases from -300 to -400 V. Conventional RIE uses both physical (dominant mechanism) and chemical components during etching and both these mechanisms cannot be independently controlled. This impacts the shape of etch profiles significantly especially in the case of III-nitrides where high ion energy is required to break the bonds between their forming atoms.

### 3.1.1.2 Inductively-Coupled Plasma Reactive Ion Etching

The use of high-density plasma etch systems such as inductively-coupled plasma reactive ion etching (ICP-RIE), offers a high-density plasma etch platform to pattern Group III nitrides and has resulted in improved etch characteristics for GaN-based devices [111]. Figure 3.2 shows a schematic of an ICP-RIE system [110]. ICP plasmas are formed in a dielectric vessel encircled by an inductive coil into which RF power is applied. The alternating electric field between the coils induces a strong alternating magnetic field trapping electrons in the centre of the chamber and generating a high density plasma. Since ion energy and plasma density can be effectively decoupled, uniform density and energy distributions are transferred to the sample while keeping ion and electron energy low. Thus, ICP etching can produce low damage while maintaining fast etch rates. Anisotropy is achieved by superimposing an RF bias on the sample. In the JWNC, the Oxford Instruments PlasmaPro System 100 ICP180 RIE tool is available for performing ICP-RIE on devices.



FIGURE 3.2: A schematic of an inductively-couple plasma reactive ion etching (ICP-RIE) tool.

ICP etching is generally believed to have several advantages including improved plasma uniformity over a wider area and higher etch rates. Higher plasma densities and larger ion fluxes available in ICP-RIE tools lead to higher etch rates in these systems. Inductivelycoupled-plasma etching (ICP-RIE) offers low damage etch compared to RIE. The higher efficiency of plasma generation during ICP-RIE method also means that plasma can be generated and sustained in a higher vacuum environment than possible for conventional RIE tools. Etch rates as high as 688 nm/min at DC bias of -280 V and 980 nm/min at -450 V have been reported for GaN using  $\text{Cl}_2/\text{H}_2/\text{Ar}$  and  $\text{Cl}_2/\text{Ar}$  plasmas, respectively [110, 111, 116]. An etch rate of 850 nm/min was also demonstrated using  $\text{BCl}_3/\text{Cl}_2$ plasma at DC bias of -120 V and 30 mTorr pressure [117].

### 3.1.2 Wet Etching

The excellent chemical stability makes group III-nitride materials highly resistant to chemical etching. Only AlN has a known wet etch recipe [118–121]. As such, the processing of GaN devices is done with dry etching.

## 3.2 Lithography

The production of micro and nano-electronics has made incredible strides over the past 50 years. The main driver of lithography improvements is reduced device cost. This has come about through improvements in resolution, resulting in more chips per wafer, higher performance devices, and improvements in productivity, all of which continue to reduce the cost per function in integrated circuits. Lithography is an important and fundamental process used during semiconductor device fabrication to define very small features onto the material being used. This process involves the transfer of patterns from a mask plate unto a wafer surface which is covered with a thin layer of photoresist. Lithography can be performed by using either optical or electron beam (e-beam.) In this section, we will introduce a brief description of these two lithography methods.

### 3.2.1 Optical Lithography

Optical lithography, also known as photolithography creates a resist image of a mask on a wafer. The thin film of resist on the wafer is selectively removed or built up and the subsequent etching, lift-off, ion implantation or metallization is masked by the resist at the areas dictated by the lithography mask. Optical lithography is usually performed using a mask aligner. Figure 3.3 shows an illustration of the optical system of a mask aligner which is used to replicate a mask pattern during optical lithography.

Replicating the mask pattern with a mask aligner produces a resist image and the condenser collects ultra-violet (UV) light from the source and illuminates the mask pattern. The illuminated light then passes through the imaging lens to form an aerial image to



FIGURE 3.3: The optical system of a mask aligner used to replicate a mask pattern during optical lithography.

selectively expose the resist. The substrate can then be isotropically or anisotropically etched, lifted-off, plated, metallized, or implanted using the patterned resist as the mask. Photoresist is a radiation-sensitive compound. There are two types of photoresists used for optical lithography, these are positive and negative photoresist. Figure 3.4 illustrates the outcome of a positive and negative resist once a mask pattern is exposed onto a resist coated wafer and the pattern is etched after resist development of the mask pattern. The photoresists are classified according to their reaction to UV light or radiation. For positive photoresist, the exposed regions become more soluble in developer solutions and can be easily removed through resist development process, which then results in the mask pattern being replicated on the resist coated wafer. On the other hand, the exposed areas in a negative photoresist become less soluble in developer solutions and remain unexposed during resist development, which results in patterns opposite to those on the mask being defined on the wafer.

Optical lithography is, however, limited in resolution to around  $1 \,\mu\text{m}$  due to the wavelength of the lamp source used. An ultra-violet (UV) lamp with a wavelength of 365 nm is used in the JWNC.

### 3.2.2 Electron Beam Lithography

Electron beam lithography (EBL) is primarily used for mask plate production. However, due to sub-micron or deep sub-micron patterning requirements, the technique can be



FIGURE 3.4: An image of a negative photoresist (left) and positive photoresist (right) after exposure, development and etch.

used to write patterns directly onto a resist coated wafer surface by a focused electron beam without a mask plate. The resists used for electron beam (e-beam) lithography are polymers. The behaviour of an e-beam resist is similar to photoresist, that is, a chemical or physical change is induced in the e-beam resist by radiation, which allows the resist to be patterned. For positive electron resist, the polymer interacts with electrons, causing chemical bonds to be broken to form shorter molecular fragments which are more easily dissolved by the developer solution during the development process. On the other hand, for a negative electron resist, irradiation causes radiation-induced polymer linking which creates a complex three-dimensional structure and makes the irradiated resist more difficult to be dissolved during the development process.

Some advantages of using EBL include:

- patterning of sub-micron and deep sub-micron features
- highly automated and precisely controlled operation
- positioning/alignment accuracy
- direct patterning on a semiconductor wafer without using a mask plate
- can be easily accesible for corrections since the layout design for EBL are stored and used electronically in the machine tool system

Despite the several advantages of EBL, there are a few drawbacks of using this lithography technique including low throughput, high cost of operation, and complicated and time consuming processing.

### **3.3** Dielectric Deposition Techniques

In semiconductor materials, and in particular GaN-based device processing, insulating thin (dielectric) films can serve three essential purposes. They can be:

- 1. used as a protecting layer (also known as passivation layer) at the end of device fabrication so as to protect the device from harsh ambience and to ensure reliable device operation over time,
- 2. a part of the active device (for example, gate dielectric of an AlGaN/GaN HEMT), and
- 3. used as hard masks to protect against diffusion during ion implantation, etching or annealing.

For these applications, high quality dielectric films deposited at low temperatures that are stable and of low hydrogen content are required. The most commonly used dielectric materials in III-V and in particular GaN-based device technology are silicon dioxide  $(SiO_2)$  and silicon nitride  $(Si_3N_4)$ , although other dielectrics such as alumina  $(Al_2O_3)$ , hafnia  $(HfO_2)$  and zirconia  $(ZrO_2)$  are gaining ground in popularity due to their higher dielectric constants of approximately 10, 25, and 22, respectively [122, 123]. The most common techniques for depositing dielectric films on GaN-based devices are plasma enhanced chemical vapour deposition (PECVD), inductively-coupled plasma chemical vapour deposition (ICP-CVD) and atomic layer deposition (ALD). In this section, we will give a brief overview of how the dielectrics are deposited using these techniques.

### 3.3.1 Plasma Enhanced Chemical Vapour Deposition

Plasma enhanced chemical vapour deposition (PECVD) is the conventional technique for depositing SiO<sub>2</sub> and SiN<sub>x</sub> dielectric films. Figure 3.5 shows a schematic of a typical PECVD tool from Oxford Instruments [124]. This figure illustrates the wafer, parallel electrodes, RF generator, plasma gas, reactive gas and the chamber of the PECVD system. The presence of a gas plasma allows reactions (which would otherwise only occur at high temperatures) to take place at a comparatively low temperature. Plasma is a partially ionized gas with free electron content (about 50%). Plasmas are divided into two groups; thermal and cold (also known as non-thermal). In thermal plasmas, both electrons and gas particles are at the same temperature; however, in cold plasmas, the electrons have a much higher temperature than the neutral particles and ions. This feature allows PECVD systems to operate at low temperatures of between 100 ° C and 400 ° C. PECVD tools must contain two electrodes (in parallel), plasma gas, and reactive gas in a chamber. In order to excite and sustain the excitement state of the plasma, a voltage must be applied to the plasma by using an RF signal between two electrodes.

The PECVD process begins with placing a wafer on the bottom electrode and the reactive gas with the deposition elements is introduced into the chamber. Deposition is then achieved by introducing reactant gases between the parallel electrodes (grounded electrode and RF-energized electrode). The capacitive coupling between the electrodes excites the reactant gases into a plasma, which induces a chemical reaction and results in the reaction product being deposited on the substrate. The substrate, which is placed on the grounded electrode, is typically heated to between 250 ° C and 350 ° C, depending on the specific film requirements.



FIGURE 3.5: A schematic of a plasma enhanced chemical vapour deposition (PECVD) tool.

Using PECVD to deposit dielectric films has many advantages including low operation temperature, lower chances of cracking of the deposited layer, good dielectric properties of deposited layer including good stoichiometry (consistence) and good adhesion, good step coverage (the step coverage is the ratio of film thickness along the walls of a step to the thickness of film at the bottom of the step), good control of deposition thickness of dielectric (deposition rates of 30 nm/min to 70 nm/min), good uniformity, and less temperature dependent. However, low temperature films prepared by this technique can contain significant amounts of hydrogen. Generally, hydrogen content in a dielectric film is undesirable for microelectronics applications as it reduces the insulating behaviour of a gate insulator due to the formation of defects by the presence of hydrogen. Another important reason for this undesirable property is that at high temperatures hydrogen induces large stress and imperfections within the dielectric film [125, 126]. Hydrogen incorporation is inevitable in dielectrics deposited by PECVD as the films are typically prepared from hydride source gases such as silane  $(SiH_4)$  and ammonia  $(NH_3)$ . Also plasma damage during PECVD can result in physical and electronic degradation of the III-V devices.

### 3.3.2 Inductively-Coupled Plasma Chemical Vapour Deposition

During this technique, dielectric films are deposited by plasma enhanced chemical vapour deposition using an inductively coupled plasma (ICP) as the source. Conventional PECVD systems use high substrate temperatures of greater than 200 °C to achieve a dense film, however, using the high plasma density of the ICP source makes it possible to achieve film densities similar to PECVD at temperatures as low as 20 °C. Figure 3.6 shows a schematic of an ICP-CVD system from Oxford Instruments [127]. The dopant gases are fed through the gas distribution ring located just above the wafer. This minimises the deposition in the source area and reduces the need for cleaning the system. Inert gases are then fed into the chamber via the top of the ICP source. For SiO<sub>2</sub> dielectric, the inert (source) gas is usually O<sub>2</sub> and the dopant gas is usually SiH<sub>4</sub>, whilst for Si<sub>3</sub>N<sub>4</sub> dielectric, the inert (source) gases is usually N<sub>2</sub> and Ar, and the dopant gas is usually SiH<sub>4</sub>. A high conductance chamber and vacuum system enables the ability to flow these process gases at relatively high rate while maintaining the low process pressures of between 1 to 10 mTorr necessary to sustain the high-density plasma.



FIGURE 3.6: A schematic of an inductively-coupled plasma chemical vapour deposition (ICP-CVD tool).

Advantages of ICP-CVD include better control of deposited dielectric thickness than PECVD deposited dielectrics (deposition rate of 15 nm/min) and good step coverage. This deposition method also allows better control of the deposited film stress by modifying various parameters such as process pressure, temperature, RF power, and process gases ratio. Furthermore, ICP-CVD allows deposition on temperature sensitive devices and provides better uniformity of deposited film thickness than PECVD. At low pressures, a high dielectric film quality can be maintained in the ICP-CVD process, giving a very high film quality. However, ICP-CVD equipments are more expensive than PECVD ones.

### 3.3.3 Atomic Layer Deposition

Atomic layer deposition (ALD) is a vapour phase technique capable of producing thin films of a variety of materials including hafnia and zirconia. Based on sequential, selflimiting reactions, ALD offers exceptional conformality on high-aspect ratio structures, thickness control at the nanometre level, and tunable film composition [128]. With these advantages, ALD has emerged as a powerful tool for many industrial and research applications. Figure 3.7 shows a schematic of an ALD tool [129]. During ALD, separate precursor gases for a target material are sequentially dosed into a vacuum chamber under computer control. The first precursor is chemisorbed into the substrate surface and the chamber is then purged. The second precursor is then introduced and reacts with the first, yielding a monolayer of film. The cycle then continues by reintroducing the first precursor, and so on [130].

Dielectric films deposited by ALD offer many benefits in semiconductor engineering. Some advantages are that, as the ALD process deposits precisely one atomic layer in each cycle, complete control over the deposition process is obtained at the nanometre scale. Another advantage of ALD is that conformal coating can be achieved even in high aspect ratio and complex structures, whilst pin-hole and particle free deposition can also be achieved. However, ALD is a very time consuming technique which makes depositing thicker layers of film difficult (average deposition rates of around 100 nm to 300 nm/hr). The complicated processes involved during ALD also presents another drawback of this deposition method.

## 3.4 Summary

This chapter reviewed the various etching techniques used during GaN-based device fabrication. The two main lithography techniques used, optical and electron beam lithography were also described. The last section of this chapter then briefly discussed the



FIGURE 3.7: A schematic of an atomic layer deposition (ALD) tool.

three deposition techniques for dielectric films. In the next chapter, a review of some of the current techniques used to realise GaN-based normally-off high electron mobility transistors (HEMTs) will be given.

# CHAPTER 4

# REVIEW OF GaN-BASED E-MODE HEMTs

Owing to their high breakdown voltages and low on-resistances, GaN based enhancement -mode (E-mode) HEMTs are a major attraction for various electronic applications and in particular power switching applications due to the added safety of a normally off device. There have been several techniques used to fabricate E-mode devices to date, all incorporating different technologies to achieve the thin barrier layer or empty the 2DEG channel underneath the device gate (at zero gate bias) necessary for a normally-off operation. Some of the key challenges of using AlGaN/GaN heterostructure for making normally-off electronic devices are listed below:

- The strong polarization effect in the conventional AlGaN/GaN heterostructure leads to more negative threshold voltages [59], demanding a large positive shift in the threshold voltage  $(V_{th})$  for the device to be converted to an enhancement-mode (E-mode) one during the device processing stage.
- Due to the strong chemical inertness of GaN, wet etching techniques which are generally used in gate recessed InP and GaAs-based E-mode HEMTs have proved very difficult in AlGaN/GaN devices.

This chapter reviews some of the commonly used techniques to overcome the above challenges and realise GaN-based E-mode HEMTs.

# 4.1 Double Barrier HEMT with an Etch-Stop Layer

As the name suggests, the double-barrier technique employs two active barrier layers as shown in Figure 4.1. The device consists of a very thin AlGaN layer capped with an AlN layer. The incorporated thin AlGaN layer is only used to control the threshold voltage of the device and hence the thickness is kept around the critical thickness of the layer for the specific Al-content. Therefore, the threshold voltage is set by epitaxy.



FIGURE 4.1: Double barrier AlGaN/GaN HEMT structure with etch-stop layer.

The polarization charge induced by the thin AlGaN layer is not sufficient to generate a high carrier concentration in the channel. Adding an AlN cap on top of the AlGaN layer induces a large polarization field because of the inherent spontaneous polarization as a result of the lattice mismatch and larger conduction band offset, thereby enhancing the carrier concentration in the channel. The top active layer can, however, be selectively dry/wet etched until the second active layer which acts as an etch-stop layer is reached [120, 131–133]. The addition of the etch-stop layer makes this technique reproducible, however, there is lack of variable options of the top active layer structure as only AlN is known to be selectively chemically etched using AZ400K developer solution at (> 60 °C) over other materials in the GaN family of semiconductors, limiting the device structure to only have AlN as the top active layer [119, 134]. Another disadvantage of using this technique is that, for dry recess etching, the etch-stop layer is limited to only Indium (In) related chemistries such as InN, InAlN or InGaN as they require very high table-temperature in order to etch [120, 132, 133].

### 4.2 Fluorine Ion Implantation

Another common technique for realising GaN-based e-mode HEMTs is the use of fluorine ion implantation. The fluorine ions can be implanted into the conventional  $\approx 20$  nm thick AlGaN barrier layer as illustrated in Figure 4.2. The implantation of these negatively charged fluorine ions causes the channel of electrons underneath the gate-foot region to be depleted [37, 51, 135–138]. Although fluorine ion implantation eliminates the problems incurred during dry etching techniques, it is known to be very unreliable as the implanted fluorine ions have the tendency to move around in the GaN crystal lattice;
making the device very unstable, and thus, degrading the drain breakdown voltage of the device. Also, the threshold voltage  $V_{th}$  is not uniform across a wafer of several devices fabricated using this method.



FIGURE 4.2: Achieving an e-mode GaN HEMT device using fluorine ion implantation.

# 4.3 P-N Junction Gate GaN HEMTs

In a technique pioneered by Panasonic, a standard AlGaN/GaN HEMT structure is capped with a p-type AlGaN or GaN layer to realise normally-off operation [139, 140]. This device structure is illustrated in Figure 4.3 and has a p-GaN cap layer of around 100 nm-120 nm thick [139, 140]. At zero gate bias, the electrons in the device channel underneath the gate are depleted due to the high built-in potential of the p-n junction. However, the application of forward bias gate voltage allows the injection of holes in the barrier layer, which in turn attracts free electrons to complete the 2DEG channel and so allows the flow of current if a drain-to-source voltage is applied.

This technique enables the fabrication of both depletion and enhancement mode HFETs on the same wafer, thus opening up the possibility of designing high-speed, low-consumption GaN-based logic integrated circuits. Despite this advantage of using the p-n junction gate device fabrication technique, growing high-quality and uniform p-type material in group III-Nitrides is still difficult. Also, p-GaN/AlGaN interface trap-related dispersion presents this technique with another drawback. Also, the threshold voltage achievable with this device is around +0.55 V and the forward bias voltage is limited to around +3 V.



FIGURE 4.3: P-N junction gate e-mode GaN HEMT.

# 4.4 InGaN Cap Layer GaN HEMTs

The use of an InGaN cap layer was first proposed in 2007 by Mizutani et al. [141]. The key idea of this method is to employ the polarization-induced field in the InGaN cap layer as illustrated in Figure 4.4a, by which the conduction band is raised leading to a normally-off operation [141–143]. A thin layer of InGaN was grown on a conventional n-AlGaN/i-GaN HEMT structure. Even though the direction of spontaneous polarization is the same for all the layers in the device, the directions of piezoelectric polarization are opposite between the InGaN cap layer and the AlGaN barrier layer. The polarization-induced field in the InGaN cap layer causes the conduction band to raise leading to normally-off operation of the device as illustrated in Figure 4.5 [141, 143]. However, an application of a forward bias voltage causes the conduction band to lower below the Fermi level and hence there is a presence of 2DEG which allows conduction. In the case of HEMTs without the InGaN cap layer, 2DEG is formed in the AlGaN/GaN interface which leads to the normally-on operation as illustrated in the energy band diagram in Figure 4.5b. InGaN cap E-mode devices do however suffer from small threshold voltage, high sheet resistance, low transconductance and high contact resistance.

## 4.5 Tunnel-Junction GaN HEMTs

One of the most recent techniques proposed for realising e-mode GaN-based HEMTs is the use of a tunnel-junction in the devices. In this method of achieving enhancement mode GaN devices, the control of the source-to-drain current flow is realized through a gate controlled tunnel-junction instead of a gate controlled 2DEG channel. The device structure of a tunnel-junction e-mode GaN HEMT is shown in Figure 4.6. The operating

		Gate		
Source	InGal	N Psp	РРЕ	Drain
n-AlGaN PSP PPE				
i-GaN		Psp		
Saphirre				

FIGURE 4.4: Schematic cross section diagram of an AlGaN/GaN HEMT with InGaN cap layer.



FIGURE 4.5: Energy band diagrams: a) with and b) without InGaN cap layer.

principle of the AlGaN/GaN tunnel-junction FET is that, at zero gate bias, the 2DEG channel is fully turned on. However, the effective Schottky barrier width at the source metal 2DEG junction (with a Schottky barrier height of 0.8 eV) is ~ 10 nm. Such a potential barrier thickness yields a negligible tunnelling current and effectively blocks the current flow, keeping the device in the "OFF" state. As the gate bias becomes more positive, the conduction band of the GaN is pulled down, effectively reducing the schottky barrier width. At  $V_{GS}=3 \text{ V}$ , the schottky barrier width is less than 1 nm, and the tunnelling current is significantly higher, turning the device "ON" [144, 145]. Realization of E-mode tunnel-junction devices however involves very complicated fabrication processes which reduces its reproducibility and repeatability. To our knowledge, the fabrication of these devices is neither reproducible nor repeatable [146].

## 4.6 N-Polar GaN HEMTs

In N-face AlGaN/GaN heterostructures, the spontaneous and piezoelectric polarizations have opposite directions in comparison to the Ga-face structure as shown in Figure 4.7a



FIGURE 4.6: Cross section of a tunnel-junction AlGaN/GaN FET.

[147]. The polarization induced charge at the interface of this heterostructure is negative, and holes can be accumulated at this interface. However, electrons will be confined at the GaN/AlGaN interface if GaN is grown on top of the AlGaN due to the positive sheet charge which will be formed in this case. A thick enough AlGaN layer is then grown on top of the GaN layer in order to deplete any of the 2DEG induced at the bottom AlGaN/GaN interface [148–150]. In the drain, source and access regions, this top AlGaN is etched away, leading to a high electron density in these regions as shown in Figure 4.7 [147]. The 2DEG under the gate is then induced when the gate is forward biased, while the charge in the access region is always present. Although N-polar devices provide E-mode without gate recess etch and the  $V_{th}$  of these devices are not affected by processing steps, these devices suffer from high leakage current and the growth of good quality N-polar GaN is very complex [151, 152].



FIGURE 4.7: Cross section of N-Polar AlGaN/GaN HEMT: a) showing directions of polarization charges and b) with AlGaN Layer.

# 4.7 Gate Recessed GaN HEMTs

Gate recess technique is one of the most common methods of achieving an enhancement mode device and it is done by physically thinning the barrier by etching away the barrier under the local gate-foot region as shown in Figure 4.8. Saito et al., 2006 realized a normally off AlGaN/GaN HEMT using recessed gate fabrication technique which exhibited a maximum drain current of 100 mA/mm, a threshold voltage of -0.14 V and a breakdown voltage of around 435 V [153]. As earlier explained, dry etching is the only effective and easiest method for GaN. However, as Reactive Ion Etching (RIE) using inductively coupled plasma etcher with BCl<sub>3</sub> or SiCl<sub>4</sub>/Cl<sub>2</sub> involves physical etching, it damages the sample surfaces and it is also very difficult to control the etch rate, and thus over-etching becomes an issue [50, 84, 118, 153–158]. Over-etching due to dry etching makes achieving uniform threshold voltage and repeatability between devices fabricated on the same wafer (let alone between wafers) extremely difficult, therefore, gate recess etching is unattractive without an etch stop layer (and to date, no suitable such layer that can be incorporated in the conventional AlGaN/GaN HEMT structure has been identified).



FIGURE 4.8: Gate recessed AlGaN/GaN HEMT structure.

### 4.8 Thin Barrier Devices

The use of a thin barrier to realise e-mode GaN-based devices has been proposed by several groups. Uniform device performance can be obtained through the use of the thin barrier, i.e. an AlGaN thickness of under ~ 10 nm because it requires no recessing [119, 159–162]. In this case, the channel can be easily depleted under the gate for normally-off operation due to the reduced 2DEG density, but the devices suffer from reduced drain currents and high On-resistances since the low density 2DEG is across the entire source-drain region. To reduce the access resistance between the gate-source and gate-drain regions, SiO<sub>2</sub> passivation layers have been used in these regions [160, 161]. In Reference [119] the use of an epitaxial AlN cap layer that was selectively etched from the gate region was also demonstrated.  $V_{th}$  of these devices is around 0 V.

Recently, a thin 12 nm barrier normally-off AlGaN/GaN MOS-HEMT with an overlap gate structure, i.e. also no gate-source and no gate-drain separation, and using a thick

HfO<sub>2</sub> with a large dielectric constant as a gate insulator was proposed [163]. The device demonstrated a  $V_{th}$  of +3 V and  $I_{DSmax}$  of 730 mA/mm at  $V_{GS}$  of 10 V. Normally-off operation was attributed to negative charge in the gate dielectric which depleted the 2DEG channel for  $V_{GS} < V_{th}$ . The main drawback for this approach was the reduced device breakdown voltage.

## 4.9 Summary

In this chapter, the current techniques being used to fabricate enhancement-mode GaNbased HEMTs were reviewed. Due to the drawbacks of current approaches, there is on-going research into realising normally-off GaN devices using a design that is easily repeatable and reproducible and one which does not require too many complicated fabrication steps. The next chapter will provide details of the actual fabrication processes used in realising GaN HEMTs (on this project).

# CHAPTER 5

# DEVICE FABRICATION PROCESSES

In this chapter, the fabrication processes used to realise normally-off GaN transistors will be described. Details of fabrication processes for single finger normally-off or enhancement-mode transistors will be given with focus on the mask design and layout of the transistors.

### 5.1 Device Layout and Process Flows

Prior to fabrication, devices must be designed using a computer aided design (CAD) software; in this case, the Tanner EDA's L-Edit, Version Win 32 14.13 layout editor was used [164]. Once the patterns are designed, the design file is sent to an electron beam lithography tool which writes the required design on a mask plate as described in Section 3.2.2. The quartz mask plate is chrome plated on one side on which patterns are created. The final mask plate is then used for every photolithography step as required throughout the fabrication process of the devices. Figure 5.1 shows an example of a complete transistor layout design in L-Edit. This design layout is  $10 \text{ mm} \times 10 \text{ mm}$  in size and includes a set of horizontal and vertical alignment markers, 2 circular TLM structures, 16 linear TLM structures, 24 metal-oxide-semiconductor (MOS) capacitors, and 182 transistors.

#### 5.1.1 Device Structure

The epitaxial layer structure used for making the devices consists of (from top to bottom) an ultrathin sub-critical  $3 \text{ nm Al}_{0.25}\text{Ga}_{0.75}\text{N}$  barrier layer, a  $3 \mu \text{m}$  GaN channel, a 2 to 5 nm AlN nucleation and transition layers grown on either a sapphire or silicon substrate.



FIGURE 5.1: A screen shot of a complete device layout design in L-Edit including alignment markers, test structures and DC transistors.

Note the very thin AlGaN barrier layer of 3nm, which is crucial to the realisation of normally-off operation. Unlike conventional device structures which use approximately 20 nm thick AlGaN barrier layers, there is no inherent 2DEG channel at the AlGaN/GaN interface in this structure as the AlGaN barrier thickness is below that required for 2DEG formation (4 nm). This was discussed earlier in detail in Section 2.1.2.1 of Chapter 2.

The device layout consisted of a gate contact that nominally overlaps the source and drain contacts, with a dielectric in between the gate and the semiconductor (and the source/drain contacts). This layout is similar to that in Reference [163]. To demonstrate the device concept, only DC characterizations was required, and so a gate wrap-around (drain) transistor layout design was considered in this project. In this type of transistor, the gate contact encircles the drain contact of the device. The advantage of using such a design is that the device mesa isolation is not required which in-turn enables a fast turnaround in device production as only two main lithography steps (excluding bondpad lithography) are needed in order to fabricate devices.

#### 5.1.2 Process Flows

The main steps involved in the process flow of fabricating the normally-off AlGaN/GaN HEMTs are sample cleaning and preparation, photolithography (Ohmics and gate), metallization (Ohmic and gate metals), Ohmic contacts annealing, dielectric deposition, and dry etching of dielectric. Figure 5.2 shows the process flow for device fabrication. Some process details, for example use of different photoresist, are applicable using these same fabrication steps. Where there is a modification of the process flow used to fabricate a particular device, this will be indicated in the relevant section.

The device structure is a (drain) wrap-around gate requiring a minimum number of processing steps. The source/drain Ohmic contacts were formed by photolithography, metal deposition and lift-off as illustrated in Figure 5.2(a) and (b). The Ohmic metal scheme used was Ti/Al/Ni/Au with thicknesses of 30 nm, 180 nm, 40 nm, and 100 nm, respectively. The sample was then annealed at 775 °C for 30 seconds in nitrogen (N<sub>2</sub>) ambient and then a layer of SiO<sub>2</sub> with a refractive index of around 1.5 was deposited by plasma enhanced chemical vapour deposition (PECVD) as illustrated in Figure 5.2(c). It should be noted that the sample was not cleaned after annealing and before the deposition of the SiO<sub>2</sub> film. This dielectric pre-deposition cleaning process is vital to getting rid of surface traps as is demonstrated in the CV graphs shown in Appendix C.2 of this thesis. Further work is planned to investigate the full extent of the effect of surface cleaning before dielectric deposition. The surface of the sample underneath the gate region was fully exposed to nitrogen (N<sub>2</sub>) during the annealing step. The exposure of the very thin (3 nm) AlGaN barrier layer to such high annealing temperature could

cause some damge in the material which could affect the device performance. The gate contacts were then formed by lithography, metal deposition and lift-off as indicated in Figure 5.2(d) and (e). The metal scheme used for the gate was Ni/Au with thicknesses of 50 nm and 150 nm, respectively. Ohmic bond pad patterns were then defined by photolithography and the SiO<sub>2</sub> dielectric was etched away from the top of the Ohmic contacts by RIE as indicated in Figure 5.2(f). The bondpad metal was then deposited on top of the regions of the Ohmic contacts which were exposed during the dielectric etch. The metal scheme used for the bondpads was Ti/Au with thicknesses of 20 nm and 200 nm, respectively. The final cross-section of the fabricated e-mode HEMT is indicated in Figure 5.2(g). Further details of these processes are presented in the following sections.



FIGURE 5.2: Typical process flow used to fabricate e-mode HEMTs in this research. PR stands for photoresist.

#### 5.1.2.1 Sample Cleaning and Preparation

Sample cleaning is essential before starting any fabrication process for two main reasons. The first reason is to remove the native oxide that forms when the semiconductor surface is exposed to air or other ambient, while the second reason is to remove any dust particles and the remains or residues that arise, e.g. while the wafer is being cleaved.

#### 5.1.2.2 Photolithography

Two photolithography recipes were used for Ohmic and gate patterning in this project. The photoresists used for the first recipe (photolithography recipe 1) are the Shipley

S1818 for Ohmic lithography and the Shipley S1805 for gate lithography. They are both positive photoresists and were chosen because of their well developed processes available in the James Watt Nanofabrication Centre (JWNC). The optical lithography processing starts with the sample cleaning prior to the photoresist coating. The mask plate is also cleaned prior to the mask exposure. To form a good patterning, the original mask (the chrome plated mask plate) instead of a copy was used. After cleaning, the photoresist (S1818 for Ohmics and S1805 for gate) is applied to the wafer surface with constant spinning speed of 4000 revolutions per minute (rpm) for 2 minutes. After spinning, the sample is baked on a hot plate at  $65 \,^{\circ}$ C for 2 minutes. Spinning of the sample provides uniform photoresist coating while the baking is done to remove the solvent from the photoresist film and also to increase the adhesion of the photoresist to the sample. The S1818 photoresist results in a resist layer thickness of  $1.8\,\mu\text{m}$  while the S1805 photoresist results in a thickness of  $0.5 \,\mu\text{m}$ . The sample is then pre-developed in a developer solution containing Shipley Microposit Developer Concentrate and water at a ratio of 1:1 for 60 seconds and then it is rinsed with reverse osmosis (RO) water for 60 seconds. The pre-development of the photoresist causes a hardening of the upper layer of the unexposed photoresist. After that, the sample is then aligned with respect to the mask using the Karl Suss MA6 mask aligner machine. When the sample is correctly aligned to the pattern on the mask (this is done by checking through the optical microscope on the mask aligner), it is brought into contact with the mask and the photoresist is exposed under ultraviolet (UV) light (wavelength of 365 nm). The exposure time for Ohmic and gate lithography is 5 and 3 seconds, respectively. The exposed photoresist is dissolved in the developer solution containing Shipley Microposit Developer Concentrate and water at a ratio of 1:1 for 75 seconds with slight agitation. Once developed, the photoresist profile exhibits an undercut created by an increased development time. The sample is then rinsed with RO water for 60 seconds and blow dried with a nitrogen gun. After the photoresist development, the sample is ashed in the Oxygen Barrel Asher at 40 watts for 3 minutes in order to remove the resist residues left in the developed regions of the sample surface after development. Figure 5.3 illustrates the process flow for performing photolithgraphy and metallization using the first recipe. This process flow will be referred to as photolithography recipe 1 in the remainder of this thesis.

A second recipe for photolithography (photolithography recipe 2) was developed in order to improve the undercut profile and hence provide better lift-off. Lift-off after metallization of samples using this new recipe was found to be better and cleaner (hardly any unwanted metal deposits were left on sample surface). The second recipe involved the use of two photoresists; the LOR 10A and the S1818 for both Ohmic and gate lithography. The lithography process for this is started after sample cleaning. The sample surface is firstly covered by the first layer of photoresist LOR 10A, and span at 6000 rpm



FIGURE 5.3: Photolithography process using recipe 1. a) sample is cleaned b) photoresist is spun onto sample c) sample is soaked in developer solution and exposed under UV light with mask plate in contact with sample d) photoresist is developed giving an undercut profile e) sample is metallized f) lift-off process is performed to remove photoresist and unwanted metal while leaving behind only the metal which is directly evaporated onto the sample surface.

for 30 seconds, then it is baked on a hot plate at 150 °C for 2 minutes. After that, the second layer of photoresist is applied to the sample and span at 4000 rpm for 30 seconds, then it is baked at 115 °C for 3 minutes. Once the coating of the sample is completed, the next step is exposing it in the mask aligner. Once the sample is accurately aligned with the pattern on the mask, the exposure can be performed. The photoresist is exposed to UV light when the sample is brought in contact with the mask and the exposure time is 6 seconds. The exposed sample is then soaked in MF319 developer solution for 2 minutes and 30 seconds and then rinsed with RO water for 60 seconds. Finally the sample is blow dried with a nitrogen gun. After the photoresist development, the sample is ashed in the Oxygen Barrel Asher at 60 watts for 5 minutes in order to remove the resist residues left in the developed regions of the sample surface after development. Figure 5.4 illustrates the process flow for performing photolithography and metallization using the second recipe. Wherever photolithography recipe 2 is stated in this thesis, this description here should be used as reference of the process flow.



FIGURE 5.4: Photolithography process using recipe 2. a) clean sample b) spin and bake LOR 10A and photoresist c) expose sample under UV light d) develop resist and LOR 10A e) deposit metal f) lift-off the bi-layer stack and unwanted metal from sample surface.

#### 5.1.2.3 Metallization

Deposition of metals onto semiconductor wafers or samples can be performed through several different methods including thermal or electron beam evaporation and sputtering. In this project, electron beam evaporation was employed for the metallization process by using either a Plassys MEB 400S electron beam evaporator (Plassys II) or a Plassys MEB 550S (Plassys IV). Before Ohmic metal deposition, the sample is deoxidised by soaking it in hydrochloric acid:water (HCl : H<sub>2</sub>O) solution of 1:4 ratio for 1 minute at room temperature, in order to remove the native oxide which forms when the AlGaN barrier layer surface is exposed to ambient air. The sample is loaded into the chamber of the Plassys and then the required metal stack and layer thickness is specified on the computer connected to the equipment. The Ohmic metal scheme is Ti/Al/Ni/Au (30 nm/180 nm/40 nm/100 nm), respectively. The gate metal scheme is Ni/Au (50 nm/150 nm), respectively. This process is automatic and the equipment deposits the chosen metals on the sample in sequence through a recipe that appears on the computer display. Since the patterns have already been defined on the photoresist during the photolithograpy stage, when the evaporated metal covers the sample surface, part of the deposited metal covers the exposed areas directly on the semiconductor surface, while the others are deposited on top of the photoresist on the unexposed areas. Lift-off technique is employed for simple and easy patterning of the metals that are deposited. During lift-off, the metals which are deposited on top of photoresist are removed, while the metals directly deposited on the exposed areas on top of the semiconductor material remain to form the contacts. For photolithography recipe 1, the lift-off process is performed by soaking the sample in a 50 °C warm acetone for 15 minutes for complete lift-off of photoresist and unwanted metal. For photolithography recipe 2, lift-off is performed by firstly soaking the sample in a 50 °C warm acetone for 15 minutes to remove the photoresist and then in a 50 °C warm 1165 stripper for 10 minutes to remove the LOR and unwanted metal. After the lift-off of the metals, the sample is rinsed with RO water for 1 minute, soaked in IPA for 1 minute and then blow dried with a nitrogen gun.

#### 5.1.2.4 Ohmic Annealing

Having deposited the metal stack of the Ohmic contacts, annealing is the fundamental process used to achieve many aims including creating an alloy of low resistance, creating nitrogen vacancies (in the AlGaN barrier layer) which act as n-type doping, enhancing the adhesion of metal to the semiconductor and increasing mechanical stability [165]. In Section 2.5.2, the function of each metal in the Ohmic metal scheme was explained. Ohmic contact rapid thermal annealing is often carried out in nitrogen (N<sub>2</sub>) or hydrogen (H<sub>2</sub>) ambient for around 30 to 60 seconds. The Joint Industrial Processors for Electronics (JIP ELEC) JetFirst rapid thermal annealer (RTA) system was used for annealing the Ohmic metal contacts at a temperature of 775 °C for 30 seconds under nitrogen (N<sub>2</sub>) ambient. Figure 5.5 shows microscope images of transistors during fabrication. The image on the left shows the source and drain Ohmics after they are metallized and the image on the right shows the Ohmics after they are annealed at high temperature. The roughness of the Ohmic metals shows the effect of the high temperature during annealing process which enables the mixture of the first metal in the Ohmic metal stack (Ti) with the semiconductor and hence the formation of an alloy.

#### 5.1.2.5 Dielectric Deposition

After annealing of the Ohmic contacts, silicon dioxide  $(SiO_2)$  was deposited on the samples by plasma enhanced chemical vapour deposition (PECVD) in the dry etch lab of the JWNC using the Oxford Instruments PECVD 80 Plus tool. The following deposition conditions were used: temperature of 300°C, RF power of 15 watts, and pressure of 1 torr. The gases used for dielectric deposition were SiH<sub>4</sub>, N<sub>2</sub>O and N<sub>2</sub> at



FIGURE 5.5: Optical microscope image of Ohmic contacts for a gate wrap-around device after metallization before and after rapid thermal annealing.

flow rates of 7 sccm, 200 sccm and 85 sccm, respectively. Deposition times depend on the thickness of dielectric film required. The dielectric isolates the gate metal from the Ohmic contacts and the semiconductor. When a positive gate voltage is applied (with respect to the source), the metal gate, dielectric and semiconductor heterojunction act similar to a parallel plate capacitor and electrons are attracted to the heterojunction interface.

#### 5.1.2.6 Dry Etching

Dry etching was used to remove the dielectric from certain parts of the sample surface to expose the metal contacts for device characterization. In the JWNC, the PECVD SiO<sub>2</sub> was etched using the Oxford Instruments RIE 80 Plus tool. The gases used were CHF<sub>3</sub> and Ar at flow rates of 25 sccm and 18 sccm, respectively. During this process, an RF power of 200 watts and a pressure of 20 millitorr are used at room temperature. All equipment in the lab which require room temperature for operation are run at 22 °C. The etch rate for SiO<sub>2</sub> using this etch chemistry is around 37 nm/min, this enables the etching time for various thicknesses of SiO<sub>2</sub> to be calculated accordingly. For example, to etch a 20 nm thick SiO<sub>2</sub> film, the etch time ia calculated to be around 32.5 seconds. Figure 5.6 shows microscope images of gate wrap-around transistors before the dielectric is etched away from the surface of the Ohmic contact metals and after the bondpads are exposed and metallized on top of the source and drain contacts.

#### 5.1.3 Mask Design and Layout

As mentioned earlier, the mask plate is required to pattern the photoresist during the photolithography process. After the L-Edit CAD software is used to create a pattern



FIGURE 5.6: Optical microscope image of gate wrap-around device before and after dielectric etch and Ohmic bondpad formation.

file, this file is exported into a GDS II file format and sent to the JWNC in University of Glasgow for fabrication on a quartz mask. The devices considered and fabricated in this project are DC devices. Figure 5.7 shows a picture of an L-Edit layout for a gate wrap-around transistor indicating all the layers which will be patterned on the mask plate. Each pattern corresponds to a photolithography step performed during the fabrication of the transistor. Figure 5.8 shows L-Edit images of gate wrap-around (drain) transistors with the gate contacts of each transistor overlapping the source and drain Ohmic contacts nominally. The transistors in "a", "b", "c" and "d" of this figure have gate lengths of  $4 \,\mu$ m,  $6 \,\mu$ m,  $8 \,\mu$ m and  $10 \,\mu$ m, respectively. The e-mode HEMTs fabricated in this project all use the same design layout as shown in this figure. Figures 5.9, 5.10 and 5.11 shows images of the Ohmic, gate and bondpad lithography layers, respectively, while Figure 5.12 shows an image of the combination of all three lithography layers on the mask plate. They also show images of the metal-oxide-semiconductor capacitors (MOSCAPs) which are presented in Appendix C.

# 5.2 Summary

This chapter presented a schematic overview and description of the process flows used to fabricate single finger normally-off GaN devices. A summary of the individual process details is provided in Appendix A for easy reference. A brief discussion on the mask design and layout of these devices was then presented. A detailed presentation and analysis of the DC characteristics of the e-mode GaN devices fabricated in this work will be provided in the next chapter.



FIGURE 5.7: L-Edit layout of individual cells of a gate wrap-around transistor layers to be formed by photolithography.



FIGURE 5.8: L-Edit layout of a gate wrap-around HEMT (source and drain Ohmic contacts are nominally overlapped by the gate contact).



FIGURE 5.9: L-Edit layout of Ohmic contacts on mask plate which was used for Ohmic lithography during fabrication of e-mode HEMTs.



FIGURE 5.10: L-Edit layout of gate contacts on mask plate which was used for gate lithography during fabrication of e-mode HEMTs.



FIGURE 5.11: L-Edit layout of bondpads on mask plate which was used for bondpad lithography during fabrication of e-mode HEMTs.



FIGURE 5.12: L-Edit layout of the three lithography layers used during photolithograpy to fabricate e-mode HEMTs after successful alignment.

# CHAPTER 6

# NOVEL NORMALLY-OFF AlGaN/GaN MOS-HEMT

In this chapter, results of the proposed normally-off AlGaN/GaN transistors are presented. Devices grown on different substrates, sapphire and silicon, and employing different gate lengths will be presented. The device characteristics including output characteristics, transfer characteristics, and transconductance of the devices will be given and analysed. We will also present the breakdown voltages of these devices and a preliminary study of their reliability.

# 6.1 Normally-off AlGaN/GaN MOS-HEMT on Sapphire Substrate

#### 6.1.1 Device Structure and Fabrication

Figure 6.1 shows a cross-section of the proposed normally-off AlGaN/GaN HEMT. It consists of a sapphire substrate of approximately  $350 \,\mu$ m thickness, a 2-5 nm thick aluminium nitride (AlN) nucleation layer, a  $3 \,\mu$ m GaN channel, and a  $3 \,\text{nm} \,\text{Al}_{0.25}\text{Ga}_{0.75}\text{N}$  barrier layer. The structure was grown using molecular beam epitaxy (MBE) by SVT Assiciates, USA. The device consists of source and drain Ohmic contacts nominally overlapped by the gate contact and employs a gate dielectric. With no or low gate-to-source voltage(V<sub>GS</sub>), there is no two dimensional electron gas (2DEG) channel at the AlGaN/GaN interface to allow conduction of current between the drain and source contacts as the AlGaN barrier thickness is below the critical thickness required for the formation of such channel [55]. However, if a large enough positive bias voltage V<sub>GS</sub> is applied, it causes the formation of a quantum well at the AlGaN/GaN interface into

which electrons from the source and drain Ohmic regions are attracted (by the positive gate voltage), effectively creating a 2DEG channel, and so the structure is a normally-off field effect transistor. As the critical device feature required for normally off operation, the AlGaN layer, is grown epitaxially, and this eliminates the drawbacks experienced with previous techniques such as recess-etching in which the uniformity of the etch and hence that of the threshold voltages ( $V_{th}$ ) is difficult to achieve.  $V_{th}$  is set further by the thickness and properties of the gate dielectric.



FIGURE 6.1: Cross-section of the proposed e-mode AlGaN/GaN device.

Single finger 100  $\mu$ m wide devices were fabricated to test the device concept using process modules described in the earlier chapters. The source and drain Ohmic contacts were formed by photolithography, metal deposition using Ti/Al/Ni/Au metallization scheme, and lift-off. The sample was then annealed at 775 °C for 30 seconds in N<sub>2</sub> ambient and then a layer of SiO<sub>2</sub> was deposited by PECVD at 300 °C using the following conditions: RF power of 15 watts, pressure of 1 torr, flow rate (SiH<sub>4</sub>/N<sub>2</sub>O/N<sub>2</sub>) of 7/200/85 sccm, and a deposition rate of about 70 nm/min. The gate contacts were then formed by photolithography, metal deposition (Ni/Au, 50 nm/150 nm) and lift-off. Ohmic bond pad patterns were then defined by photolithography and the SiO<sub>2</sub> dielectric etched to expose the Ohmic contacts to complete the processing.

#### 6.1.2 Device Output and Transfer Characteristics

Results from typical devices from 2 samples are presented here. One sample employs a nominal 10 nm, while the other 20 nm thick SiO<sub>2</sub> dielectric. Both devices employ  $6\,\mu$ m long and 100  $\mu$ m wide gates. Output and transfer characteristics as well as the transconductance were measured using Agilent's B1500A Device Analyzer. Figure 6.2 shows the output characteristics of the device with 10 nm thick SiO<sub>2</sub> gate dielectric which demonstrated maximum drain current  $I_{DSmax}$ , of over 450 mA/mm at gate voltage  $V_{GS} = 6 V$  and drain voltage  $V_{DS} = 10 V$ .



FIGURE 6.2: Output characteristics of an e-mode  $6 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 10 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V, 1 V, 2 V, and 3 V.

Figure 6.3 shows the output characteristics of the device with 20 nm thick dielectric. This device demonstrated  $I_{DSmax}$  of over 650 mA/mm at  $V_{GS} = 6 V$  and  $V_{DS} = 8 V$ . The measured currents are comparable to those of conventional depletion-mode devices demonstrating the equally good properties of an induced 2DEG channel. A procedure to extract contact resistances for this non-conducting structure (when no 2DEG is present) that uses gated TLM structures is being developed, but the low knee voltages indicate good contacts.

The specific on-resistance  $R_{ON_{spec}}$  of the device was obtained by multiplying the area,  $(L_{SD} \times W_g)$  with the resistance (R=V/I), indicated by blue short-dash lines in graph) of the device in the linear region. Note that the specific on-resistance of the device would be higher if the transfer length of the device is included in the calculation as this would result in the area being calculated as  $[(L_{SD} + 2L_T) \times W_g]$ . For the device output characteristics shown in Figures 6.2 and 6.3, the specific on-resistances are calculated/extracted as  $43.5 \text{ m}\Omega \text{mm}^2$  and  $42.3 \text{ m}\Omega \text{mm}^2$ , respectively.

Transfer characteristics and transconductance at low drain bias voltages for the devices whose output characteristics are shown in Figures 6.2 and 6.3 are plotted in Figure 6.4. The threshold voltages are +3 V and +2 V (estimated at 1 mA/mm) for the 10 nm and



FIGURE 6.3: Output characteristics of an e-mode  $6 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric. Drain current traces for V<sub>GS</sub> of 0 V, 1 V, and 2 V are overlapping.

20 nm thick gate dielectric devices. The enhancement-mode nature of the devices can be seen from these characteristics. The drop in threshold voltage with higher thickness of the SiO<sub>2</sub> is attributed to the increase in the SiO<sub>2</sub> positive charge with increasing thickness, increased film stress of thicker deposited dielectric, etc. The devices are known to suffer from extra capacitance due to the added capacitance of the gate-source and gate-drain overlap of the device. The added capacitance will affect the switching speed of the gate-overlap devices as the switching time depends on the device capacitance as given in the equation ( $\tau = \text{RC}$ ), where  $\tau$  is the switching time, R is the resistance of the device, and C is the capacitance in the device. Therefore, increasing the gate-overlap length will result in an increase in capacitance.

#### 6.1.3 Offstate Breakdown Characteristics

Off-state ( $V_{GS} = 0 V$ ) breakdown characteristics are shown in Figure 6.5. The breakdown voltages were 9 V and 17 V for the devices with 10 nm and 20 nm thick SiO<sub>2</sub> gate dielectrics, respectively. The current rises sharply from about 1 mA/mm to over 1000 mA/mm (set by equipment compliance) on breakdown. The measured breakdown field of approximately 9 MV/cm suggests that device breakdown is limited by the dielectric thickness, as seen in devices of similar geometry [162].



FIGURE 6.4: Transfer characteristics and transconductance of e-mode AlGaN/GaN MOS-HEMTs at low drain bias voltages of 1 V and 3 V.



FIGURE 6.5: Breakdown characteristics of e-mode  $6\,\mu{\rm m}\times100\,\mu{\rm m}$  AlGaN/GaN MOS-HEMTs with 10 nm and 20 nm thick SiO\_2 gate dielectrics.

#### 6.1.4 Device Reliability

Devices with  $20 \text{ nm SiO}_2$  gate dielectric were assessed for stability. A stress test with  $I_{DS} = 300 \,\mathrm{mA/mm}$  and  $V_{DS} = 10 \,\mathrm{V}$  was performed at 10 minute intervals. Figure 6.6 shows the variation of the threshold voltage,  $V_{\rm th}$  with stress time.  $V_{\rm th}$  increased by  $0.25 \,\mathrm{V}$  within the first hour and continued to rise with stress time, and increased by approximately 1.25 V after 10 hours. The time-dependent shift of the threshold voltage in the device is attributed to the bulk traps in the gate dielectric and also interface traps located at the dielectric/semiconductor interface [166]. Unlike interface traps which are filled quickly, the bulk traps inside the gate dielectric can only be filled by the electrons tunnelling through the  $SiO_2$  layer from the interface. The farther the trap is located from the interface, the lower the probability of the tunnelling process is. It is demonstrated in Figure 6.6 that the time constant to fill the bulk traps inside  $SiO_2$ greatly influence the shift of threshold voltage as the traps are gradually filled under forward gate bias. In on-going work in the research group, device stability is being investigated via capacitance-voltage (CV) techniques for atomic layer deposited (ALD) high-k dielectrics namely alumina, hafnia and zirconia that are deemed the best suited for this device structure [167]. Note that similar instability is a present problem in other insulated gate GaN devices [166].



FIGURE 6.6: Variation of the threshold voltage of the proposed e-mode MOS-HEMT employing 20 nm thick SiO<sub>2</sub> gate dielectric with stress time.

# 6.2 Normally-off AlGaN/GaN MOS-HEMT on Silicon Substrate

#### 6.2.1 Device Structure and Fabrication

The proposed normally-off AlGaN/GaN HEMT was also realised from GaN epitaxial material grown on a silicon substrate due to its low cost and availability of bigger size (diameter) wafers. Devices with varying SiO<sub>2</sub> dielectric thickness were fabricated in order to test for the reproducibility and repeatability of our device fabrication techniques. This also allowed us to test the consistency of the device DC characteristics. Figure 6.7shows a cross-section of the device structure. It consists of a  $600 \,\mu\text{m}$  silicon substrate, nucleation and transition/buffer layers, a  $3 \,\mu m$  GaN channel, a  $3 \,nm$  Al<sub>0.25</sub>Ga<sub>0.75</sub>N barrier layer, and a 1 nm GaN cap layer. In AlGaN/GaN HEMTs with a high 2DEG sheet density, the 2D-electron lie very close to the AlGaN/GaN interface, which makes them very susceptible to any physical processes occurring at the interface such as interface roughness scattering, which is the dominant scattering mechanism at low temperatures. In order to reduce such affects, the separation between the 2D-carriers and the interface needs to be increased, which can be achieved if the AlGaN/GaN structures are capped with a GaN layer of appropriate thickness [168]. The GaN cap layer is also known to reduce gate leakage current in devices. The material was grown by metal organic chemical vapour deposition (MOCVD) by IQE Limited. Single finger devices of gate lengths of  $4\,\mu\text{m}$ ,  $6\,\mu\text{m}$ , and  $8\,\mu\text{m}$  and gate width of  $100\,\mu\text{m}$  were fabricated using the same process modules as in Section 6.1.1. A micrograph of a fabricated  $4 \,\mu m \times 100 \,\mu m$  device is shown in Figure 6.8.



FIGURE 6.7: Cross-section of the proposed e-mode AlGaN/GaN device on silicon substrate.



FIGURE 6.8: Scanning electron microscopy (SEM) image of normally-off AlGaN/-GaN device structure fabricated for DC measurements. Figure a) shows the top view of the gate, source and drain, Figure b) shows the image of the section which was cut in order to take transmission electron microscopy (TEM) images of the cross-section of the device, and Figure c) shows a zoomed in image of the section which was cut for TEM analysis

The cross-section of the device was further analysed using transmission electron microscopy (TEM) in order to check the step coverage of the gate and gate dielectric over the source and drain Ohmic metals of the device. The TEM image of the device crosssection is shown in Figure 6.9. It can be seen that the gate metal overlaps both the source and the drain metals of the device and the step coverage of both the gate and the dielectric over the source and drain Ohmic metals can clearly be seen on the zoomed images at the top of the image.

#### 6.2.2 Device Output and Transfer Characteristics

Results from typical devices from 2 samples are presented here. One sample employs a nominal 20 nm, while the other 30 nm thick SiO<sub>2</sub> dielectric. Device characteristics were measured using Agilent's B1500A Device Analyzer. Figures 6.10, 6.11, and 6.12 show the output characteristics of devices with gate lengths of 4  $\mu$ m, 6  $\mu$ m, and 8  $\mu$ m, respectively. These devices have a 20 nm nominal thick SiO<sub>2</sub> dielectric. The devices exhibited maximum drain currents of 240 mA/mm, 180 mA/mm, and 170 mA/mm, respectively, at gate voltage V<sub>GS</sub> = 6 V and drain voltage V<sub>DS</sub> = 10 V. For the device output characteristics shown in Figures 6.10, 6.11, and 6.12, the specific on-resistances are calculated/extracted as 71.4 mΩmm<sup>2</sup>, 13.3 mΩmm<sup>2</sup>, and 181.8 mΩmm<sup>2</sup>, respectively.



FIGURE 6.9: TEM image of cross-section of gate overlap device showing the dielectric step coverage



FIGURE 6.10: Output characteristics of an e-mode  $4\,\mu\rm{m}\times100\,\mu\rm{m}$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V and 1 V.



FIGURE 6.11: Output characteristics of an e-mode 6  $\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V, 1 V and 2 V.



FIGURE 6.12: Output characteristics of an e-mode  $8 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V, 1 V and 2 V.

Transfer characteristics and transconductance at linear ( $V_{DS} = 1 V$ ) and saturation ( $V_{DS} = 10 V$ ) regions for these devices are plotted in Figures 6.13, 6.14, and 6.15. The threshold voltages are +1 V, +1.5 V, and +1.5 V (estimated at 1 mA/mm), respectively, for these devices. The enhancement-mode nature of these devices can again be seen from these characteristics. The devices exhibited maximum transconductance ( $G_{m-max}$ ) of 60 mS/mm at  $V_{GS} = 5 V$  and  $V_{DS} = 10 V$ , 50 mS/mm at  $V_{GS} = 5 V$  and  $V_{DS} = 10 V$ , and 50 mS/mm at  $V_{GS} = 6 V$  and  $V_{DS} = 10 V$  for the devices with gate lengths of  $4 \mu m$ ,  $6 \mu m$ , and  $8 \mu m$ , respectively.



FIGURE 6.13: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $4 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 1 V and 10 V.

Figures 6.16, 6.17, and 6.18 show the output characteristics of devices with gate lengths of 4  $\mu$ m, 6  $\mu$ m, and 8  $\mu$ m, respectively. These devices have a 30 nm nominal thick SiO<sub>2</sub> dielectric. The devices exhibited maximum drain currents of 400 mA/mm, 130 mA/mm, and 105 mA/mm, respectively, at gate voltage V<sub>GS</sub> = 6 V and drain voltage V<sub>DS</sub> = 10 V. For the device output characteristics shown in Figures 6.16, 6.17, and 6.18, the specific on-resistances are calculated as 28.2 mΩmm<sup>2</sup>, 125 mΩmm<sup>2</sup>, and 166.7 mΩmm<sup>2</sup>, respectively.

Transfer characteristics and transconductance at linear ( $V_{DS} = 0.5 V$ ) and saturation ( $V_{DS} = 10 V$ ) regions for these devices are plotted in Figures 6.19, 6.20, and 6.21. The threshold voltages are +1 V, +1.5 V, and +1.5 V (estimated at 1 mA/mm), respectively, for these devices. The devices exhibited maximum transconductance of around



FIGURE 6.14: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $6 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 1 V and 10 V.



FIGURE 6.15: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $8 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 1 V and 10 V.

60 mS/mm, 50 mS/mm, and 50 mS/mm at a gate voltage of around 5 V and drain voltage of 10 V. The enhancement-mode nature of these devices can again be seen from these characteristics.



FIGURE 6.16: Output characteristics of an e-mode  $4 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V and 1 V.

Figure 6.22 shows the gate leakage current of the normally-off devices with gate lengths of  $4 \,\mu\text{m}$ ,  $6 \,\mu\text{m}$ , and  $8 \,\mu\text{m}$ , respectively for devices with a 20 nm nominal thick SiO<sub>2</sub> dielectric. It can be seen on the graph that the devices with gate lengths of  $4 \,\mu\text{m}$ ,  $6 \,\mu\text{m}$ , and  $8 \,\mu\text{m}$  demonstrate gate leakage currents of less than  $1 \,\mu\text{A/mm}$  at V<sub>GS</sub> of less than  $6 \,\text{V}$ ,  $5 \,\text{V}$ , and  $4.5 \,\text{V}$ , respectively. The devices therefore meet the required leakage current for power electronics application which is set to  $1 \,\mu\text{A/mm}$  [169–171].

## 6.3 Summary

A new high performance normally-off GaN-based MOS-HEMT that employs an ultrathin sub-critical 3 nm thick  $Al_{0.25}Ga_{0.75}N$  barrier layer and relies on an induced 2DEG for operation was presented in this chapter. Single finger devices were fabricated using 10 nm and 20 nm PECVD SiO<sub>2</sub> as the gate dielectric. These devices, grown on a sapphire substrate, demonstrated threshold voltages (V<sub>th</sub>) of +3 V and +2 V, and very high maximum drain currents (I<sub>DSmax</sub>) of over 450 mA/mm and 650 mA/mm, at a gate voltage (V<sub>GS</sub>) of 6 V, respectively.



FIGURE 6.17: Output characteristics of an e-mode 6  $\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V, 1 V and 2 V.



FIGURE 6.18: Output characteristics of an e-mode  $8 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric. Note the (near) zero drain currents for V<sub>GS</sub> of 0 V, 1 V and 2 V.



FIGURE 6.19: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $4 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 0.5 V and 10 V.



FIGURE 6.20: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $6\,\mu\text{m} \times 100\,\mu\text{m}$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 0.5 V and 10 V.



FIGURE 6.21: Transfer characteristics (left y-axis) and transconductance (right y-axis) of an e-mode  $8 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 30 nm thick SiO<sub>2</sub> dielectric at drain bias voltages of 0.5 V and 10 V.



FIGURE 6.22: Gate leakage current as functions of gate bias of e-mode AlGaN/GaN MOS-HEMTs with 20 nm thick  $SiO_2$  dielectric.

Single finger devices were also fabricated on GaN epitaxial material grown on a silicon substrate using 20 nm and 30 nm PECVD SiO<sub>2</sub> as the gate dielectric. The devices fabricated using a 20 nm SiO<sub>2</sub> gate dielectric had gate lengths (L<sub>g</sub>) of 4  $\mu$ m, 6  $\mu$ m, and 8  $\mu$ m and gate width of 100  $\mu$ m. They demonstrated threshold voltages of +1 V and +1.5 V, and +1.5 V, respectively, and maximum drain currents of over 240 mA/mm and 180 mA/mm, and 170 mA/mm at a gate voltage (V<sub>GS</sub>) of 6 V, respectively. The devices fabricated using a 30 nm SiO<sub>2</sub> gate dielectric had gate lengths of 4  $\mu$ m, 6  $\mu$ m, and 8  $\mu$ m and gate width of 100  $\mu$ m and they demonstrated threshold voltages of +0.8 V and maximum drain currents of over 400 mA/mm and 130 mA/mm, and 105 mA/mm at a gate voltage of 6 V, respectively.

The devices on sapphire substrate demonstrated higher drain currents than those on silicon substrate. MBE techniques used to grow the GaN on sapphire make it possible to grow high quality layers due to the slow deposition rate of the technique. On the other hand, silicon has a higher lattice mismatch to GaN (compared to sapphire), and this mismatch causes defects in the material thereby affecting the quality of the material. Also, GaN on silicon material generally tend to suffer from stress related defects due to the difference in coefficient of thermal expansion between GaN and silicon. As these materials are grown by MOCVD under high temperatures of between 800°C and 1000°C, stresses will build up from the difference in the thermal expansion coefficient between silicon and GaN. Therefore, higher dislocation defects in the material used for the devices grown on silicon substrates could have contributed to the lower drain currents exhibited in these devices compared to those grown by MBE on sapphire substrates.

The measured breakdown field of the devices suggest that device breakdown is limited by the dielectric thickness. Despite the low breakdown field of the devices, these devices could be used to replace the low-voltage enhancement-mode silicon transistor used to drive the high-voltage depletion-mode transistor in the hybrid high-voltage cascoded GaN switch configuration.

As the e-mode GaN device described here is very similar in operation to the silicon MOS-FET, a short description of the latter is provided in Appendix B for completeness. Also, devices similar to those described in this chapter but employing a  $SiN_x$  gate dielectric were fabricated and characterized. The results are presented in Appendix C.
# CHAPTER 7

# CONCLUSION AND FUTURE WORK

# 7.1 Conclusion

Due to its excellent material properties, GaN-based technology is set to become the dominant semiconductor in high-power and high-speed electronics. The research and development aimed at finding engineering solutions to gain the maximum potential from this material is gathering pace, and the work reported in this thesis is one of the many contributions to this goal. This thesis explored a novel high performance enhancement-mode GaN based MOS-HEMT that employs an ultrathin 3 nm Al<sub>0.25</sub>Ga<sub>0.75</sub>N barrier layer and relies on an induced 2DEG for operation. As the critical device feature required for normally-off operation, the AlGaN barrier layer, is grown epitaxially, this eliminates the drawbacks experienced with previous techniques such as recess-etching in which the uniformity of the etch and hence that of the threshold voltages is difficult to achieve.

Device fabrication of the new AlGaN/GaN MOS-HEMT is simple, requiring only easily repeatable and reproducible steps. The present device design could be used to realise high current low voltage power devices, operating at say 5 Å, required, for instance, for buck converters for mobile phones, tablets, laptop chargers, etc. Although these devices demonstrated high drain currents and high threshold voltages, the breakdown voltages were found to be very low for high power electronics devices. Nonetheless, the proposed design could also be used to replace the low-voltage depletion-mode transistor in the hybrid high-voltage cascoded GaN switch configuration [172]. The SiO<sub>2</sub> gate dielectric layers deposited by PECVD have a well known problem of non-uniformity, therefore, it is possible that this might have had an effect on the expected breakdown voltages of the devices. Therefore, an alternative gate dielectric for these devices might be to incorporate an atomic layer deposited dielectric such as  $HfO_2$  or  $Al_2O_3$  which is currently being investigated by the group. Not only have these dielectrics shown to improve device performance in the past [173, 174], but they are deposited in such a way that they are uniform across an entire wafer. Further work should be focused on a better understanding of the device operation, characterizing and mitigating the effects of oxide/interface charge and on realizing and characterizing larger gate periphery and high breakdown voltage devices.

# 7.2 Future Work and Research Recommendations

We have demonstrated a novel normally-off AlGaN/GaN HEMT which demonstrates high drain currents. However, further work on process optimization, device characterization, reliability and circuit integration is required in order for these devices to be considered for actual applications.

### 7.2.1 Ohmic Contacts

Using gated TLM structures, contact resistance of around 7  $\Omega$ mm and sheet resistances of approximately 557  $\Omega/\Box$  have been extracted for conventionally processed devices as those reported in this thesis [175]. It is expected that realising quality Ohmic contacts with very low contact resistance will improve the drain currents in the devices, therefore, further work is required to develop metal schemes for both Ohmic and gate contacts, which will enable devices to exhibit  $R_{sh}$  of below 500  $\Omega/\Box$  and  $R_c$  of better than 1  $\Omega$ mm in order for these devices to match the values currently demonstrated by their conventional normally-on counterparts. Future work should therefore be focused on better understanding and characterizing the gated linear and circular TLM structures.

### 7.2.2 All GaN Cascode Device Configuration

Future work should consider the integration of the normally-off AlGaN/GaN devices in a cascode configuration with a normally-on GaN HEMT in order to provide a highvoltage normally-off mode of operation. Currently, to use a normally-on GaN HEMT in circuit design, it is connected in cascode with a low-voltage silicon MOSFET [176–178]. Figure 7.1 illustrates a GaN HEMT in a cascode structure.

The operating principle of the cascode GaN HEMT is briefly illustrated as follows:

 Because the d-mode GaN HEMT's gate is connected to the source of the e-mode GaN HEMT, the V<sub>DS</sub> of the e-mode device becomes the negative V<sub>GS</sub> of the dmode GaN HEMT, providing the necessary negative voltage bias for turn-off when V<sub>GS</sub> reaches the threshold voltage required to turn the d-mode device off.



FIGURE 7.1: A cascode configuration featuring a high-voltage normally-on GaN HEMT with a low-voltage normally-off GaN HEMT.

• To turn the cascode structure on, the gate voltage of the e-mode device is increased until it reaches the threshold voltage which in turn lowers the d-mode device's source voltage until it reaches/goes beyond its threshold voltage thereby allowing the d-mode GaN HEMT to conduct current.

In general, controlling the on/off state of the low-voltage enhancement-mode device to control the on/off state of the high-voltage depletion-mode GaN HEMT makes the cascode GaN HEMT behave as a high-voltage enhancement-mode device.

The enhancement-mode device developed in this work seems suitable for cascode configuration with a high-voltage d-mode GaN device as it only requires an operating voltage of under 6 V [179]. Having an all GaN HEMT structure in the cascode configuration can increase the switching speed of the circuit and also enable the circuit to operate in harsh (high temperature) environments where the Si MOSFET might not operate efficiently. Also, the enhancement-mode GaN HEMT will also exhibit a lower on-resistance  $R_{on}$  than the normally-off Si MOSFET.

# Appendix A

# FABRICATION PROCESSES

# A.1 Sample Scribing

The wafer is usually coated with a thick resist film (s1818) in order to protect the sample surface from particles generated during scribing. The wafers are cleaved to small rectangular samples measuring approximately  $10 \text{ } mm \times 10 \text{ } mm$ .

# A.2 Sample Cleaning

- 1. Ultrasonic bath in acetone for  $10 \mathrm{mins}$
- 2. IPA for  $5 \min s$
- 3. Rinse with reverse osmosis (RO) water for 60 secs

# A.3 Ohmic Lithography Recipe 1

- 1. Spin S1818 at  $4000\,\mathrm{rpm}$  for  $120\,\mathrm{secs}$
- 2. Bake on hotplate for  $2\,{\rm mins}$  at  $65\,{}^{\circ}{\rm C}$
- 3. Develop in (1:1) Microposit : RO water for 60 secs
- 4. Rinse in RO water for 60 secs
- 5. Blow-dry with  $N_2$  gun
- 6. Expose with Mask Aligner (MA6) for 5 secs using vacuum contact
- 7. Develop in (1:1) Microposit : RO water for 75 secs with slight agitation

- 8. Rinse in RO water for 60 secs
- 9. Blow-dry with  $N_2$  gun
- 10. Ash in Gala Asher for  $3\,\mathrm{mins}$  at  $40\,\mathrm{W}$

# A.4 Ohmic Metallization Recipe 1

- 1. De-oxidise in  $H_2O$ : HCl using a ratio of 4:1 for 1 min
- 2. Deposit n-type ohmic contact of  $30\,\mathrm{nm}$  Ti/  $180\,\mathrm{nm}$  Al/  $40\,\mathrm{nm}$  Ni/  $100\,\mathrm{nm}$  Au using Plassys II or IV
- 3. Soak in warm acetone (50 °C) for 30 mins for lift-off
- 4. Agitate solution with a dropper to facilitate lift off and transfer to isopropyl alcohol (IPA)
- 5. Blow-dry with  $N_2$  gun
- 6. Anneal the sample at 775 °C for 30 secs in nitrogen ambient using a rapid thermal annealer (RTA)

# A.5 Gate Lithography Recipe 1

- 1. Spin S1805 at 4000 rpm for 120 secs
- 2. Bake on hotplate for 2 mins at  $65 \,^{\circ}\text{C}$
- 3. Develop in (1:1) Microposit : RO water for 60 secs
- 4. Rinse in RO water for 60 secs
- 5. Blow-dry with  $N_2$  gun
- 6. Expose with Mask Aligner (MA6) for 3 secs using vacuum contact
- 7. Develop in (1:1) Microposit : RO water for 75 secs with slight agitation
- 8. Rinse in RO water for 60 secs
- 9. Blow-dry with  $N_2$  gun
- 10. Ash in Gala Asher for  $3 \min at 40 W$

# A.6 Gate Metallization Recipe 1

- 1. Deposit 50 nm Ni/ $150\,\mathrm{nm}$  Au
- 2. Soak in warm acetone  $(50 \,^{\circ}\text{C})$  for 30 mins for lift-off
- 3. Agitate solution with a dropper to facilitate lift-off and transfer to IPA
- 4. Blow-dry with  $N_2$  gun

# A.7 Ohmic Lithography Recipe 2

- 1. Spin LOR 10A at 6000 rpm for 30 secs
- 2. Bake on hotplate at  $150\,^{\circ}\mathrm{C}$  for  $2\,\mathrm{mins}$
- 3. Spin S1818 at 4000 rpm for 30 secs
- 4. Bake on hotplate at  $115 \,^{\circ}\text{C}$  for  $3 \,\text{mins}$
- 5. Expose with Mask Aligner (MA6) for 5 secs using vacuum contact
- 6. Develop in Microposit MF319 developer for 2 mins 30 secs
- 7. Rinse in RO water for  $60 \sec s$
- 8. Blow-dry with  $N_2$  gun
- 9. Ash in Gala Asher for 5 mins at 60 W

### A.8 Ohmic Metallization Recipe 2

- 1. De-oxidise in  $H_2O$  : HCl 4:1 for 1 min
- 2. Rinse in RO water for 60 secs
- 3. Blow-dry with  $N_2$  gun
- 4. Deposit Ohmic metal contact of 30 nm Ti/ 180 nm Al/ 40 nm Ni/ 100 nm Au using Plassys II or IV
- 5. Soak in warm acetone  $(50 \,^{\circ}\text{C})$  for 15 mins for lift-off
- 6. Agitate solution with a dropper to facilitate lift off and transfer to IPA for 1 min
- 7. Blow-dry with  $N_2$  gun

- 8. Use the 1165 stripper to remove the LOR 10A in the 50  $^{\circ}\mathrm{C}$  hot water bath for 10 mins
- 9. Rinse in RO water for 60 secs
- 10. Blow-dry with  $N_2$  gun
- 11. Anneal sample at 775 °C for 30 secs in nitrogen ambient using a rapid thermal annealer (RTA)

# A.9 Gate Lithography Recipe 2

- 1. Spin LOR 10A at 6000 rpm for 30 secs
- 2. Bake on hotplate at  $150 \,^{\circ}\text{C}$  for  $2 \,\text{mins}$
- 3. Spin S1818 at 4000 rpm for 30 secs
- 4. Bake on hotplate at 115 °C for 3 mins
- 5. Expose with Mask Aligner (MA6) for 5 secs using vacuum contact
- 6. Develop in Microposit MF319 developer for 2 mins 30 secs
- 7. Rinse in RO water for 60 secs
- 8. Blow-dry with  $N_2$  gun
- 9. Ash in Gala Asher for  $5 \min at 60 W$

# A.10 Gate Metallization Recipe 2

- 1. Deposit gate metal contact of 50 nm Ni/ 150 nm Au using Plassys II or IV
- 2. Soak in warm acetone  $(50 \,^{\circ}\text{C})$  for 15 mins for lift-off
- 3. Agitate solution with a dropper to facilitate lift off and transfer to IPA for 1 min
- 4. Blow-dry with  $N_2$  gun
- 5. Use the 1165 stripper to remove the LOR 10A in the 50  $^{\circ}\mathrm{C}$  hot water bath for 10 mins
- 6. Rinse in RO Water for 60 secs
- 7. Blow-dry with  $N_2$  gun

# A.11 Bondpad Formation Recipe 2

- 1. Spin LOR 10A at 6000 rpm for 30 secs
- 2. Bake on hotplate at  $150 \,^{\circ}$ C for 2 mins
- 3. Spin S1818 at 4000 rpm for 30 secs
- 4. Bake on hotplate at  $115 \,^{\circ}$ C for  $3 \,\text{mins}$
- 5. Expose using Mask Aligner (MA6) for 5 secs using vacuum contact
- 6. Develop in Microposit MF319 developer for 2 mins 30 secs
- 7. Rinse in RO water for  $60 \sec s$
- 8. Blow-dry with  $N_2$  gun
- 9. Ash in Gala Asher for 5 mins at 60 W

# A.12 Bondpad Metallization Recipe 2

- 1. Deposit bond pad metal contact of 10 nm Ti/150 nm Au
- 2. Soak in warm acetone (50 °C) for 15 mins for lift-off
- 3. Agitate solution with a dropper to facilitate lift off and transfer to IPA for 1 min
- 4. Blow-dry with  $N_2$  gun
- 5. Use the 1165 stripper to remove the LOR 10A in the 50  $^{\circ}\mathrm{C}$  hot water bath for 10 mins
- 6. Rinse in RO Water for 60 secs
- 7. Blow-dry with  $N_2$

# A.13 E-mode AlGaN/GaN MISHEMT with Silicon Dioxide Gate Insulator Device Fabrication

- 1. Clean sample
- 2. Ohmic lithography
- 3. Ohmic metallization of 30 nm Ti/ 180 nm Al/ 40 nm Ni/ 100 nm Au using Plassys II or IV

- 4. Ohmic metal annealing at  $775 \,^{\circ}$ C for 30 secs in nitrogen ambient using a rapid thermal annealer (RTA)
- 5. Blanket PECVD dielectric deposition with SiH<sub>4</sub>, N<sub>2</sub>O and N<sub>2</sub> gases at flow rates of 7 sccm, 200 sccm and 85 sccm, respectively, at temperature of 300°C, RF power of 15 watts, and pressure of 1 torr using Oxford Instruments PECVD 80 Plus tool
- 6. Gate lithography
- 7. Gate metallization of 50 nm Ni/ 150 nm Au using Plassys II or IV
- 8. Bondpad lithography
- 9. Silicon dioxide dielectric etch with Oxford Instruments RIE 80 Plus tool using  $CHF_3$  and Ar gases at flow rates of 25 sccm and 18 sccm, RF power of 200 watts and a pressure of 20 millitorr at room temperature
- 10. Bond pad metallization of 10 nm Ti/ 150 nm Au using Plassys II or IV

# A.14 AlGaN/GaN MISHEMT with Silicon Nitride Gate Insulator Device Fabrication

Results from these devices are in Appendix C in Section C.1.

- 1. Clean sample thoroughly with acetone and IPA
- 2. Define and expose ohmic contacts
- 3. Metallise ohmic contacts using Ohmic recipe-1
- 4. Lift-off
- 5. Anneal at  $775 \,^{\circ}C$  for  $30 \sec n N_2$  ambient
- 6. Blanket PECVD deposition of SiN<sub>x</sub> with SiH<sub>4</sub>, NH<sub>3</sub> and N<sub>2</sub> gases at flow rates of 10 sccm, 12 sccm and 170 sccm, respectively, at temperature of 300°C, RF power of 20 watts, and pressure of 1 torr using Oxford Instruments PECVD 80 Plus tool
- 7. Define and expose gate contact
- 8. Metallise gate contact using gate recipe-1
- 9. Lift-off
- 10. Define and expose SiN etch layer

- 11. Dry etch SiN with the RIE 80+ tool using CHF<sub>3</sub> and O<sub>2</sub> gases at a flow rate of 50 sccm and 5 sccm, respectively, RF power of 200 watts and a pressure of 30 millitorr at room temperature
- 12. Clean photoresist from sample by putting in a cetone in the 50  $^{\circ}\mathrm{C}$  hotbath for 10 mins

# Appendix B

# SOME SEMICONDUCTOR DEVICE BASICS

A crucial part of making GaN-based devices is the formation of metal/semiconductor contacts, the Ohmic and gate metal contacts. In order to form good quality metal contacts, it is important to understand the physics behind how carriers are transported to and from the metal/semiconductor junctions. Therefore, some semiconductor device basics will be introduced in this section, including metal/semiconductor contacts and the carrier transport mechanisms across metal/semiconductor junctions.

# B.1 Metal/Semiconductor Contacts

Both Ohmic and Schottky (rectifying) contacts are of critical importance with respect to device functionality, performance and long term reliability. A great variety of metalsystems have been proposed for the formation of contacts on GaN-based devices with low-resistances and high thermal and electrical stability, which are discussed in sections 2.5.1 and 2.5.2.

The electrochemical potential (Fermi level) of any two solids in contact must be equal in thermal equilibrium. The difference between the Fermi energy and the vacuum level is known as the work function. The work function of a metal is denoted  $\phi_M$  and the work function of a semiconductor is denoted  $\phi_S$  as shown in Figure B.1a. Note that a higher work function corresponds lower energy level and vice versa. Therefore, when the metal and semiconductor are placed in contact with each other, electrons will flow from the material with the lower workfunction and higher energy level (semiconductor) to the material with the higher work function and lower energy level(metal) until the Fermi levels become equal, as shown in Figure B.1b. As a result of this, the lower work function material (semiconductor) becomes slightly more positively charged while the higher workfunction material (metal) becomes more negative. The resulting electrostatic potential is known as the built-in potential  $(V_{bi})$ . This contact potential will occur between any two solids and is the main cause of the occurrence of rectification in diodes. The built-in field is the cause of band-bending in the semiconductor near the junction [92].



FIGURE B.1: A metal/semiconductor junction

In order to cross the barrier, a carrier in the semiconductor must gain enough energy to move from the Fermi level to the top of the bent conduction band. The needed barrier surmounting energy  $\phi_B$  is the sum of the built-in potential and the offset between the Fermi level and the conduction band.

For n-type semiconductors,

$$\phi_B = \phi_M - \chi_S \tag{B.1}$$

where  $\chi_S$  is the semiconductor's electron affinity, defined to be the difference between the vacuum level and the conduction band level.

For p-type materials,

$$\phi_B = E_g - (\phi_M - \chi_S) \tag{B.2}$$

The depletion width of a contact is the length scale that the built-in field penetrates into the semiconductor.

$$W = \sqrt{\frac{2\epsilon V_{bi}}{\rho}} \tag{B.3}$$

and

$$X_d = \sqrt{\frac{2k\varepsilon_o\phi_i}{qN_d}} \tag{B.4}$$

W = depletion width  $\epsilon = \text{dielectric constant}$  $V_{bi} = \text{built-in potential}$   $\rho$  = charge density of semiconductor

 $N_d =$ doping concentration

From Equation B.4, it can be deduced that increasing the doping concentration will decrease the depletion layer width and vice versa. The barrier height (dependent on electron affinity and built-in field) and barrier thickness (dependent on built-in field, dielectric constant and doping density) can be modified by changing the metal used or changing the doping density of the semiconductor.

# B.2 Carrier Transport Mechanisms across Metal/Semiconductor Junctions

The main conduction mechanisms across metal/semiconductor interfaces/junctions are thermionic emission, field emission and thermionic field emission. The dominant transport mechanism is determined by the barrier height and barrier thickness of the metal/semiconductor junction.

### B.2.1 Thermionic Emission

The current density under thermionic emission  $(J_{TE})$  conditions is given by [92, 180]:

$$J_{TE} \propto exp(-\frac{E_{barrier}}{k_B T}) \tag{B.5}$$

 $k_B$  = Boltzmann's constant  $J_{TE}$  = Current density under thermionic emission  $E_{barrier}$  = Barrier Energy T = Absolute temperature

Under zero bias conditions:

$$E_{barrier} = q\phi_b \tag{B.6}$$

 $\phi_b = \text{Barrier height}$ 

q = Charge

and hence under constant temperature the current density is inversely exponentially dependent on the barrier height.

### B.2.2 Field Emission

The tunnelling probability is dependent on the barrier energy and width leading to tunnelling current density  $(J_{FE})$ [92].

$$J_{FE} \propto exp(\frac{E_{barrier}}{E_{00}}) \tag{B.7}$$

where  $E_{00}$  is the tunnelling parameter:

$$E_{00} = \frac{q\hbar}{2} \sqrt{\frac{N_D}{\epsilon_s m_e}} \tag{B.8}$$

 $m_e$  = Mass of electron  $\hbar$  = Reduced Planck's constant  $N_D$  = Doping density

From Equation B.8, it can be seen that tunnelling current depends directly on doping density. Since barrier width varies with dopant density, the tunnelling probability increases as the barrier thins. Annealing of the ohmic contacts in GaN devices creates a lot of nitrogen vacancies under the contact region and due to the presence of excessive carriers, the tunnelling probability/current increases, resulting in low resistance. In this, current will flow in equal magnitude from metal to semiconductor and vice versa under both forward and reverse bias.

### B.2.3 Thermionic Field Emission

Thermionic field emission (TFE) occurs where carriers have insufficient energy to cross the barrier thermally, and the barrier is too wide for direct tunnelling. In this case, the electrons to combine both thermionic and field emission (tunnelling) to crossover the barrier. Electrons with a degree of thermal excitation, however, may be able to tunnel through the barrier, since its width decreases with increasing energy, which results in the flow of thermionic field current density  $(J_{TFE})$  and is given by:

$$J_{TFE} \propto exp(\frac{E_{barrier}}{E_{00}coth(\frac{E_{00}}{k_{B}T})})$$
(B.9)

Thermionic field emission is therefore expected to increase with increasing temperature and is also strongly dependent on barrier thickness and dopant density.

# B.3 Metal Insulator Semiconductor Field Effect Transistor (MISFET)

Field Effect Transistors (FETs) employing a gate dielectric are generally called Metal Insulator-Semiconductor (MIS) FETs or MISFETs. The most common MISFET is the silicon Metal-Oxide-Semiconductor Field-Effect Transistor (MOSFET) so the name MOSFET is used much more frequently than MISFET. The basic principle of the MOS-FET is that the source-to-drain current ( $I_{SD}$ ) is controlled by the gate voltage, or better, by the gate to source electric field. The electric field induces charge (field effect) at the semiconductor-oxide interface.



FIGURE B.2: Structure of an n-channel silicon MOSFET.

The understanding of how an enhancement-mode MOSFET operates is helpful to the understanding of the operation of the normally-off gate overlapping HEMTs fabricated in this project. The similarities of a MOSFET and the gate overlap HEMT lie in the nominal overlap of the source and drain by the gate metal and also the gate dielectric used in the devices.

The cross-section physical structure of an enhancement-mode n-type metal-oxide semiconductor (NMOS)-Transistor is shown in Figure B.2. The transistor is fabricated on a p-type silicon substrate. The transistor has two heavily doped n-type regions. A thin (20 - 100 nm) layer of silicon dioxide (SiO<sub>2</sub>), which is an excellent electrical insulator, is grown on the surface of the substrate, covering the area between the source and drain regions. Metal is deposited on top of the oxide layer to form the gate electrode. Metal contacts are then made to the source and the drain.

### B.3.1 Operation of a MOSFET with No Gate Voltage

With no bias voltage applied to the gate, two back-to-back diodes exist in series between the drain and source (pn junctions). These back-to-back diodes prevent current conduction from the drain to source when a voltage  $V_{DS}$  is applied. The path between the drain and source has very high resistance (in the order of  $10^{12} \Omega$ )

# $V_{GS} + U_{GS} + U$

### B.3.2 Creating a Channel for Current Flow

FIGURE B.3: A MOSFET with a channel for current flow.

Consider the source and drain grounded and positive voltage is applied to the gate as shown in Figure B.3. Since the source is grounded,  $V_{GS}$  appears in effect between the gate and source. The positive voltage on the gate causes, in the first instance, the free holes (positively charged) to be repelled from the region under the gate (channel region). These holes are pushed downward into the substrate, leaving behind a carrier-free region, the depletion region which is populated by the bound negative charge associated with the acceptor atoms. The positive gate voltage also attracts electrons from the n<sup>+</sup> source and drain regions into the channel region. When a sufficient number of electrons accumulate near the surface under the gate, an n-type region is created, connecting the source and drain regions. Now if a voltage is applied between the drain and source, current flows through this induced n region, carried by the mobile electrons. The value of  $V_{GS}$  at which a sufficient number of mobile electrons accumulate in the channel region to form a conducting channel is called the threshold voltage ( $V_{th}$ ) and is usually between 1-3 V for silicon MOSFETs.

The gate metal and body (p-type silicon substrate) of the MOSFET can be considered to form a parallel-plate capacitor dielectric. The positive gate voltage causes positive charge

to accumulate on the top plate of the capacitor (gate electrode). The corresponding negative charge on the bottom plate is formed by the electrons in the induced channel. An electric field thus develops in the vertical direction. It is this field that controls the amount of charge in the channel, and thus, it determines the channel conductivity and, in turn, the current that will flow through the channel when a voltage  $V_{DS}$  is applied.





FIGURE B.4: A MOSFET with small drain-source voltage applied.

Having induced a channel, we now apply a positive voltage  $V_{DS}$  between the drain and source as shown in Figure B.4. First, consider the case where  $V_{DS}$  is small (0.1 or 0.2 V). The voltage  $V_{DS}$  causes a current  $I_D$  to flow through the induced channel. Current is carried by free electrons travelling from source to drain. The magnitude of  $I_D$  depends on the magnitude of  $V_{GS}$ . For  $V_{GS} = V_{th}$ , the channel is just induced and the current conducted is still negligibly small. As  $V_{GS}$  exceeds  $V_{th}$ , more electrons are attracted into the channel. The result is a channel of increased conductance or reduced resistance. The conductance of the channel is proportional to the excess gate voltage ( $V_{GS} - V_{th}$ ), also known as the effective voltage. It follows that the current  $I_D$  will be proportional to  $V_{GS} - V_{th}$  and to the voltage  $V_{DS}$  that causes  $I_D$  to flow. The current that leaves the source terminal ( $I_S$ ) is equal to the current that enters the drain terminal ( $I_D$ ), and the current ( $I_G = 0$ ).

### B.3.4 Operation at Higher Drain-Source Voltage

Let  $V_{GS}$  be held constant at a value greater than  $V_{th}$ .  $V_{DS}$  appears as a voltage drop across the length of the channel. That is, as we travel along the channel from source to drain, the voltage(measured relative to the source) increases from 0 to  $V_{DS}$ . Thus the voltage between D and points along the channel decreases from  $V_{GS}$  at the source end



FIGURE B.5: A MOSFET with high drain-source voltage applied.

to  $V_{GS} - V_{DS}$  at the drain end. Since the channel depth depends on this voltage, we find that the channel is no longer of uniform depth; rather, the channel will take a tapered form as shown in Figure B.5, being deepest at the source end and shallowest at the drain end. As  $V_{DS}$  is increased, the channel becomes more tapered and its resistance increases correspondingly. Thus the  $I_D - V_{DS}$  curve does not continue as a straight line but bends as shown in Figure B.6. Eventually, when  $V_{DS}$  is increased to the value that reduces the voltage between the gate and channel at the drain end to  $V_{th}$ , i.e.,  $V_{GS} - V_{DS} = 0$ or  $V_{DS} = V_{GS} - V_{th}$ , the channel depth at the drain end decreases to almost zero, and the channel is said to be pinched off. Increasing  $V_{DS}$  beyond this value has little effect (theoretically no effect) on the channel shape, and the current through the channel remains constant at the value reached for  $V_{DS} = V_{GS} - V_{th}$ . The drain current thus saturates at this value, and the MOSFET is said to have entered the saturation region of operation. The voltage  $V_{DS}$  at which saturation occurs is  $V_{DSsat}$ .

$$V_{DSsat} = V_{GS} - V_{th} \tag{B.10}$$

For every value of  $V_{GS} \ge V_{th}$ , there is a corresponding value of  $V_{DSsat}$ . The region of the  $I_D$ - $V_{DS}$  characteristic obtained for  $V_{DS} < V_{DSsat}$  is called the triode region.

# **B.4** Summary

The theory of metal/semiconductor contacts and the carrier transport mechanisms across these metal/semiconductor junctions have been described in this Appendix. In addition, the principle of operation of the common silicon metal-oxide-semiconductor



FIGURE B.6: IV Characteristic of an enhancement-type NMOS transistor.

field-effect transistors was presented since it is very similar to the proposed normally-off AlGaN/GaN MOSHEMTs.

# Appendix C

# SILICON NITRIDE GATE DIELECTRIC MISHEMTs AND INTERFACE TRAPS

Results of the gate overlap devices fabricated which employed silicon nitride  $(SiN_x)$  gate dilectric (refracrtive index of around 1.97) are presented here. This appendix also presents initial results for work being carried out on metal-oxide-semiconductor capacitor (MOSCAP) structures in order to understand the effect of interface trapped charges in the normally-off GaN devices presented in this thesis.

# C.1 AlGaN/GaN MISHEMT with SiN Gate Insulator

The same epitaxial material used for e-mode GaN devices was used for devices with a  $SiN_x$  gate dielectric. Figure C.1shows the cross-section of the device structure of the MISHEMT with a silicon nitride gate insulator/dielectric.

The device structure consists of a sapphire substrate of approximately 350  $\mu$ m thickness, a 2-5 nm thick aluminium nitride (AlN) nucleation layer, a 3  $\mu$ m GaN channel, and a 3 nm Al<sub>0.25</sub>Ga<sub>0.75</sub>N barrier layer. The epi-structure was grown using molecular beam epitaxy (MBE) by SVT Assiciates, USA. The device consists of source and drain Ohmic contacts nominally overlapped by the gate contact and employs a gate dielectric. The SiN<sub>x</sub> dielectric underneath the gate is 20 nm thick. Figure C.2 shows that the devices with gate length of 4  $\mu$ m and gate width of 100  $\mu$ m have a maximum drain current ( $I_{DSMAX}$ ) of approximately 175 mA/mm. The devices were measured at gate voltages ( $V_{GS}$ ) of between -3 and 4 V and drain voltages ( $V_{DS}$ ) of between 0 and 10 V. Figure C.3 shows the transfer characteristics of the devices. It is illustrated that the threshold voltage



FIGURE C.1: MISHEMT with  $SiN_x$  gate insulator.

 $(V_{th})$  of the devices is approximately -3 V which makes them normally-on transistors instead of the desired normally-off transistors.



FIGURE C.2: Output characteristics of a 3 nm thick barrier  $4\,\mu{\rm m}\times100\,\mu{\rm m}$  AlGaN/-GaN MOS-HEMT with 20 nm thick SiN dielectric.



FIGURE C.3: Transfer characteristics and  $G_m$  of a 3 nm thick barrier  $4 \,\mu m \times 100 \,\mu m$  AlGaN/GaN MOS-HEMT with 20 nm thick SiN dielectric.

# C.2 Metal-Oxide-Semiconductor Capacitors

Forming a stable interface with a semiconductor is a very important requirement for a gate dielectric material. The dielectric must be able to meet this requirement even after being put through the high temperatures that the wafer is exposed to during fabrication process and its density of interface states (traps) must be low. The former condition is met by many dielectrics, but the challenge is meeting the latter condition, which is a low density of interface states (traps). Interface states exist in a semiconductor material for many reasons, for instance, the energy band diagram of the ideal structure of a crystal consists of a conduction band and valence band separated by a band gap. The existence of defects or foreign atoms introduces discrete energy levels in the band gap [181]. For example, in the crystalline structure of any semiconductor, each atom requires a number of electrons from neighbouring atoms to be saturated at the surface of the semiconductor. When some of these bonds are missing due to the broken bonds, traps can be formed. Metal-oxide-semiconductor capacitor (MOSCAP) structures, also known as MOS diodes form the heart of any metal-oxide-semiconductor field-effect transistor (MOSFET), and can be used to determine the trap density at the interface of the dielectric and the semiconductor by performing capacitance-voltage (CV) characterization. Interface trapped charges are positive or negative charges, due to structural defects, oxidation-induced

defects, metal impurities, or other defects caused by radiation or similar bond breaking processes (e.g., hot electrons). A typical MOSCAP is similar to a MOSFET structure but instead of a source and drain it contains one Ohmic contact. It mainly consists of a dielectric layer (oxide) placed between a metal gate and a semiconductor layer. Realising a III-V MOSFET with an effect channel is affected by the type of dielectric used. For instance the relative permittivity of the dielectric must be high enough to minimise the gate leakage current and allow for device scaling.

In work co-supervised by the author of this thesis to further understand the effect of interface trapped charges the GaN MOS-HEMTs, MOSCAP structures were fabricated with 20 nm thick SiO<sub>2</sub> gate dielectric and characterized with three variations of process flows [182]. The first sample (sample 1) was realized using the standard process modules derived from the normally-off devices discussed in this thesis. The second sample (sample 2) also went through the standard e-mode fabrication processes but the dielectric surface was cleaned before and after it was deposited using 50 °C warm acetone for 5 minutes and isopropanol for 1 minute. The third sample (sample 3) employed a dual layer SiO<sub>2</sub> gate dielectric. The first 10 nm thick dielectric layer was annealed at 900 °C for 30 minutes in nitrogen ambient, then the deposited dielectric surface was cleaned using 50 °C warm acetone for 5 minutes then isopropanol for 1 minute. This was followed by the source and drain Ohmic metallization, deposition of the second 10 nm thick SiO<sub>2</sub> gate dielectric layer, gate metal formation, then dielectric etch and bondpad formation.

The graphs obtained from CV measurements are illustrated in Figure C.4, C.5, and C.6 for capacitors with a diameter size of  $250 \,\mu$ m. The devices show hysteresis due to trap states charging and discharging during the CV sweep. Sample 2 demonstrated a narrower shift between the forward and reverse sweep than sample 1, possibly due to surface cleaning with acetone and isopropanol. The surface donor-like traps are considered to be the main origin of the 2DEG channel, therefore surface cleaning is likely to have a reasonable effect on the devices behaviour as it is believed to get rid of unwanted surface traps. However, the CV for sample 3 demonstrated identical graphs with no shift between forward and reverse sweeping, which means there are no interface traps or trap states occupied in energy levels that can respond to CV sweeping. This is possibly due to the 900 °C dielectric annealing.

In Sample 1 and Sample 2, the forward C-V graph and the reverse C-V graph are not identical, but instead are shifted from each other. This phenomenon is called hysteresis, meaning that the C-V curve in the reverse sweep does not exactly follow the C-V curve in the forward sweep for the same value of bias voltage. This is because of the existence of interface traps at the semiconductor/dielectric interface. However, in Sample 3, the forward and reverse C-V graphs are identical, which is an indication of a high quality



FIGURE C.4: C-V graph of  $250 \,\mu\text{m}$  MOSCAP structure with  $20 \,\text{nm}$  SiO<sub>2</sub> gate dielectric without surface treatment prior to dielectric deposition (Sample 1).



FIGURE C.5: C-V graph of  $250 \,\mu\text{m}$  MOSCAP structure with  $20 \,\text{nm}$  SiO<sub>2</sub> gate dielectric with surface treatment prior to dielectric deposition (Sample 2).



FIGURE C.6: C-V graph of  $250 \,\mu\text{m}$  MOSCAP structure with  $20 \,\text{nm}$  SiO<sub>2</sub> gate dielectric with surface treatment prior to dielectric deposition and with dielectric treatment (annealing at 900 °C for 30 minutes in nitrogen ambient)(Sample 3).

dielectric-semiconductor interface. In the 1 KHz C-V graphs, in both the forward and reverse sweep, the transition from inversion to accumulation was sharp, another indicator of a good quality interface. Additionally, the hysteresis window between the forward and reverse sweeps was wider in Sample 1 than in Sample 2. Section 4.4 of this chapter goes into greater detail concerning the behaviour of C-V graphs.

From CV graphs for Sample 1 and Sample 2, it can be seen that there is a shift between the forward and reverse sweep, a phenomenon known as hysteresis, which is mainly due to charge and discharge of the existing traps. The response of the traps to the C-V sweep depends on the time for electron emission rate. Based on the Shockley-Read-Hall statistics (SRH) the energy level of the deepest trap state that can respond during CV sweeping can be estimated by Equation [181, 183]:

$$E_{Tm} = kT \ln(\sigma N_C v t_{meas}) \tag{C.1}$$

where, where k is the Boltzmann constant, T is the absolute temperature,  $\sigma$  is the capture cross section, v is the thermal velocity,  $N_c$  is the effective density of states in the conduction band, and  $t_{meas}$  is CV sweeping time. Therefore, only traps with an energy level above  $E_{Tm}$  can respond. All other traps are located in the frozen zone.

From the CV graphs, the hysteresis in Sample 2 is generally narrower than in Sample 1, which may be because the density of surface traps in the former is lower than in the latter. This reflects the importance of the surface treatment on the performance of the device. In Sample 3 not all the devices exhibit good performance, but encouragingly the forward and the reverse CV sweeps are identical and there was no hysteresis. Sample 3 was treated differently to the other samples with the dielectric layer deposited in two layers. The first dielectric layer was 10 nm thick and was deposited before Ohmic contact formation. After the first layer of dielectric deposition, the sample was annealed at 900  $^{\circ}$ C in N2 ambient for 30 minutes, which possibly assisted the removal of interface traps. The second 10 nm thick dielectric layer was deposited after the Ohmic contact metallisation, therefore the dielectric thickness over the sample surface was not uniform. The thickness of the dielectric layer gate and Ohmic overlap region is only 10 nm, while the thickness in other part become 20nm, the CV measurements for Sample 3 were performed in voltage sweeping range  $V_{GS}$  from -3 V to +3 V. The fabrication process in Sample 3 might lead to good e-mode AlGaN/GaN HEMTs, free of traps, however dividing the blanket dielectric into two equal layers of 10 nm probably reduces the break down voltage of the transistor. The first layer should be thinner than the second layer, in order to increase the thickness of the dielectric between the overlapped region between the Ohmic ring in MOS-capacitors (source, drain in transistors) and the gate, which should increase the capability of devices to stand higher voltages before break down. Further work is required to investigate the effect of interface trap charges on the reliability of the normally-off devices.

# C.3 Summary

Details of an AlGaN/GaN metal-insulator-semiconductor high-electron mobility transistor (MISHEMT) employing a silicon nitride  $SiN_x$  gate dielectric have been presented. These results demonstrate that normally-on mode of operation is achieved on the same material structure as the normally-off devices by changing the gate dielectric material used to fabricate the devices. It is believed that a different gate dielectric, for example,  $SiO_2$  depletes the 2DEG channel at the AlGaN/GaN interface of electrons required to conduct current when the device has no or low gate voltage applied to the gate due to the lower positive charge present in such a dielectric. A summary of MOSCAP structures being fabricated by the group to investigate the effect of interface trap charges on the reliability of the normally-off devices was also presented.

# Bibliography

- P. Siffert and E. Krimmel, Silicon: Evolution and future of a technology. Springer, 2004.
- [2] M. Adler, K. Owyang, B. Baliga, and R. Kokosa, "The evolution of power device technology," *IEEE Transactions on Electron Devices*, vol. 31, no. 11, pp. 1570 – 1591, Nov 1984.
- [3] "Wide band gap power electronics: A path toward CO<sub>2</sub> emission decrease," Yole Développement, Villeurbanne, France, Tech. Rep., July 2014.
- [4] B. Ozpineci, L. M. Tolbert, S. K. Islam, and M. Chinthavali, "Comparison of wide bandgap semiconductors for power applications," in *European Conference on Power Electronics and Applications*, 2003, pp. 2 – 4.
- [5] U. Mishra, P. Parikh, and Y.-F. Wu, "AlGaN/GaN HEMTs-an overview of device operation and applications," *Proceedings of the IEEE*, vol. 90, no. 6, pp. 1022 – 1031, Jun 2002.
- [6] R. Kemerley, H. Wallace, and M. Yoder, "Impact of wide bandgap microwave devices on DoD systems," *Proceedings of the IEEE*, vol. 90, no. 6, pp. 1059 – 1064, Jun 2002.
- [7] R. Quay, Gallium Nitride Electronics. Heidelberg: Springer.
- [8] F. A. Marino, N. Faralli, D. K. Ferry, S. M. Goodnick, and M. Saraniti, "Figures of merit in high-frequency and high-power GaN HEMTs," *Journal of Physics: Conference Series*, vol. 193, no. 1, p. 012040, 2009.
- [9] B. J. Baliga, "Power semiconductor device figure of merit for high-frequency applications," *IEEE Electron Device Letters*, vol. 10, no. 10, pp. 455–457, Oct 1989.
- [10] M. Shur and R. Davis, GaN-based materials and devices: growth, fabrication, characterization and performance. Singapore: World Scientific, 2004.

- [11] M. Rossberg, B. Vogler, and R. Herzer, "600V SOI gate driver IC with advanced level shifter concepts for medium and high power applications," in *European Conference on Power Electronics and Applications*. IEEE, 2007, pp. 1–8.
- [12] B. J. Baliga, Fundamentals of Power Semiconductor Devices. New York: Springer, 2008.
- W. C. Johnson, J. B. Parson, and M. C. Crew, "Nitrogen compounds of gallium. III," *The Journal of Physical Chemistry*, vol. 36, no. 10, pp. 2651–2654, 1931.
- [14] R. Juza and H. Hahn, "Über die Kristallstrukturen von Cu<sub>3</sub>N, GaN und InN Metallamide und Metallnitride," Zeitschrift für anorganische und allgemeine Chemie, vol. 239, no. 3, pp. 282–287, 1938.
- [15] H. P. Maruska and J. J. Tietjen, "The preparation and properties of vapordeposited single-crystalline GaN," *Applied Physics Letters*, vol. 15, no. 10, 1969.
- [16] H. P. Maruska, "A brief history of GaN blue light-emitting diodes," Office of Naval Research. National Compound Semiconductor Roadmap.
- [17] M. A. Khan, J. M. Van Hove, J. N. Kuznia, and D. T. Olson, "High electron mobility GaN/Al<sub>x</sub>Ga<sub>1-x</sub>N heterostructures grown by low pressure metalorganic chemical vapor deposition," *Applied Physics Letters*, vol. 58, no. 21, 1991.
- [18] M. Asif Khan, J. N. Kuznia, A. R. Bhattarai, and D. T. Olson, "Metal semiconductor field effect transistor based on single crystal GaN," *Applied Physics Letters*, vol. 62, no. 15, 1993.
- [19] M. A. Khan, J. Kuznia, D. T. Olson, W. Schaff, J. Burm, and M. Shur, "Microwave performance of a 0.25 μm gate AlGaN/GaN heterostructure field effect transistor," *Applied Physics Letters*, vol. 65, no. 9, pp. 1121–1123, Aug 1994.
- [20] S. Kaiser, M. Jakob, J. Zweck, W. Gebhardt, O. Ambacher, R. Dimitrov, A. Schremer, J. Smart, and J. Shealy, "Structural properties of AlGaN/GaN heterostructures on Si(111) substrates suitable for high-electron mobility transistors," *Journal of Vacuum Science Technology B: Microelectronics and Nanometer Structures*, vol. 18, no. 2, pp. 733–740, Mar 2000.
- [21] S. Tripathy, V. K. X. Lin, S. B. Dolmanan, J. P. Y. Tan, R. S. Kajen, L. K. Bera, S. L. Teo, M. K. Kumar, S. Arulkumaran, G. I. Ng, S. Vicknesh, S. Todd, W. Z. Wang, G. Q. Lo, H. Li, D. Lee, and S. Han, "AlGaN/GaN two-dimensionalelectron gas heterostructures on 200mm diameter Si(111)," *Applied Physics Letters*, vol. 101, no. 8, p. 082110, 2012.

- [22] H. G. Grimmeiss and H. Koelmans, "Über die Kantenemission und andere Emissionen des GaN," Z. Naturforschg, no. 14a, pp. 264–271, 1959.
- [23] J. Pankove, E. Miller, D. Richman, and J. Berkeyheiser, "Electroluminescence in GaN," *Journal of Luminescence*, vol. 4, no. 1, pp. 63 – 66, 1971.
- [24] J. Pankove, E. A. Miller, and J. E. Berkeyheiser, "GaN electroluminescent diodes," in *International Electron Devices Meeting*, vol. 17, 1971, p. 78.
- [25] H. Maruska, W. Rhines, and D. Stevenson, "Preparation of Mg-doped GaN diodes exhibiting violet electroluminescence," *Materials Research Bulletin*, vol. 7, no. 8, pp. 777 – 781, 1972.
- [26] H. Amano, N. Sawaki, I. Akasaki, and Y. Toyoda, "Metalorganic vapor phase epitaxial growth of a high quality GaN film using an AlN buffer layer," *Applied Physics Letters*, vol. 48, no. 5, 1986.
- [27] H. Amano, M. Kito, K. Hiramatsu, and I. Akasaki, "P-Type conduction in Mgdoped GaN treated with low-energy electron beam irradiation (LEEBI)," *Japanese Journal of Applied Physics*, vol. 28, no. 12A, p. L2112, 1989.
- [28] S. Nakamura, "In situ monitoring of GaN growth using interference effects," Japanese Journal of Applied Physics, vol. 30, no. 8R, p. 1620, 1991.
- [29] S. Nakamura, M. Senoh, and T. Mukai, "Highly P-typed Mg-doped GaN films grown with GaN buffer layers," *Japanese Journal of Applied Physics*, vol. 30, no. 10A, p. L1708, 1991.
- [30] A. Bykhovski, B. Gelmont, and M. Shur, "The influence of the strain-induced electric field on the charge distribution in GaN-AlN-GaN structure," *Journal of Applied Physics*, vol. 74, no. 11, 1993.
- [31] A. Ozgur, W. Kim, Z. Fan, A. Botchkarev, A. Salvador, S. Mohammad, B. Sverdlov, and H. Morkoc, "High transconductance normally-off GaN MOD-FETs," *Electronics Letters*, vol. 31, no. 16, pp. 1389–1390, Aug 1995.
- [32] P. Asbeck, E. Yu, S. Lau, G. Sullivan, J. Van Hove, and J. Redwing, "Piezoelectric charge densities in AlGaN/GaN HFETs," *Electronics letters*, vol. 33, no. 14, pp. 1230–1231, 1997.
- [33] F. Ren, M. Hong, S. N. G. Chu, M. A. Marcus, M. J. Schurman, A. Baca, S. J. Pearton, and C. R. Abernathy, "Effect of temperature on Ga<sub>2</sub>O<sub>3</sub>(Gd<sub>2</sub>O<sub>3</sub>)/ GaN metal-oxide-semiconductor field-effect transistors," *Applied Physics Letters*, vol. 73, no. 26, 1998.

- [34] S. Yoshida and J. Suzuki, "High-temperature reliability of GaN metal semiconductor field-effect transistor and bipolar junction transistor," *Journal of Applied Physics*, vol. 85, no. 11, 1999.
- [35] A. P. Zhang, G. T. Dang, F. Ren, J. Han, A. G. Baca, R. J. Shul, H. Cho, C. Monier, X. A. Cao, C. R. Abernathy, and S. J. Pearton, "Direct-current characteristics of pnp AlGaN/GaN heterojunction bipolar transistors," *Applied Physics Letters*, vol. 76, no. 20, 2000.
- [36] F. Semond, P. Lorenzini, N. Grandjean, and J. Massies, "High-electron-mobility AlGaN/GaN heterostructures grown on Si(111) by molecular-beam epitaxy," *Applied Physics Letters*, vol. 78, no. 3, 2001.
- [37] Y. Cai, Y. Zhou, K. Lau, and K. Chen, "Control of threshold voltage of Al-GaN/GaN HEMTs by fluoride-based plasma treatment: From depletion mode to enhancement mode," *IEEE Transactions on Electron Devices*, vol. 53, no. 9, pp. 2207–2215, Sept. 2006.
- [38] Y. Uemoto, M. Hikita, H. Ueno, H. Matsuo, H. Ishida, M. Yanagihara, T. Ueda, T. Tanaka, and D. Ueda, "Gate Injection Transistor (GIT); A normally-off Al-GaN/GaN power transistor using conductivity modulation," *IEEE Transactions* on Electron Devices, vol. 54, no. 12, pp. 3393–3399, Dec 2007.
- [39] Y. Zhang, M. Sun, S. J. Joglekar, T. Fujishima, and T. Palacios, "Threshold voltage control by gate oxide thickness in fluorinated GaN metal-oxide-semiconductor high-electron-mobility transistors," *Applied Physics Letters*, vol. 103, no. 3, pp. –, 2013.
- [40] J. Dharma, A. Pisal, and C. Shelton, "Simple method of measuring the band gap energy value of TiO<sub>2</sub> in the powder form using a UV/Vis/NIR spectrometer," *Application Note.*
- [41] R. Kolbas, I. Shmagin, and J. Muth, "Optical properties of wide bandgap III-V nitride semiconductors," in 5th International Conference on Solid-State and Integrated Circuit Technology, 1998, pp. 609–612.
- [42] P. Roussel, "Will GaN-on-Si displace Si and SiC in power electronics?" in Proceeding of the 2011 International Conference on Compound Semiconductor Manufacturing Technology (CS MANTECH), May 2011, pp. 16–19.
- [43] C. Blake, "GaN ransistors reach the 600V operation plateau," in *PET innovations: Power Electronics Technology.* Transphorm Inc., June 2011, pp. 38 – 41.

- [44] Markets and Markets, "Gallium Nitride (GaN) Semiconductor Devices Market: Global Industry Analysis, Size, Share, Growth, Trends and Forecast 2013 - 2019."
  [Online]. Available: www.marketsandmarkets.com
- [45] "GaN technologies for power electronic applications: Industry and market status & forecasts," Yole Développement, Villeurbanne, France, Tech. Rep., 2014.
- [46] M. Scott, J. Li, and J. Wang, "Applications of gallium nitride in power electronics," in *IEEE Power and Energy Conference at Illinois (PECI)*, Feb 2013, pp. 1–7.
- [47] J. Millan, P. Godignon, X. Perpina, A. Perez-Tomas, and J. Rebollo, "A survey of wide bandgap power semiconductor devices," *IEEE Transactions on Power Electronics*, vol. 29, no. 5, pp. 2155–2163, May 2014.
- [48] T. Kachi, D. Kikuta, and T. Uesugi, "GaN power device and reliability for automotive applications," in *IEEE International Reliability Physics Symposium (IRPS)*, April 2012, pp. 3D.1.1–3D.1.4.
- [49] T. Oka, Y. Ueno, T. Ina, and K. Hasegawa, "Vertical GaN-based trench metal oxide semiconductor field-effect transistors on a free-standing GaN substrate with blocking voltage of 1.6 kV," *Applied Physics Express*, vol. 7, no. 2, p. 021002, 2014.
- [50] T. Oka and T. Nozawa, "AlGaN/GaN recessed MIS-Gate HFET with highthreshold-voltage normally-off operation for power electronics applications," *IEEE Electron Device Letters*, vol. 29, no. 7, pp. 668–670, July 2008.
- [51] Q. Si, H. Yue, M. Xiaohua, X. Yuanbin, and M. Jigang, "Enhancement-mode AlGaN/GaN HEMTs fabricated by fluorine plasma treatment," *Journal of Semi*conductors, vol. 30, no. 12, p. 124002, 2009.
- [52] S. V. Novikov, N. Zainal, A. V. Akimov, C. R. Staddon, C. T. Foxon, and A. J. Kent, "Zinc-blende (Cubic) GaN and AlGaN layers, structures and bulk crystals by molecular beam epitaxy," *AIP Conference Proceedings*, vol. 1292, no. 1, pp. 154–158, 2010.
- [53] A. Trampert, O. Brandt, and K. Ploog, "Chapter 7: Crystal Structure of Group {III} Nitrides," in *Gallium Nitride (GaN)*, ser. Semiconductors and Semimetals, J. I. Pankove and T. D. Moustakas, Eds. Elsevier, 1997, vol. 50, pp. 167 – 192.
- [54] F. Bernardini, V. Fiorentini, and D. Vanderbilt, "Spontaneous polarization and piezoelectric constants of III-V Nitrides," *Physical Review B*, vol. 56, pp. R10024– R10027, Oct 1997.

- [55] O. Ambacher, J. Smart, J. R. Shealy, N. G. Weimann, K. Chu, M. Murphy, W. J. Schaff, L. F. Eastman, R. Dimitrov, L. Wittmer, M. Stutzmann, W. Rieger, and J. Hilsenbeck, "Two-dimensional electron gases induced by spontaneous and piezo-electric polarization charges in N- and Ga-face AlGaN/GaN heterostructures," *Journal of Applied Physics*, vol. 85, pp. 3222–3233, Mar. 1999.
- [56] O. Ambacher, B. Foutz, J. Smart, J. Shealy, N. Weimann, K. Chu, M. Murphy, A. Sierakowski, W. Schaff, L. Eastman *et al.*, "Two dimensional electron gases induced by spontaneous and piezoelectric polarization in undoped and doped Al-GaN/GaN heterostructures," *Journal of Applied Physics*, vol. 87, no. 1, pp. 334– 344, 2000.
- [57] M. Shur, G. Simin, S. Rumyantsev, R. Jain, and R. Gaska, "Chapter 13: Insulated gate nitride-based field effect transistors," in *Fundamentals of III-V Semiconductor MOSFETs.* Springer, 2010, pp. 379–422.
- [58] Y.-J. Lee, Y.-C. Yao, C.-Y. Huang, T.-Y. Lin, L.-L. Cheng, C.-Y. Liu, M.-T. Wang, and J.-M. Hwang, "High breakdown voltage in AlGaN/GaN HEMTs using AlGaN/GaN/AlGaN quantum-well electron-blocking layers," *Nanoscale Research Letters*, vol. 9, no. 1, p. 433, 2014.
- [59] J. Ibbetson, P. Fini, K. Ness, S. Denbaars, J. Speck, and U. Mishra, "Polarization effects, surface states, and the source of electrons in AlGaN/GaN heterostructure field effect transistors," *Applied Physics Letters*, vol. 77, p. 250, Jul. 2000.
- [60] P. Xu, Y. Jiang, Y. Chen, Z. Ma, X. Wang, Z. Deng, Y. Li, H. Jia, W. Wang, and H. Chen, "Analyses of 2-DEG characteristics in GaN HEMT with AlN/GaN super-lattice as barrier layer grown by MOCVD," *Nanoscale research letters*, vol. 7, no. 1, pp. 1–6, 2012.
- [61] C. Wang, X. Wang, G. Hu, J. Wang, and J. Li, "Influence of Al content on electrical and structural properties of Si-doped Al<sub>x</sub>Ga<sub>1x</sub>N/GaN HEMT structures," *Physica Status Solidi* (C), vol. 3, no. 3, pp. 486–489, 2006.
- [62] L. Nguyen, L. Larson, and U. Mishra, "Ultra-high speed modulation-doped fieldeffect transistors: A Tutorial Review," *Proceedings of the IEEE*, vol. 80, no. 4, pp. 494 –518, Apr 1992.
- [63] J. N. K. M. Asif Khan, A. Bhattarai and D. T. Olson, "High electron mobility transistor based on a GaN/Al<sub>x</sub>Ga<sub>1-x</sub>N heterojunction," *Applied Physics Letters*, 1993.

- [64] S. Sheppard, K. Doverspike, W. Pribble, S. Allen, J. Palmour, L. Kehias, and T. Jenkins, "High-power microwave GaN/AlGaN HEMTs on semi-insulating silicon carbide substrates," *IEEE Electron Device Letters*, vol. 20, no. 4, pp. 161 –163, April 1999.
- [65] I. Daumiller, C. Kirchner, K. Kamp, K. Ebeling, L. Pond, C. Weitzel, and E. Kohn, "Evaluation of AlGaN-GaN HFETs up to 750 °C," in 56th Annual Device Research Conference Digest, 1998., Jun 1998, pp. 114 –115.
- [66] S. Taking, "AlN/GaN MOS-HEMTs Technology," Ph.D. dissertation, University of Glasgow, 2012.
- [67] R. F. Pierret et al., Semiconductor device fundamentals. Reading, Massachusetts: Oxford University Press, 2013.
- [68] M. Germain, M. Leys, S. Boeykens, S. Degroote, W. Wang, D. Schreurs, W. Ruythooren, K.-H. Choi, B. V. Daele, G. V. Tendeloo, and G. Borghs, "High electron mobility in AlGaN/GaN HEMT grown on sapphire: strain modification by means of AlN interlayers," in *MRS Proceedings*, vol. 798. Cambridge Univ Press, 2003, pp. Y10–22.
- [69] Properties of group III Nitrides.
- [70] Y. Ohno and M. Kuzuhara, "Application of GaN-based heterojunction FETs for advanced wireless communication," *IEEE Transactions on Electron Devices*, vol. 48, no. 3, pp. 517 –523, Mar 2001.
- [71] H.M. and Manasevit, "The use of metalorganics in the preparation of semiconductor materials: Growth on insulating substrates," *Journal of Crystal Growth*, vol. 13-14, no. 0, pp. 306 – 314, 1972.
- [72] A. Rinta-Mykky, P. Laukkanen, S. Lehkonen, S. Laaksonen, J. Dekker, A. Tukiainen, P. Uusimaa, and M. Pessa, "Plasma-assisted MBE growth of GaN on HVPE-GaN substrates," *Physica Status Solidi* (a), vol. 176, no. 1, pp. 465–468, 1999.
- [73] C. Abernathy, J. MacKenzie, and S. Donovan, "Growth of Group III nitrides by metalorganic molecular beam epitaxy," *Journal of Crystal Growth*, vol. 178, no. 1-2, pp. 74 – 86, 1997.
- [74] R. Singh, J. Cooper, J.A., M. Melloch, T. Chow, and J. Palmour, "SiC power Schottky and PiN diodes," *IEEE Transactions on Electron Devices*, vol. 49, no. 4, pp. 665–672, Apr 2002.

- [75] M. Rosker, "The wide and the narrow: DARPA/MTO programs for RF applications in wide bandgap and antimonide-based semiconductors," in *IEEE Compound Semiconductor Integrated Circuit Symposium*, 2005, p. 4.
- [76] —, "Recent Advances in GaN-on-SiC HEMT Reliability and Microwave Performance within the DARPA WBGS-RF Program," in *IEEE Compound Semiconductor Integrated Circuit Symposium*, Oct. 2007, pp. 1–4.
- [77] IQE plc, "IQE and II-VI Inc. launch 150 mm GaN HEMT epi wafers on SiC substrates," May 2013, Accessed: 13th February, 2015. [Online]. Available: http://www.iqep.com/news-2013/May\_13\_13b.asp
- [78] Acquire Media, "RFMD(R) Introduces World's First 6-Inch GaN-on-SiC Wafers for RF Power Transistors," Sept 2013, Accessed: 13th February, 2015. [Online]. Available: http://ir.rfmd.com/releasedetail.cfm?releaseid=791678
- [79] J. Palmour, S. Sheppard, R. Smith, S. Allen, W. Pribble, T. Smith, Z. Ring, J. Sumakeris, A. Saxler, and J. Milligan, "Wide bandgap semiconductor devices and MMICs for RF power applications," in *International Electron Devices Meeting* (*IEDM*). Technical Digest, 2001, pp. 17.4.1 –17.4.4.
- [80] LEDs Magazine, "Rubicon ships 200,000th 6-inch sapphire wafer," 2014, Accessed: 13th February, 2015. [Online]. Available: http://www.ledsmagazine. com/articles/2011/10/rubicon-ships-200-000th-6-inch-sapphire-wafer.html
- [81] Compound Semiconductor, "Vertical integration streamlines sapphire production." [Online]. Available: http://www.compoundsemiconductor.net/article/ 90678-vertical-integration-streamlines-sapphire-production.html
- [82] Thomas Publishing Company, "NTT Advanced Technology Corporation: GaN epitaxial wafers," February 2015, Accessed: 13th February, 2015. [Online]. Available: http://www.thomasnet.com/productsearch/ item/20093352-15139-1003-1004/ntt-advanced-technology-corporation/ gan-epitaxial-wafers/#gan-epi-product-lineup
- [83] Y. Ando, Y. Okamoto, H. Miyamoto, N. Hayama, T. Nakayama, K. Kasahara, and M. Kuzuhara, "A 110-W AlGaN/GaN heterojunction FET on thinned sapphire substrate," in *International Electron Devices Meeting (IEDM)*. Technical Digest, 2001, pp. 17.3.1 –17.3.4.
- [84] A. Chini, D. Buttari, R. Coffie, L. Shen, S. Heikman, A. Chakraborty, S. Keller, and U. Mishra, "Power and linearity characteristics of field-plated recessed-gate AlGaN/GaN HEMTs," *IEEE Electron Device Letters*, vol. 25, no. 5, pp. 229 – 231, May 2004.

- [85] Electronics 360, "GaN-on-Si to gain ground in LEDs," December 2013. [Online]. Available: http://electronics360.globalspec.com/article/3825/ gan-on-si-to-gain-ground-in-leds
- [86] J. Johnson, E. Piner, A. Vescan, R. Therrien, P. Rajagopal, J. Roberts, J. Brown, S. Singhal, and K. Linthicum, "12 W/mm AlGaN-GaN HFETs on silicon substrates," *IEEE Electron Device Letters*, vol. 25, no. 7, pp. 459 – 461, July 2004.
- [87] K. Chu, P. Chao, M. Pizzella, R. Actis, D. Meharry, K. Nichols, R. Vaudo, X. Xu, J. Flynn, J. Dion, and G. Brandes, "9.4-W/mm power density AlGaN-GaN HEMTs on free-standing GaN substrates," *IEEE Electron Device Letters*, vol. 25, no. 9, pp. 596 – 598, Sept. 2004.
- [88] S. J. Pearton, J. C. Zolper, R. J. Shul, and F. Ren, "GaN: Processing, defects, and devices," *Journal of Applied Physics*, vol. 86, pp. 1–78, Jul. 1999.
- [89] E. H. Rhoderick and R. H. Williams, Metal-semiconductor contacts, 2nd ed. Oxford [England] : Clarendon Press, 1988.
- [90] L. Jae-Seung, K. Jong-Wook, L. Jae-Hak, S. Jin-Ho, H. Sung-Ho, L. Yong-Hyun, L. Jung-Hee, S. Moo-Whan, K. Chant-Seok, and O. Jae-Eung, "Optimization of Ti/Al/Pt/Au ohmic contact and RuO<sub>2</sub>-based schottky contact in AlGaN/GaN HFET for high power and high temperature operation," *Journal of the Korean Physical Society*, vol. 39, no. SUP, pp. S385–S388, 2001.
- [91] D. Schrade-Köhn, P. Leber, R. Behtash, H. Blanck, and H. Schumacher, "Change of the material properties of Ni, Pt and Au thin films and thin film stacks for GaN Schottky contacts during thermal processing," *Semiconductor Science and Technology*, vol. 25, no. 9, p. 095009, 2010.
- [92] S. Sze, *Physics of Semiconductor Devices*, 2nd ed., 1981.
- [93] N. Chaturvedi, U. Zeimer, J. Wrfl, and G. Trnkle, "Mechanism of Ohmic contact formation in AlGaN/GaN high electron mobility transistors," *Semiconductor Science and Technology*, vol. 21, no. 2, p. 175, 2006.
- [94] A. Vertiatchikh, E. Kaminsky, J. Teetsov, and K. Robinson, "Structural properties of alloyed Ti/Al/Ti/Au and Ti/Al/Mo/Au ohmic contacts to AlGaN/GaN," Solid-State Electronics, vol. 50, no. 7-8, pp. 1425 – 1429, 2006.
- [95] D. Selvanathan, "Comparative study of Ti/Al/Mo/Au, Mo/Al/Mo/Au, and V/Al/Mo/Au ohmic contacts to AlGaN/GaN heterostructures," Journal of Vacuum Science Technology B: Microelectronics and Nanometer Structures, vol. 22, p. 2409, 2004.

- [96] D. Selvanathan, L. Zhou, V. Kumar, and I. Adesida, "Low resistance Ti/Al/-Mo/Au Ohmic contacts for AlGaN/GaN heterostructure field effect transistors," *Physica Status Solidi* (a), vol. 194, no. 2, pp. 583–586, 2002.
- [97] F. M. Mohammed, L. Wang, D. Selvanathan, H. Hu, and I. Adesida, "Ohmic contact formation mechanism of Ta/ Al/ Mo/ Au and Ti/ Al/ Mo/ Au metallizations on AlGaN/ GaN HEMTs," *Journal of Vacuum Science & Technology B*, vol. 23, no. 6, pp. 2330–2335, 2005.
- [98] D. Qiao, L. S. Yu, L. Jia, P. M. Asbeck, S. S. Lau, and T. E. Haynes, "Transport properties of the advancing interface Ohmic contact to AlGaN/GaN heterostructures," *Applied Physics Letters*, vol. 80, no. 6, 2002.
- [99] E. H. Rhoderick and R. Williams, *Metal-semiconductor contacts*. Clarendon Press Oxford, 1988.
- [100] G. Reeves and H. Harrison, "Obtaining the specific contact resistance from transmission line model measurements," *IEEE Electron Device Letters*, vol. 3, no. 5, pp. 111 – 113, May 1982.
- [101] G. Reeves, "Specific contact resistance using a circular transmission line model," Solid-State Electronics, vol. 23, no. 5, pp. 487–490, 1980.
- [102] C. Xu, J. Wang, M. Wang, H. Jin, Y. Hao, and C. P. Wen, "Reevess circular transmission line model and its scope of application to extract specific contact resistance," *Solid-State Electronics*, vol. 50, no. 5, pp. 843–847, 2006.
- [103] G. Reeves, "Specific contact resistance using a circular transmission line model," Solid-State Electronics, vol. 23, no. 5, pp. 487 – 490, 1980.
- [104] T. Arai, H. Sueyoshi, Y. Koide, M. Moriyama, and M. Murakami, "Development of Pt-based ohmic contact materials for p-type GaN," *Journal of Applied Physics*, vol. 89, no. 5, 2001.
- [105] J. Klootwijk and C. Timmering, "Merits and limitations of circular TLM structures for contact resistance determination for novel III-V HBTs," in *The International Conference on Microelectronic Test Structures*, March 2004, pp. 247–252.
- [106] G. S. Marlow and M. B. Das, "The effects of contact size and non-zero metal resistance on the determination of specific contact resistance," *Solid-State Electronics*, vol. 25, no. 2, pp. 91 – 94, 1982.
- [107] B. Jacobs, M. Kramer, E. J. Geluk, and F. Karouta, "Optimisation of the Ti/Al/Ni/Au ohmic contact on AlGaN/GaN FET structures," *Journal of Crystal Growth*, vol. 241, no. 1, pp. 15–18, 2002.
- [108] T. Y. Edward, III-V Nitride Semiconductors: Applications and Devices. CRC Press, 2002, vol. 16.
- [109] I. Adesida, C. Youtsey, A. Ping, F. Khan, L. Romano, and G. Bulman, "Dry and Wet Etching for Group III–Nitrides," in *MRS Proceedings*, vol. 537. Cambridge Univ Press, 1998, pp. G1–4.
- [110] S. J. Pearton, C. R. Abernathy, and F. Ren, "Dry etching of gallium nitride and related materials," *Gallium Nitride Processing for Electronics, Sensors and Spintronics*, pp. 97–177, 2006.
- [111] S. Pearton and R. Shul, "Chapter 5: Etching of III Nitrides," in Semiconductors and Semimetals: Gallium Nitride (GaN) I, J. I. Pankove and T. D. Moustakas, Eds. Elsevier, 1997, vol. 50, pp. 103 – 126.
- [112] D. Basak, M. Verd, M. T. Montojo, M. A. Snchez-Garca, F. J. Snchez, E. Muoz, and E. Calleja, "Reactive ion etching of GaN layers using SF<sub>6</sub>," *Semiconductor Science and Technology*, vol. 12, no. 12, p. 1654, 1997.
- [113] H. Lee, D. B. Oberman, and J. S. Harris, "Reactive ion etching of GaN using CHF<sub>3</sub>/Ar and C<sub>2</sub>ClF<sub>5</sub>/Ar plasmas," *Applied Physics Letters*, vol. 67, no. 12, 1995.
- [114] I. Adesida, A. Mahajan, E. Andideh, M. A. Khan, D. T. Olsen, and J. N. Kuznia, "Reactive ion etching of gallium nitride in silicon tetrachloride plasmas," *Applied Physics Letters*, vol. 63, no. 20, 1993.
- [115] A. Ping, I. Adesida, M. Asif Khan, and J. Kuznia, "Reactive ion etching of gallium nitride using hydrogen bromide plasmas," *Electronics Letters*, vol. 30, no. 22, pp. 1895–1897, 1994.
- [116] R. J. Shul, G. B. McClellan, S. A. Casalnuovo, D. J. Rieger, S. J. Pearton, C. Constantine, C. Barratt, R. F. Karlicek, C. Tran, and M. Schurman, "Inductively coupled plasma etching of GaN," *Applied Physics Letters*, vol. 69, no. 8, 1996.
- [117] Y. H. Lee, H. S. Kim, G. Y. Yeom, J. W. Lee, M. C. Yoo, and T. I. Kim, "Etch characteristics of GaN using inductively coupled Cl<sub>2</sub>/Ar and Cl<sub>2</sub>/BCl<sub>3</sub> plasmas," *Journal of Vacuum Science & Technology A*, vol. 16, no. 3, 1998.
- [118] D. Zhuang and J. Edgar, "Wet etching of GaN, AlN, and SiC," Materials Science and Engineering R, Reports: A Review Journal, vol. 48, no. 1, pp. 1 – 46, 2005.
- [119] T. Anderson, M. Tadjer, M. Mastro, J. Hite, K. Hobart, C. Eddy, and F. Kub, "An AlN/Ultrathin AlGaN/GaN HEMT Structure for Enhancement-Mode Operation Using Selective Etching," *IEEE Electron Device Letters*, vol. 30, no. 12, pp. 1251 –1253, Dec. 2009.

- [120] J. R. Mileham, S. J. Pearton, C. R. Abernathy, J. D. MacKenzie, R. J. Shul, and S. P. Kilcoyne, "Wet chemical etching of AlN," *Applied Physics Letters*, vol. 67, no. 8, pp. 1119 –1121, Aug 1995.
- [121] J. D. Greenlee, T. J. Anderson, B. N. Feigelson, A. D. Koehler, K. D. Hobart, and F. J. Kub, "Characterization of a selective AlN wet etchant," *Applied Physics Express*, vol. 8, no. 3, p. 036501, 2015.
- [122] R. D. Shannon, "Dielectric polarizabilities of ions in oxides and fluorides," Journal of Applied Physics, vol. 73, no. 1, 1993.
- [123] R. M. Luna-Sánchez and I. González-Martínez, "Mixed oxides as high-k gate dielectric films," ECS Transactions, vol. 2, no. 1, pp. 13–22, 2006.
- [124] O. Instruments, "Plasma Enhanced Chemical Vapour Deposition," 2015, Accessed: 5th March, 2015. [Online]. Available: http://www.oxford-instruments. com/products/etching-deposition-and-growth/plasma-etch-deposition/pecvd
- [125] J. Gope, S. Kumar, A. Parashar, S. Dayal, C. Rauthan, and P. Srivastava, "Effect of hydrogen content and bonding environment on mechanical properties of hydrogenated silicon films deposited by high-frequency PECVD process," *International Scholarly Research Notices*, 2012.
- [126] S.-S. Han, B.-H. Jun, K. No, and B.-S. Bae, "Preparation of a SiN<sub>x</sub> thin film with low hydrogen content by inductively coupled plasma enhanced chemical vapor deposition," *Journal of the Electrochemical Society*, vol. 145, no. 2, pp. 652–658, 1998.
- [127] O. "Inductively Coupled Instruments, Plasma Chemical Vapour (ICP CVD)," Deposition 2015.Accessed: 5th March, 2015.[Online]. Available: http://www.oxford-instruments.com/products/ etching-deposition-and-growth/plasma-etch-deposition/icp-cvd
- [128] M. Leskel and M. Ritala, "Atomic layer deposition (ALD): from precursors to thin film structures," Thin Solid Films: Proceedings of the 2nd Asian Conference on Chemical Vapour Deposition, vol. 409, no. 1, pp. 138 – 146, 2002.
- [129] O. Instruments, "Atomic Layer Deposition (ALD)," 2015, Accessed: 5th March, 2015. [Online]. Available: http://www.oxford-instruments.com/products/ etching-deposition-and-growth/plasma-etch-deposition/atomic-layer-deposition
- [130] D. M. Hausmann, E. Kim, J. Becker, and R. G. Gordon, "Atomic layer deposition of hafnium and zirconium oxides using metal amide precursors," *Chemistry of Materials*, vol. 14, no. 10, pp. 4350–4358, 2002.

- [131] T. Ohki, T. Kikkawa, M. Kanamura, K. Imanishi, K. Makiyama, N. Okamoto, K. Joshin, and N. Hara, "An over 100 W AlGaN/GaN enhancement-mode-HEMT power amplifier with piezoelectric-induced cap structure," *Physica Status Solidi C*, vol. 6, no. 6, pp. 1365–1368, 2009.
- [132] C. Ostermaier, G. Pozzovivo, J.-F. Carlin, B. Basnar, W. Schrenk, Y. Douvry, C. Gaquiere, J.-C. DeJaeger, K. Cico, K. Frohlich, M. Gonschorek, N. Grandjean, G. Strasser, D. Pogany, and J. Kuzmik, "Ultrathin InAlN/AlN barrier HEMT with high performance in normally off operation," *IEEE Electron Device Letters*, vol. 30, no. 10, pp. 1030 –1032, Oct. 2009.
- [133] R. S. Q. Fareed, X. Hu, A. Tarakji, J. Deng, R. Gaska, M. Shur, and M. A. Khan, "High-power AlGaN/InGaN/AlGaN/GaN recessed gate heterostructure field-effect transistors," *Applied Physics Letters*, vol. 86, no. 14, pp. 143512 143513, Apr 2005.
- [134] T. Anderson, M. Tadjer, M. Mastro, J. Hite, K. Hobart, C. Eddy, and F. Kub, "Development of enhancement mode aln/ultrathin algan/gan hemts by selective wet etching," *ECS Transactions*, vol. 28, no. 4, pp. 65–70, 2010.
- [135] U. K. Mishra, L. Shen, T. Kazior, and Y.-F. Wu, "GaN-based RF power devices and amplifiers," *Proceedings of the IEEE*, vol. 96, no. 2, pp. 287–305, Feb 2008.
- [136] F. Zhihong, X. Shengyin, Z. Rui, Y. Jiayun, Z. Wei, and C. Shujun, "A highperformance enhancement-mode AlGaN/GaN HEMT," *Journal of Semiconductors*, vol. 31, no. 8, p. 084001, 2010.
- [137] M. Xiao-Hua, Y. Hui-You, Q. Si, Y. Li-Yuan, P. Cai-Yuan, Y. Ling, W. Hao, Z. Jin-Cheng, and H. Yue, "Enhancement-mode AlGaN/GaN high electronic mobility transistors with thin barrier," *Chinese Physics B*, vol. 20, no. 2, p. 027303, 2011.
- [138] K. J. Chen and C. Zhou, "Enhancement-mode AlGaN/GaN HEMT and MIS-HEMT technology," *Physica Status Solidi* (a), vol. 208, no. 2, pp. 434–438, 2011.
- [139] X. Hu, G. Simin, J. Yang, M. Asif Khan, R. Gaska, and M. Shur, "Enhancement mode AlGaN/GaN HFET with selectively grown pn junction gate," *Electronics Letters*, vol. 36, no. 8, pp. 753–754, Apr 2000.
- [140] T. Fujii, N. Tsuyukuchi, Y. Hirose, M. Iwaya, S. Kamiyama, H. Amano, and I. Akasaki, "Fabrication of enhancement-mode Al<sub>x</sub>Ga<sub>1x</sub>N/GaN junction heterostructure field-effect transistors with p-type GaN gate contact," *Physica Status Solidi* (c), vol. 4, no. 7, pp. 2708–2711, 2007.

- [141] T. Mizutani, M. Ito, S. Kishimoto, and F. Nakamura, "AlGaN/GaN HEMTs With Thin InGaN Cap Layer for Normally Off Operation," *IEEE Electron Device Let*ters, vol. 28, no. 7, pp. 549 –551, July 2007.
- [142] S. Vitanov and V. Palankovski, "Normally-off AlGaN/GaN HEMTs with InGaN cap layer: A theoretical study," in *International Semiconductor Device Research* Symposium, Dec. 2007, pp. 1–2.
- [143] M. Ito, S. Kishimoto, F. Nakamura, and T. Mizutani, "Enhancement-mode Al-GaN/GaN HEMTs with thin InGaN cap layer," *Physica Status Solidi (c)*, vol. 5, no. 6, pp. 1929–1931, 2008.
- [144] L. Yuan, H. Chen, and K. Chen, "Normally Off AlGaN/GaN Metal-2DEG Tunnel-Junction Field-Effect Transistors," *IEEE Electron Device Letters*, vol. 32, no. 3, pp. 303 –305, March 2011.
- [145] L. Yuan, H. Chen, Q. Zhou, C. Zhou, and K. Chen, "A novel normally-off GaN power tunnel junction FET," in *IEEE 23rd International Symposium on Power* Semiconductor Devices and ICs (ISPSD), May 2011, pp. 276–279.
- [146] K. Chen, Private Communications, 2013.
- [147] S. Rajan, C. Suh, J. Speck, and U. Mishra, "N-polar aluminum gallium nitride/gallium nitride enhancement-mode field effect transistor," May 24 2011, US Patent 7,948,011.
- [148] U. Singisetti, M. H. Wong, S. Dasgupta, Nidhi, B. Swenson, B. Thibeault, J. Speck, and U. Mishra, "Enhancement-mode N-polar GaN MISFETs with self-aligned source/drain regrowth," *IEEE Electron Device Letters*, vol. 32, no. 2, pp. 137 –139, Feb. 2011.
- [149] S. Rajan, A. Chini, M. H. Wong, J. S. Speck, and U. K. Mishra, "N-polar GaN/Al-GaN/GaN high electron mobility transistors," *Journal of Applied Physics*, vol. 102, no. 4, p. 044501, 2007.
- [150] R. Dimitrov, M. Murphy, J. Smart, W. Schaff, J. R. Shealy, L. F. Eastman, O. Ambacher, and M. Stutzmann, "Two-dimensional electron gases in Ga-face and Nface AlGaN/GaN heterostructures grown by plasma-induced molecular beam epitaxy and metalorganic chemical vapor deposition on sapphire," *Journal of Applied Physics*, vol. 87, no. 7, pp. 3375–3380, 2000.

- [151] S. Keller, N. A. Fichtenbaum, F. Wu, D. Brown, A. Rosales, S. P. DenBaars, J. S. Speck, and U. K. Mishra, "Influence of the substrate misorientation on the properties of N-polar GaN films grown by metal organic chemical vapor deposition," *Journal of Applied Physics*, vol. 102, no. 8, p. 083546, 2007.
- [152] Y. Zhang, X. Dong, G. Li, W. Li, B. Zhang, and G. Du, "High quality N-polar GaN two-dimensional growth on c-plane sapphire by metalorganic vapor phase epitaxy," *Journal of Crystal Growth*, vol. 366, no. 0, pp. 35 – 38, 2013.
- [153] W. Saito, Y. Takada, M. Kuraguchi, K. Tsuda, and I. Omura, "Recessed-gate structure approach toward normally off high-voltage AlGaN/GaN HEMT for power electronics applications," *IEEE Transactions on Electron Devices*, vol. 53, no. 2, pp. 356 – 362, Feb. 2006.
- [154] M. S. Minsky, M. White, and E. L. Hu, "Room temperature photoenhanced wet etching of GaN," *Applied Physics Letters*, vol. 68, no. 11, pp. 1531 –1533, Mar 1996.
- [155] C. Youtsey, I. Adesida, and G. Bulman, "Highly anisotropic photoenhanced wet etching of n-type GaN," *Applied Physics Letters*, vol. 71, no. 15, pp. 2151 –2153, Oct 1997.
- [156] L.-H. Peng, C.-W. Chuang, Y.-C. Hsu, J.-K. Ho, C.-N. Huang, and C.-Y. Chen, "Deep ultraviolet enhanced wet chemical oxidation and etching of Gallium Nitride," in Summaries of papers presented at the Conference on Lasers and Electro-Optics (CLEO). Technical Digest., May 1998, p. 153.
- [157] T. Rotter, D. Mistele, J. Stemmer, F. Fedler, J. Aderhold, J. Graul, V. Schwegler, C. Kirchner, M. Kamp, and M. Heuken, "Photoinduced oxide film formation on n-type GaN surfaces using alkaline solutions," *Applied Physics Letters*, vol. 76, p. 3923, Jun. 2000.
- [158] W. Lanford, T. Tanaka, Y. Otoki, and I. Adesida, "Recessed-gate enhancementmode GaN HEMT with high threshold voltage," *Electronics Letters*, vol. 41, no. 7, pp. 449 – 450, march 2005.
- [159] T. Hashizume, S. Anantathanasarn, N. Negoro, E. Sano, H. Hasegawa, K. Kumakura, and T. Makimoto, "Al<sub>2</sub>O<sub>3</sub> insulated-gate structure for AlGaN/GaN heterostructure field effect transistors having thin AlGaN barrier layers," *Japanese Journal of Applied Physics*, vol. 43, no. 6B, p. L777, 2004.
- [160] A. Endoh, Y. Yamashita, K. Ikeda, M. Higashiwaki, K. Hikosaka, T. Matsui, S. Hiyamizu, and T. Mimura, "Non-recessed-gate enhancement-mode AlGaN/GaN

high electron mobility transistors with high RF performance," *Japanese Journal* of Applied Physics, vol. 43, no. 4S, p. 2255, 2004.

- [161] Y. Ohmaki, M. Tanimoto, S. Akamatsu, and T. Mukai, "Enhancement-mode Al-GaN/AlN/GaN high electron mobility transistor with low On-state resistance and high breakdown voltage," *Japanese Journal of Applied Physics*, vol. 45, no. 11L, p. L1168, 2006.
- [162] K. Matocha, T. Chow, and R. Gutmann, "High-voltage normally off GaN MOS-FETs on sapphire substrates," *IEEE Transactions on Electron Devices*, vol. 52, no. 1, pp. 6 – 10, jan. 2005.
- [163] S. Sugiura, S. Kishimoto, T. Mizutani, M. Kuroda, T. Ueda, and T. Tanaka, "Normally-off AlGaN/GaN MOSHFETs with HfO<sub>2</sub> gate oxide," *Physica Status Solidi* (c), vol. 5, no. 6, pp. 1923–1925, 2008.
- [164] T. EDA, High Performance EDA Tools for Analog/Mixed-Signal ICs & ASICs -L-EDIT FOR PHYSICAL DESIGN. [Online]. Available: http://www.tannereda. com/images/pdfs/Revised\_Datasheets/ds\_ledit\_rev.pdf
- [165] Z. Fan, S. N. Mohammad, W. Kim, z. Aktas, A. E. Botchkarev, and H. Morko, "Very low resistance multilayer Ohmic contact to n-GaN," *Applied Physics Letters*, vol. 68, no. 12, 1996.
- [166] Y. Lu, S. Yang, Q. Jiang, Z. Tang, B. Li, and K. J. Chen, "Characterization of V<sub>T</sub>-instability in enhancement-mode Al<sub>2</sub>O<sub>3</sub>-AlGaN/GaN MIS-HEMTs," *Physica Status Solidi* (c), vol. 10, no. 11, pp. 1397–1400, 2013.
- [167] I. Thayne, M. Cäsar, P. Chalker, S.-J. Cho, P. Evans, K. Floros, A. Forsyth, T. Foxon, I. Guiney, S. H. Vajargah, P. Houston, C. Humphreys, S. Jiang, M. Johnson, S. Karboyan, M. Kuball, K. B. Lee, X. Li, S. Novikov, D. Paul, H. Qian, J. Roberts, G. Ternent, M. Uren, D. Wallis, W. Waller, E. Wasige, and Z. Zaidi, "The PowerGaN Consortium Current status and future plans," UK Nitrides Consortium, Nottingham, Tech. Rep., January 2015.
- [168] S. Arulkumaran, T. Egawa, L. Selvaraj, and H. Ishikawa, "On the effects of gaterecess etching in current-collapse of different cap layers grown AlGaN/GaN highelectron-mobility transistors," *Japanese Journal of Applied Physics*, vol. 45, no. 3L, p. L220, 2006.
- [169] W. Saito, M. Kuraguchi, Y. Takada, K. Tsuda, I. Omura, and T. Ogura, "Influence of surface defect charge at AlGaN-GaN-HEMT upon Schottky gate leakage current and breakdown voltage," *IEEE Transactions on Electron Devices*, vol. 52, no. 2, pp. 159–164, Feb 2005.

- [170] M. Briere, "Bringing GaN on Si based power devices to market," Bodo's Power Systems. [Online]. Available: www.bodospower.com
- [171] B. Gil, III-nitride semiconductors and their modern devices. Oxford University Press, 2013.
- [172] J. Zhang, "GaN-based device cascoded with an integrated FET/Schottky diode device," Dec. 27 2011, US Patent 8,084,783.
- [173] J. Shi, L. F. Eastman, X. Xin, and M. Pophristic, "High performance AlGaN/GaN power switch with HfO<sub>2</sub> insulation," *Applied Physics Letters*, vol. 95, no. 4, 2009.
- [174] R. Lossy, H. Gargouri, M. Arens, and J. Würfl, "Gallium nitride MIS-HEMT using atomic layer deposited Al<sub>2</sub>O<sub>3</sub> as gate dielectric," *Journal of Vacuum Science & Technology A*, vol. 31, no. 1, 2013.
- [175] K. Floros, Private Communications, 2014.
- [176] X. Huang, Z. Liu, Q. Li, and F. Lee, "Evaluation and application of 600 V GaN HEMT in cascode structure," *IEEE Transactions on Power Electronics*, vol. 29, no. 5, pp. 2453–2461, May 2014.
- [177] X. Huang, Q. Li, Z. Liu, and F. Lee, "Analytical loss model of high voltage GaN HEMT in cascode configuration," in *IEEE Energy Conversion Congress and Exposition (ECCE)*, Sept 2013, pp. 3587–3594.
- [178] T. Hirose, M. Imai, K. Joshin, K. Watanabe, T. Ogino, Y. Miyazaki, K. Shono, T. Hosoda, and Y. Asai, "Dynamic performances of GaN-HEMT on Si in cascode configuration," in 29th Annual IEEE Applied Power Electronics Conference and Exposition (APEC), March 2014, pp. 174–181.
- [179] GaN Systems Inc., "Design considerations for driving GaN Systems power switches," 2015, Accessed: 25th March, 2015. [Online]. Available: http://www.gansystems.com/\_uploads/whitepapers/897460\_App%20Note% 20-%20Design%20considerations%20for%20driving%20GaN%20HEMTs.pdf
- [180] S. Tiwari, *Compound semiconductor device physics*. Academic Press, 1992.
- [181] D. K. Schroder, Semiconductor material and device characterization. John Wiley & Sons, 2006.
- [182] M. Ahtaiba, "Fabrication and characterisation of E-mode AlGaN/GaN MOS Capacitor and E-mode AlGaN/GaN HEMT with sub-critical barrier thickness," MSc. Thesis, November 2014.

[183] Y. Hori, Z. Yatabe, and T. Hashizume, "Characterization of interface states in Al2O3/AlGaN/GaN structures for improved performance of high-electronmobility transistors," *Journal of Applied Physics*, vol. 114, no. 24, pp. –, 2013.